The Impact of COVID-19 on Inclusive Education at the European Level





THE IMPACT OF COVID-19 ON INCLUSIVE EDUCATION AT THE EUROPEAN LEVEL



The European Agency for Special Needs and Inclusive Education (the Agency) is an independent and self-governing organisation. The Agency is co-funded by the ministries of education in its member countries and by the European Commission via an operating grant within the European Union (EU) education programme.



The European Commission's support for the production of this publication does not constitute an endorsement of the contents, which reflect the views only of the authors, and the Commission cannot be held responsible for any use which may be made of the information contained therein.

The views expressed by any individual in this document do not necessarily represent the official views of the Agency, its member countries or the European Commission.

© European Agency for Special Needs and Inclusive Education 2021

Editor: Cristina Popescu

This publication is an open-access resource. This means you are free to access, use and disseminate it with appropriate credit to the European Agency for Special Needs and Inclusive Education. Please refer to the Agency's Open Access Policy for more information: www.european-agency.org/open-access-policy.

You may cite this publication as follows: European Agency for Special Needs and Inclusive Education, 2021. *The Impact of COVID-19 on Inclusive Education at the European Level: Literature Review.* (C. Popescu, ed.). Odense, Denmark



This work is licensed under a <u>Creative Commons Attribution-NonCommercial-NoDerivatives 4.0 International License</u>.

You may not modify or translate this publication without the Agency's approval.

With a view to greater accessibility, this report is available in accessible electronic format on the Agency's website: www.european-agency.org

ISBN: 978-87-7110-938-2 (Electronic)

Secretariat

Østre Stationsvej 33 DK-5000 Odense C Denmark

Tel.: +45 64 41 00 20

secretariat@european-agency.org

Brussels Office

Rue Montoyer 21 BE-1000 Brussels Belgium

Tel.: +32 2 213 62 80

brussels.office@european-agency.org



CONTENTS

EXECUTIVE SUMMARY	7
1. INTRODUCTION	9
1.1. Aims and questions the review addresses	9
1.2. Review structure	12
2. METHODOLOGY	13
2.1. Search criteria overview	13
2.2. Analysis and synthesis	14
3. GLOBAL AND EUROPEAN EDUCATION CO-ORDINATION	16
3.1. Worldwide responses to the COVID-19 crisis in education	16
3.1.1. A 'build back better' perspective	16
3.1.2. Responses to a regressive impact	_ 17
3.1.3. Key points	
3.2. Initial European reactions and responses	17
3.2.1. Emergency measures for disadvantaged learners and inclusive education	_
3.2.2. A common direction for education and democracy	
3.2.3. Key points	
3.3. Challenges and opportunities for the European Union	
3.3.1. Redefining policy co-operation at European level	
3.3.2. Limited policy actions for disadvantaged learners	_
3.3.3. Key points	_ 22 24
3.4. European Union contributions to support equity in education and combat new	_ 24
3.4.1. European Union programmes to support investment in education and training 3.4.2. European Child Guarantee	24 25
3.4.3. Key points	_ 25 _ 25
4. GOVERNANCE OF INCLUSIVE EDUCATION: LESSONS LEARNED FOR NECESSARY FUTURE CHANGES	26
4.1. COVID-19 and education: assessment and accountability in times of crisis	
4.1.1. National assessment of post-COVID-19 education in Europe 4.1.2. International assessment and accountability of post-COVID-19 education	_ 26 27
4.1.3. Key points	_ 27 28
4.2. Financing and provision	_ 28
4.2.1. Limited evidence on COVID-19's impact on inclusive education funding	_ 28



4.2.2. Emergency funding measures at European level	29
4.2.3. Recommendations from civil society organisations	_ _ 29
4.2.4. Future scenarios on funding for disadvantaged learners	_ 30
4.2.5. Key points	_31
4.3. Helping schools to mitigate inequalities	_32
4.3.1. Various guidelines	_ 32
4.3.2. School closure and re-opening strategies	_ _ 33
4.3.3. Support for teachers	_ 34
4.3.4. Support for vulnerable and disadvantaged schools	
4.3.5. Key points	_36
4.4. Holistic approach to education and cross-sectoral collaboration	_36
4.4.1. Cross-sectoral collaboration for more adapted responses to the COVID-19 crisi	s 36
4.4.2. The COVID-19 crisis as a reason for new partnerships in education	
4.4.3. Further recommendations for cross-sectoral collaborations	_
4.4.4. Key points	_ 39
5. TRANSFORMATION OF EDUCATION	_40
5.1. Acknowledgment of learning loss	_40
5.1.1. Severe learning loss for vulnerable learners	_ 40
5.1.2. Action to reduce learning gaps and learning loss	_41
5.1.3. Key points	_41
5.2. Impact on curricula, assessment and learner progress	_41
5.2.1. The need to redefine curricula, assessment and learner progress	41
5.2.2. Recommendation for educational practices during COVID-19	_ _ 42
5.2.3. Key points	_ 43
5.3. Remote and distance learning	_43
5.3.1. Online distance learning as a global solution	_ 43
5.3.2. International recommendations	_ _ 44
5.3.3. Key points	_46
5.4. Addressing the digital divide	_46
5.4.1. Access to digital technologies and materials	46
5.4.2. Inclusive digital practices and pedagogies	
5.4.3. Key points	
6. LEARNERS' VOICES AND FAMILIES' EFFORTS IN EDUCATION	_48
6.1. Parents' transformed role in education	_48
6.1.1. Assessing COVID-19's impact on families	_ 48
6.1.2. Recommendations on working with families	_ _ 49
6.1.3. Evidence of measures encouraging communication between schools and	
families	_ 49
6.1.4. Kev points	50



6.2. Children's rights and disability rights to education	50
6.2.1. The right to education	50
6.2.2. Participation in education responses	
6.2.3. Key points	52
6.3. Well-being – mental and emotional health	52
6.3.1. Increased vulnerability for at-risk children	52
6.3.2. Making children's well-being a priority for policy action	53
6.3.3. Key points	55
7. A SYNTHESIS OF KEY MESSAGES AND RECOMMENDATIONS	56
8. CONCLUDING COMMENTS	58
9. REFERENCES	60
10. BIBLIOGRAPHY – COMPLETE LIST OF RESOURCES	71
10.1 Bibliography	71
10.2 List of surveys on COVID-19's impact on education	93
11. ANNEX: OPERATIONAL DEFINITIONS	96



EXECUTIVE SUMMARY

This literature review aims to map evidence and identify acknowledged ways in which COVID-19 has impacted on education in general and inclusive education in particular at European and national levels. The intention is to provide a comprehensive overview of key messages for member countries of the European Agency for Special Needs and Inclusive Education (the Agency). It will also inform possible future Agency work on this topic.

Several questions guide the review:

- What information is available about COVID-19's impact on education? How are the topics framed in international, European and national documents?
- How have education stakeholders adapted to the needs and context of vulnerable learners (e.g. change in teaching methods; governments and private sector providing accessible devices and connectivity)?
- What evidence from European countries can support future decision-making and collaboration with various stakeholders?
- How can European countries learn and collaborate to minimise the effects of the current education crisis on vulnerable learners?

In total, 250 documents were reviewed, including international and national guidelines, policy briefs, white papers, political declarations, reports based on quantitative and qualitative surveys, academic papers, blog articles, online databases, portals and webinars. The documents covered the period March—December 2020. They were analysed using a **grounded approach**. This means that general ideas, recommendations and explanations were grounded in, or emerged from, 'data systematically obtained from social research' (Glaser and Strauss, 1967, p. 2).

Two phases of analysis identified four main thematic areas:

- Global and European education
- Governance of inclusive education
- Transformation of education
- · Learners' voices and families' efforts.

Each thematic area developed into a specific section of the review. Lessons learned and future directions for action in inclusive education were identified from the documents reviewed in each thematic area.

The review's findings focus on:

1. COVID-19's impact on the co-ordination of inclusive education at the European level

Limited initial guidance on inclusive education at European level was available. COVID-19 was considered both a challenge and an opportunity to review and implement more inclusive legislation, policies and actions in education across Europe. A 'build back better' message linked to education is predominant. Equality and inclusion are its core values.



2. COVID-19's impact on the governance of inclusive education at national level

Evidence showed that all European countries took measures to ensure the continuity of education during the pandemic. However, these measures were sometimes inadequate, and limited guidance was available for inclusion in education. Recommendations from various stakeholders highlighted the need to replace the initial compensatory response to the crisis with more preventive policies based on cross-sectoral collaborations.

3. COVID-19's impact and the transformation of inclusive education

Remote, digital education was the preferred first response to the COVID-19 crisis in all European countries. Vulnerable learners were already at greater risk of learning loss and were more affected by the digital divide. Existing recommendations focus on the need for co-ordination and remediation measures to reduce learning loss. More preventive measures are also recommended for accessibility of learning and quality in education.

4. COVID-19's impact on vulnerable learners and their families

Limited support for vulnerable children and their families was available across European countries. The crisis affected multiple dimensions of children's well-being. Children's voices were seldom heard in public debates about education. The analysed documents recommend more collaboration between schools, parents and vulnerable learners. Policy-makers worldwide made preserving children's well-being a priority in their action. These measures are expected to be developed further.

Based on these findings, the review recommends further directions for Agency work:

- Focus on and monitor future developments in priority topics, such as learning loss, cross-sectoral partnerships, remote digital education, and children's well-being.
- Additional monitoring, based on the framework of four thematic areas developed in the review. As the situation is dynamic and changing, new priorities may appear.
 A follow-up will be needed to highlight new interests and topics on the agenda of inclusive education during the COVID-19 pandemic.
- More systematic information collection and coherent presentation of evidence and measures on inclusive education practice during COVID-19 times in Europe.
- Use the current document as a living document to build on. The literature review offers a base framework that can be transformed over time, according to future Agency work on the topic. Key resources could be collected together in one place for the reference of all Agency members.



1. INTRODUCTION

The World Health Organization (WHO) Director-General declared the coronavirus (COVID-19) outbreak a Public Health Emergency of International Concern on 30 January 2020 (WHO, 2020a). On 11 March 2020, the WHO upgraded the COVID-19 outbreak to a global pandemic (WHO, 2020b).

At the end of 2020, school closures were still affecting approximately 139,000,000 learners enrolled in pre-primary, primary, lower-secondary and upper-secondary levels of education (International Standard Classification of Education (ISCED) levels 0 to 3), as well as at tertiary education levels (ISCED levels 5 to 8) (UNESCO, 2020a).

The COVID-19 crisis highlighted a pre-existing crisis in education. It made inequalities more visible (UNESCO, 2021a), and became a reason for systemic change in education worldwide (Ainscow, 2005; Ainscow and Haile-Giorgis, 1999). Prior to the COVID-19 crisis, Ainscow and Haile-Giorgis explained the meaning and importance of systemic change in education to promote more inclusive approaches:

... methodological and organizational changes made in response to pupils experiencing barriers to their learning are, under certain conditions, likely to benefit all children, thus linking together the pursuit of equity and excellence. It involves a continuous process of school improvement aimed at using available resources, particularly human resources, in order to support the participation and learning of all pupils within a local community. In this way, those pupils seen as having special needs come to be regarded as the stimulus that can encourage developments towards a much richer overall educational environment (1999, p. 106).

To achieve educational change, 'a debate on purpose and outcomes involving all stakeholders, including learners and families' is also required (European Agency, 2013, p. 22).

During the COVID-19 pandemic, educational change focused on emergency responses to the crisis, through various modes of emergency remote schooling (Vuorikari, Velicu, Chaudron, Cachia and Di Gioia, 2020) or other preparedness measures inspired by the field of Education in Emergencies (INEE, 2020a; 2020b). Then, calls were made for more permanent action, involving the development of prevention-based policy approaches (Watkins, Donnelly and Meijer, forthcoming).

1.1. Aims and questions the review addresses

Since the beginning of the crisis, multiple documents at international, European and national levels have mentioned and analysed COVID-19's impact on education. However, the common messages in the documents have to be further explored and better explained.

Despite the multiplicity of sources, there is an urgent need for information and evidence on effective measures to improve policy, school activities and vulnerable learners'



learning and life experiences during the COVID-19 crisis. The official documents included in this literature review acknowledged the need to strengthen inclusive education support and practice during the COVID-19 pandemic. New assessment reports and tools, opportunities for exchange and co-ordination at European level are still to be developed.

This literature review aims to map evidence and identify COVID-19's acknowledged impacts on education in general and inclusive education in particular at European and national levels. The intention is to provide a comprehensive overview of key messages for Agency member countries and to inform future Agency work. The review aims to provide useful insights for European policy-makers – national, state or local education authorities, or leaders of education networks – and support their responses and intervention in relation to the on-going crisis.

Co-operation between education leaders could be a strategy to respond to and face the current crisis in education. One of the 'first and simplest' forms of co-operation in education (Reimers and Schleicher, 2020, p. 4) is to exchange knowledge on practices and lessons learned at the level of each school, community or country.

Access to information and evidence is essential for change, development and improvement in all education systems (Ramberg and Watkins, 2020) and supports the decisions and actions of policy-makers in inclusive education. 'There is widespread awareness that evidenced-based policy-making is critical for the long-term development of inclusive education systems' (Watkins and Ebersold, 2016, p. 229).

This review is linked to the Agency's intention to conduct a longer-term project investigation about COVID-19's impact on inclusive education at the European level. It aims to explore how the COVID-19 pandemic has affected inclusive education systems in Europe, and how improved co-ordination at European level can 'build back better'.

Several questions guide the review:

- What information is available about COVID-19's impact on education? How are the topics framed in international, European and national documents?
- How have education stakeholders adapted to the needs and context of vulnerable learners (e.g. change in teaching methods; governments and private sector providing accessible devices and connectivity)?
- What evidence from European countries can support future decision-making and collaboration with the various stakeholders?
- How can European countries learn and collaborate to minimise the effects of the current education crisis on vulnerable learners?

COVID-19 has affected all stakeholders, all levels of education and all countries around the world. In particular, the current review focuses on the European situation. It aims to:

- identify relevant documents at international, European and national levels;
- determine themes related to COVID-19's impact on education in Europe;
- highlight lessons already learned, that are shown in the responses to the crisis;
- summarise future directions for action, as suggested by the analysed information.



Evidence about lessons learned and recommendations for future actions could potentially inform several levels relevant to public action:

- the international and European level, where country representatives can find guidance, but also meet and exchange practices;
- the national level, where a deeper understanding of governance measures during the crisis is needed;
- the school level, where challenges to education need to be better systematised;
- the level of learners and families, where the rights to education and civic participation are under threat.

The levels are not isolated, and the stakeholders directly or indirectly communicate with each other. For instance, decisions at national level have an impact on school organisation, while learners and families can formulate recommendations for policy-making processes. These dimensions resonate with the 'Ecosystem of Support for Inclusive Education' model (European Agency, 2017a; 2017b), which allows for better understanding of the processes and interactions that promote or challenge progress in inclusive education.

This model was originally developed within the Agency's Inclusive Early Childhood Education project ... It sets out the main structures and processes that affect learning and achievement. It highlights the inter-connections between national policy and the processes and structures at all other levels. It may be used to reflect on how wider policy supports local and school practice (European Agency, 2017b, p. 4).

In accordance with previous Agency work (e.g. European Agency, 2009; 2017c), the review pays attention to:

- inputs to education systems, such as the transformation of resources and funding, emergency teacher training, and adjustments to existing curricula;
- processes within education systems, such as the transformation of procedures, the transition to digital classrooms and learning, the transformation of teaching strategies, and co-operation between teachers;
- outputs, or the consequences of the inputs and processes in education, e.g. school attendance and achievements, participation in learning-related decisions, and the transformation of learners' well-being.



1.2. Review structure

The review is divided into eight sections and additional information:

- This <u>section 1</u> introduces the reasons for the review and the essential line of argument. It also clarifies specific terms used in the review.
- Section 2 explains the methodology and analysis criteria used for the review.
- Sections 3–6 summarise COVID-19's impact on four areas where educational change is at work. Each of these sections embeds the findings and recommendations of the considered sources, and the identification of key learning points for future work about COVID-19's impact on inclusive education in European countries.
 - Section 3 focuses on a transnational vision of COVID-19's impact on education in general and inclusive education in particular. It examines common responses and recommendations for action worldwide and in Europe.
 - Section 4 explores the national responses to the crisis and the challenges or innovations in the governance of (inclusive) education.
 - Section 5 addresses new vulnerabilities, tensions and solutions linked to the current transformations of (inclusive) education.
 - Section 6 investigates the impact of the crisis on the well-being of learners and their families, and focuses on the democratic and social functions of education.
- <u>Section 7</u> summarises the lessons learned and the recommendations made in the reviewed literature for future actions related to the COVID-19 pandemic's impact on inclusive education. It links and compares information available in sections 3–6.
- Section 8 highlights emerging issues for possible future Agency work on this topic.
- The <u>References</u> include the documents that support evidence and in-depth analysis in the literature review.
- The <u>Bibliography</u> contains the complete list of 250 documents collected for the literature review.
- The <u>Annex</u> lists a number of operational definitions used in the literature review.



2. METHODOLOGY

This section outlines the methodology used to identify and further analyse the international, European and national documents considered in the literature review.

2.1. Search criteria overview

Documents linked to COVID-19's impact on inclusive education and general education produced between March and December 2020 were identified for review.

The search process took place at several levels and involved protocol-driven or recommendation-based collection techniques. It included:

- an online search of the websites of major international bodies: United Nations Educational, Scientific and Cultural Organization (UNESCO), United Nations Children's Fund (UNICEF), Organisation for Economic Co-operation and Development (OECD), World Bank, etc.;
- systematic monitoring of new policy and academic publications (based on newsletters, webinar resource sharing and other online updates), given the contemporary period covered;
- an online search of the websites of major academic journals specialising in education, inclusive education or special educational needs;
- a search through 'snowball methods', such as 'pursuing references of references and electronic citation' tracking. These are especially 'powerful for identifying high quality sources' that may not necessarily be accessible through a primary search (Greenhalgh and Peacock, 2005);
- personal knowledge and suggestions from within the Agency.

Basic keywords and search syntaxes, such as 'COVID-19 AND education', 'COVID-19 AND inclusion' and 'COVID-19 AND inclusive education', were used for the protocol-driven collection steps.

The collected documents were 'cleaned' based on a first screening, mainly at the level of titles and abstracts. Duplications were deleted, along with documents that had limited information relevant to the research question. Documents with national, European or international coverage of COVID-19's impact on (inclusive) education were considered in the corpus, while documents focusing on very local aspects, without evidence potential, were eliminated.

The major portals of the international organisations offered some challenges, as they include exhaustive information. Each portal was considered as an item/document within the final list of documents, as each could include general relevant information. Relevant documents collected from these portals were considered as separate items.

The data collection included several categories of documents: international and national guidelines, policy briefs, white papers, political declarations, reports based on quantitative and qualitative surveys, academic papers (both journal articles and research reports), blog



articles, online databases and major portals, and webinars. The documents have various formats: PDF, Word, online pages, audio files, and video files. The main categories of sources that produced these documents are international organisations (e.g. UNESCO, OECD, Council of Europe), civil society organisations (including disabled people's organisations), universities and academic journals.

The process identified 250 documents, which gave a first general overview of the coverage of COVID-19's impact on (inclusive) education at the national, European and international levels. The complete list of these documents is available in the <u>Bibliography</u>.

The documents' contents were further analysed for the report.

2.2. Analysis and synthesis

Analysis is understood as a systematic process of examining and arranging collected documents and information to increase understanding of a specific topic. The analysis process for the current review is based on a 'grounded approach', which was initially defined as 'the discovery of theory from data systematically obtained from social research' (Glaser and Strauss, 1967, p. 2). In other words, general ideas, recommendations and explanations that will inform future work and decisions are grounded in – or emerge from – previously collected data.

This literature review analysed the collected documents during two phases. In the first phase of the analysis:

- All the collected documents were organised and coded, according to the information they included, in Zotero. Zotero is a free reference management software that helps to manage bibliographic data and research materials.
- The codes or categories were based on the words and phrases used in the
 documents, but also on themes of interest formulated by Agency country
 representatives: actions for inclusive education and learners with special
 educational needs, accessible digital education, lessons learned, cross-sectoral
 collaboration, learners' views (children's voices), social and mental health effects
 on learners, academic impact.
- Categories were refined when slight differences between them were identified.
- Based on the initial categories, and following a 'saturation' criterion (Glaser and Strauss, 1967, p. 61), a framework specific to the topic developed. This allowed better organisation and summary of the documents and findings.

Data saturation refers to the point in the research process when no new information is discovered in data analysis, and this redundancy signals to researchers that data collection may cease (Faulkner and Trotter, 2017).

- Evidence and findings from 250 documents were organised under four key topic areas, which sections 3 to 6 of this report explore (some of the documents covered multiple topic areas):
 - Global and European education (37 items/documents)



- Governance of inclusive education (96 items/documents)
- Transformation of education (121 items/documents)
- Learners' voices and families' efforts (62 items/documents).

In the second phase of the analysis:

- Documents were assessed and marked as being of low, medium or high relevance to the review.
- The relevance level depended on the following criteria:
 - Documents included data from surveys at national, European and international levels
 - Documents were produced by international organisations and had a strong summarising potential and impact on further policy decisions at international or country levels
 - Documents were used as reference documents by policy-makers at European and country levels
 - Documents had a main focus on inclusive education during the COVID-19 pandemic
 - Documents referred to the compulsory education systems in Agency member countries.
- Approximately 70 documents were identified using these criteria. They were
 actively used and are directly quoted in the review. The <u>References</u> list them.

The advantage of the 'grounded approach' is its dynamic dimension. Established categories and subcategories can change when new materials are collected. The current review offers a base framework that can be transformed over time, according to future Agency work on the topic.



3. GLOBAL AND EUROPEAN EDUCATION CO-ORDINATION

This section focuses on the lessons learned about the impact of the COVID-19 crisis on education at European level. First, it connects the European situation to dynamics identified worldwide. Then, it adopts a temporal perspective and shows how European institutions and countries adapted to the new circumstances. It continues by showing how challenges can be transformed into opportunities for the European Union (EU) to redefine its priorities for quality inclusive education. Finally, it highlights new European contributions to support equity in education.

3.1. Worldwide responses to the COVID-19 crisis in education

3.1.1. A 'build back better' perspective

The COVID-19 crisis is multi-layered. 'The pandemic is more than a health crisis; it is an economic crisis, a humanitarian crisis, a security crisis, and a human rights crisis' (United Nations, 2020a, p. 1). It has affected individuals, families and societies. Many discourses underline how the crisis has highlighted existing vulnerabilities, and see it as an occasion to remake or reimagine better societies.

A 'build back better' perspective and programme for action is apparent throughout the declarations, documents and recommendations. Fragmented approaches, focusing on individual countries or domains, are not expected to succeed. Joint action at health, socioeconomic, humanitarian or human rights levels is expected: 'Coming out of this crisis will require a whole-of-society, whole-of-government and whole-of-the-world approach driven by compassion and solidarity' (ibid., p. 1).

International and European organisations see the crisis as an opportunity to address previously vulnerable policies and actions, by maintaining the existing guidance of the Sustainable Development Agenda (United Nations, 2020a). However, the crisis might also have a **regressive impact** on the progress of the United Nations (UN)'s Sustainable Development Goals (SDGs): health and well-being are affected (SDG 3), a global economic recession might increase inequalities and poverty (SDG 1) and food insecurity (SDG 2), environmental sustainability is endangered (SDGs 7–9 and 11–15), and inclusive and equitable access to education is limited (SDG 4).

The aim of SDG 4 – to achieve inclusive and equitable access to education – also looks likely to be missed, with a projection that more than 200 million children will still be out of education by 2030. Most of the world's children have been deprived of formal education during the COVID-19 outbreak – a legacy that could threaten the SDGs' underlying ambition to leave no one behind (Lancet Public Health, 2020).



3.1.2. Responses to a regressive impact

Realigning all countries' national priorities, long-term co-operation and accelerated actions are recommended to counteract the dangers linked to the COVID-19 crisis. Equality and inclusion emerge as main values to rebuild a better world. For António Guterres, UN Secretary-General:

Everything we do during and after this crisis [COVID-19] must be with a strong focus on building more equal, inclusive and sustainable economies and societies that are more resilient in the face of pandemics, climate change, and the many other global challenges we face (United Nations, 2020b).

All the international organisations examined in this literature review have produced guidelines on the possible responses to the COVID-19 crisis or have formulated recommendations for country governments. Measures for vulnerable or disadvantaged learners and schools, inclusive education and pedagogical practices are included in a limited manner. Section 4.1.2 provides a more detailed list of these resources.

3.1.3. Key points

- A 'build back better' perspective and programme for action is common to the majority of discourses.
- The recommendation for joint action to 'build back better', and implicitly for a whole-system approach, is a response to evidence on existing fragmented or isolated country responses to the crisis.
- Equality and inclusion are core values of the 'build back better' approach.
- COVID-19 is both a challenge and an opportunity for inclusive education worldwide: previous SDG progress might have been halted, but the crisis could encourage a review of non-inclusive policies and actions.
- International organisations initially offered limited guidance on actions for vulnerable learners and inclusive education.

3.2. Initial European reactions and responses

3.2.1. Emergency measures for disadvantaged learners and inclusive education

At the European level, countries implemented several emergency measures in response to COVID-19 as early as March 2020 (Croatian Presidency of the Council of the European Union, 2020). These included initial actions for learners with special educational needs and inclusive education. Analysis of the available evidence allows a summary of responses:

- Psychological counselling for vulnerable learners, parents and/or teachers (Austria, Cyprus, Finland, Ireland, Latvia, Portugal, Slovakia)
- Measures to compensate for learning disadvantages: lack of technological infrastructure, language barriers, skills gaps, learning support; specific financial resources for computers and digital material for learners from socio-economically disadvantaged backgrounds (Austria, Croatia, Cyprus, Czech Republic, Denmark,



Germany, Italy, Lithuania, Luxembourg, Netherlands, Portugal, Romania, Slovakia, Slovenia)

- Measures for social support and to reduce the social gap: free meals for disadvantaged children (Finland, Hungary, Lithuania, Portugal)
- Schools and social care centres open or accessible to learners with special educational needs (Belgium, Netherlands, Portugal, Slovenia)
- Socio-educational services and home services for learners with disabilities and special educational needs (Italy, Luxembourg, Portugal)
- Recommendation for teachers to adapt pedagogies, design online classes that were accessible to learners with special educational needs, or improve communication with families (Croatia, Cyprus, Denmark, Estonia)
- Guidelines on teaching disadvantaged children; support and counselling on inclusive education (Slovakia).

However, these initial measures were not very structured and focused on local or national necessities. International collaboration or reference to international standards or recommendations was absent.

3.2.2. A common direction for education and democracy

In October 2020, Ministers of Education in EU Member States reaffirmed their intention to reinforce the right to education in times of crisis. They showed a common direction in education and democracy to face the COVID-19 crisis.

The right to education is enshrined in the European Convention on Human Rights (Protocol 1, Article 2) and in articles 7, 10, 15 and 17 of the European Social Charter (Greek Chairmanship of the Committee of Ministers, 2020). The right to education is directly linked to quality education, which should seek to be inclusive (Council of Europe, 2012). Education appears as support for the continuity and unity of democracies in the face of various crises. Investment in education is therefore seen as investment in democracy at a European level and a recognition of human dignity and human rights (Council of Europe, 2020a).

Education is the basis of tomorrow's societies. The Council of Europe member states:

Further reaffirm the determination not to allow the health crisis to turn into a crisis of democracy, and underline that education is key to this endeavour to make our societies sustainable and resilient (Greek Chairmanship of the Committee of Ministers, 2020, p. 2).

They also acknowledge common patterns of action and the intervention of various stakeholders across Europe to respond to the new crisis. European policy-makers:

Express gratitude to and appreciation of the way in which education systems and institutions as well as teachers, administrative and other education staff, leaders of schools and higher education institutions, students, and parents responded to the COVID-19 pandemic that hit Europe in the first half of 2020.



Recognize that the speedy response was all the more impressive as the COVID-19 pandemic presented an emergency situation that could not be foreseen and for which reliable plans could not be laid (ibid.).

Acknowledgment of longstanding weaknesses in education systems and willingness to draw lessons from past experiences are the basis for common future action. This action includes:

 attention to teacher education and openness to participation of various stakeholders:

In this context, we underline the importance of integrating the lessons of the COVID-19 pandemic in the pre- and in-service education of our teachers, taking into consideration the voice of parents and families as well as of civil society (ibid., p. 4).

adopting whole-school approaches:

Our response to the COVID-19 pandemic will encompass whole school, whole institution, and whole system approaches. It will emphasize the need for cooperation between all stakeholders as well as across and between countries and education systems. We recognize, in particular, the crucial role of teachers (ibid.).

 acknowledging co-operation between education institutions, public authorities and private providers to transform online and blended education towards more stable forms of learning.

A focus on social inclusion and education for learners with special educational needs is a key component of future quality education:

This recommendation underlines that social inclusion must be an integral part of our understanding of quality education. Ensuring equal opportunities to education requires a systemic response within our education systems.

The COVID crisis underscores the need for education systems and institutions to reassess and review the way in which they teach, and students learn, including how they provide education for students with special needs. While this will require them to take account of the experience of the immediate COVID response, permanent adaptations cannot copy emergency measures (ibid., p. 6)

The Council of Europe is asked to:

Develop guidelines and materials that will help education systems and institutions ensure the right to quality education also for vulnerable students, adapted to both the 'new normal' and preparing for future crises.

In cooperation with member States, ensure greater visibility for key messages on inclusive, plurilingual and intercultural education for decision-makers,



school leaders, teacher educators and teachers from the Council's rich acquis in this area, including through the <u>platform</u> of resources and references for plurilingual and intercultural education (ibid., p. 10).

Action to mitigate COVID-19's impact on education and democratic citizenship includes *A Roadmap for Action on the Council of Europe education response to COVID-19* from October 2020 (Council of Europe, 2020a).

Following the initial COVID-19 outbreak, policies and practices related to quality inclusive education remain a major component of the Council of Europe 'Better Education for Better Democracies' education programme. Inclusion is mentioned next to other key notions such as the 'right to education', 'democracy' and 'sustainable and resilient societies'.

The Education programme is the Council of Europe's main instrument for ensuring the right to education, as enshrined in the European Convention on Human Rights and the European Social Charter, and for developing a culture of democracy, human rights and the rule of law through education. In addition, the Education programme contributes to the implementation of conventions and the work of Council of Europe bodies in other areas in so far as these address education issues (ibid., p. 29).

In the Roadmap, the Council of Europe defines five directions for future action:

- Strengthening democracy through education
- Innovating the ways in which we learn and teach
- Assessing and recognising qualifications
- Working with others
- Ensuring the right to education for the most vulnerable students (Council of Europe, 2020a).

In particular, 'ensuring the right to education for the most vulnerable students' focuses on actions for quality and inclusive education:

- supporting member States in developing and implementing inclusive education policies that ensure quality education provision to all learners, on-line and in person;
- continuing its holistic approach, combining targeted support to vulnerable groups while convincing the 'majority' of the benefits of inclusiveness and diversity in schools and communities;
- developing and delivering teacher training programmes to enhance teachers' competences in child and youth-centred teaching and learning;
- developing programmes and tools for parents to enable them to provide education support to their children;



- promoting the integration of digital citizenship education and media literacy in school curricula;
- facilitating the sharing of best practices in the field of inclusive education through its inter-governmental bodies and the co-operation and capacity building programme (ibid., p. 26).

3.2.3. Key points

- Unstructured measures for vulnerable learners and inclusive education were available across European countries at the beginning of the crisis.
- European collaboration on a common response for inclusive education could not be identified.
- There is a common willingness to learn from past experience to shape future quality education, especially in the case of social inclusion and learners with special educational needs.
- European resources and guidelines for action aim to mitigate COVID-19's impact on education and democratic citizenship, therefore recognising the right to education for all learners regardless of their levels of need and vulnerability.

3.3. Challenges and opportunities for the European Union

3.3.1. Redefining policy co-operation at European level

The Council of the European Union underlines collaboration between Member States in the exceptional circumstances of the COVID-19 pandemic:

... there has been a remarkable collaboration between Member States, who have shown solidarity and mutual support in a time of crisis by regularly exchanging information on their national situations, challenges, envisaged or already implemented measures and plans, as well as, where possible, openly sharing educational resources (2020, p. 3).

The EU, however, only plays a 'supporting and coordinating role' for policy change; 'education policy is the exclusive competence of the Member States' (European Parliament, 2020, p. 3).

European agencies and networks followed the developments related to COVID-19's impact on education in European countries. In addition, 'video-conferences between education ministers have enabled them to discuss the issues and learn from each other's best practices' (Bassot, 2020, p. 8).

Toolkits and support for policy co-operation have been encouraged and shared at European level (Council of the European Union, 2020). A system of more structured 'exchange of good practices' for policy-makers in education, with a specific focus on inclusive and equitable practices, will be a next step (Council of Europe, 2020b).



A unified system to monitor the efficiency of various policy measures and COVID-19's impact on learners is also needed. As such, and with a focus on digital learning, the European Parliament:

Calls on the Commission to collect, assess and publish data from across the Member States on the pandemic's impact on learners' participation in distance education, with a particular focus on where they could not participate due to a lack of digital means; calls on the Commission, furthermore, to collect data on teachers' digital skills across Member States (2020, p. 4).

These considerations refer to education in general; policy co-operation in the area of inclusive education could not be specifically identified in the analysed European documents.

3.3.2. Limited policy actions for disadvantaged learners

In October 2020, discrepancies in national responses to COVID-19 crisis at EU level were still evident. Inequalities affected disadvantaged learners at several levels:

... up to 32 % of pupils without access to education for several months in some Member States; ... for many learners, this lack of access stemmed from an absence of digital equipment, inadequate digital skills or pre-existing disadvantage; ... even where learners had access to digital education, they still often had to learn without teacher, peer or home support and sometimes in an unstable home environment (European Parliament, 2020, p. 2).

The Education and Training Monitor 2020 (European Commission, 2020a) does not provide direct analysis of the link between COVID-19 and inclusion or inclusive education strategies. However, an analysis of the Monitor's country evidence on general education highlights COVID-19's impact on disadvantaged learners across Europe. Analysis of the evidence available for this review shows that:

- The COVID-19 crisis increased the education gap between learners from wealthier backgrounds and those from disadvantaged or migrant backgrounds (Austria, Bulgaria, Cyprus, Czech Republic, Denmark, France, Germany, Greece, Italy, Lithuania, Malta, Romania, Spain).
- Improving access to digital skills, technologies and education became an urgent need or a priority for most countries and learners. European countries had more or less rapid responses to the need to develop digital education (Austria, Belgium, Croatia, Denmark, Estonia, Finland, Germany, Greece, Ireland, Italy, Latvia, Luxembourg, Malta, Poland, Portugal, Romania, Slovakia, Slovenia, Spain).
- The COVID-19 crisis delayed on-going legislative initiatives and curriculum reforms for learners with special educational needs (Belgium, Malta).

3.3.3. Opportunities for the European Union

Some sources consider that COVID-19's impact on education can become an opportunity for the EU. Immediate action to respond to COVID-19 at European level could be systematised as shown in Table 1.



Table 1. The impact of the pandemic on education (Bassot, 2020, pp. 8–9)

(Potential) impact	Opportunity for the EU	
(Partial) Closure of early childhood education, schools, universities and vocational education and training institutes, with some gradual and partial reopening later. Use of remote tools to compensate	Update aspects of formal education to 21st century. Develop more mature use of technology, review curricula and bridge non-formal, informal and formal learning within life-wide and lifelong frameworks. Protect apprentices and trainees in VET. Develop emergency remote teaching models.	
Suspension of on-site adult education, with some providers potentially unable to survive the impact of closure	Develop support for adult educators and providers and a safety net for small-scale, grassroots adult education initiatives that reach the disadvantaged.	
Suspension of face-to-face guidance and counselling in many Member States	Improve easy, up-to-date information for services on skills needs, education and employment opportunities. Train staff in alternative ways to reach end-users.	
Special measures to reorganise exams and admit students to the next level	Reflect on the adequacy of current models of assessment and reconsider role of continuous and formative assessments.	
Obligation on parents or guardians of young children to provide non-stop care and support the continuation of lessons from home	Build on home-school communication and cooperation. Support learning by low-skilled parents, and resources to give parents the ability to support children's education and internet safety, while maintaining their own resilience.	
Obligation upon inadequately trained teachers to improvise a sudden shift to online teaching, exposing them to increased stress and even censure	Provide teachers with training options to add online tools to their repertoires. Encourage teachers' networks to reflect on their practices together and share experiences and resources. Consider providing technical support staff.	
Students' lost access to resources such as library equipment, computers, school meals and gyms	Ensure access to safe public recreational spaces for children; access to books and digital tools at home; make schools more permeable to the local community.	
Effective remote learning depended on family resources and measures for special educational needs, increasing inequalities. Risk of increased numbers of early school leavers and young people not in education or training (NEETs)	Focus on a child guarantee to combat poverty. Place greater emphasis on study skills, digital literacy and critical thinking skills. Design tools for special educational needs. Offer greater autonomy to students who struggled at school but thrived on remote learning. Strengthen policies to lessen early leaver and NEET numbers.	
Initial stranding of mobility students	Develop protocols for emergencies.	



3.3.3. Key points

- Policy actions for disadvantaged learners were not always co-ordinated.
- The COVID-19 crisis delayed legislative initiatives and curriculum reforms for learners with special educational needs.
- Common responses across European countries show an initial concern with reducing the education gap for disadvantaged learners and improving access to digital technologies and skills.
- The crisis can be transformed into an opportunity to examine ineffective policy measures.

3.4. European Union contributions to support equity in education and combat new vulnerabilities

COVID-19 can be seen as an opportunity to build back a better world for children and future generations (WHO-UNICEF-Lancet Commissioners, 2020). The European Parliament considers that:

... the primary lesson to be learnt from the crisis is that inclusion and equal opportunities, both in terms of access and quality, must be placed at the heart of the Union's future education and training policies (2020, p. 5).

Several European measures and initiatives work to support equity in education and to combat new vulnerabilities.

3.4.1. European Union programmes to support investment in education and training

As the pandemic points to another financial crisis, 'greater coherence between the invitation to investment in education and optimisation measures' is needed (Bassot, 2020, p. 9). Tools to cover gaps in education at a general level within the EU include the European Social Fund Plus (which will become ESF+ under the 2021–2027 multi-annual financial framework), the European Investment Fund, Digital Europe, the Connecting Europe Facility, Horizon Europe and Erasmus+.

The European Commission's Digital Education Action Plan 2021–2027, adopted in late 2020, prioritises education and training for the digital age (European Commission, 2020b). It aims to address the challenges brought by the COVID-19 crisis and on-going digital transformations in education (European Distance and E-Learning Network, 2020).

At the same time, the Council of the European Union invites the European Commission to increase attention to inclusion and equal opportunities in its upcoming strategies:

When developing the European Education Area, together with the post-2020 strategic framework for European cooperation in education and training, the new Digital Education Action Plan and the updated Skills Agenda, take into account the lessons learnt from the COVID-19 crisis regarding the digital transformation of education and training systems, paying special attention to quality and societal challenges such as ensuring inclusion, providing equal



opportunities, promoting cohesion and combating disinformation (Council of the European Union, 2020, p. 11).

3.4.2. European Child Guarantee

In 2015, the European Parliament called for a child guarantee to tackle child poverty in the EU in a sustainable way and to strengthen the relationship between education and health (Chircop, 2020). As a result, the European Commission:

... announced a proposal for a Council Recommendation on the European Child Guarantee for 2021, which will be one measure of the comprehensive EU Strategy on the Rights of the Child and the implementation of the European Pillar of Social Rights, and will form part of the Commission's intention for a corresponding Action Plan to be presented by early 2021 (Bundesministerium für Familie, Senioren, Frauen und Jugend, 2020, p. 4).

The Child Guarantee will act on access to free healthcare, free education, affordable early childhood education and care, decent housing and adequate nutrition for children at risk of poverty and social exclusion.

3.4.3. Key points

- Greater coherence is needed at European level between investment in education and co-ordinated, non-overlapping measures.
- Digital education became a priority in the new EU education programme.
- The EU started to develop new frameworks to combat child poverty and to promote free education for all learners.



4. GOVERNANCE OF INCLUSIVE EDUCATION: LESSONS LEARNED FOR NECESSARY FUTURE CHANGES

While section 3 explored the transnational level of the COVID-19 crisis in education, this section focuses on national dimensions of responses and lessons learned by authorities and other stakeholders. First, it discusses the new forms of assessment and accountability in times of crisis. Second, it redefines funding solutions in the COVID-19 crisis and highlights the urgent need to support vulnerable learners. Third, it introduces possible support for schools from public authorities. Finally, the need for cross-sectoral collaboration provides a major response to build back school systems more successfully.

4.1. COVID-19 and education: assessment and accountability in times of crisis

Governments' actions at the beginning of the COVID-19 crisis were perceived as very limited, as an OECD survey from March 2020 states:

When asked what has the government or network of schools done to date to support the ongoing academic instruction of students, a large percentage indicate 'nothing', followed by providing encouragement to schools to use online resources (Reimers and Schleicher, 2020, p. 11).

However, later European documents and reports showed a high level of cross-national action by European governments to respond to the new crisis in education for all learners (Council of the European Union, 2020).

Assessment of the situation was a priority for multiple stakeholders, both at international and national levels. Assessment initiatives aimed to inform policy, practice and research. Surveys of educational needs and policy responses took place broadly. However, no systematised overview – such as a database – of these surveys is available at the European level and sometimes not even at country level. Some surveys focusing entirely on inclusive education and practices were identified and organised in a list for the current review (see section 10.2: List of surveys on COVID-19's impact on education).

4.1.1. National assessment of post-COVID-19 education in Europe

Several national quantitative surveys or qualitative studies about COVID-19's impact on (inclusive) education systems and learners in various European countries were identified (the <u>Bibliography</u> and <u>list of surveys</u> provide a complete list). They were initiated by academic representatives, disabled people's organisations and civil society organisations. They aimed to show the perspectives of various stakeholders, such as parents, learners, school staff and school leaders, but also to collect evidence and encourage data-informed discussions between families, schools and authorities. In some cases, education authorities re-used the survey findings for further action (Huber and Helm, 2020).



Evidence of authorities' accountability measures for education in general and inclusive education in particular after COVID-19 is still limited or inconsistent (Joynes and Gibbs, 2020; UNESCO, 2021a). Education authorities in Central and Eastern Europe have, for instance, tried to estimate learners' participation in remote learning. They used learning platforms and registration systems for families, learners and teachers (e.g. in Montenegro) and telephone interviews with schools (e.g. in Czech Republic and Estonia).

Surveys and reports from disabled people's organisations, however, aim to make authorities more accountable for their actions related to inclusive education and other measures for vulnerable learners (Alliance for Inclusive Education, 2020; Inclusion Europe, 2020).

The evidence collected on lessons learned in education since the beginning of COVID-19 is based on **progress** and **action**, rather than on **impact** and **outcome** of the new policies and education strategies (Joynes and Gibbs, 2020).

International organisations provided support to collect national data to help policy-makers in their decisions about education in general.

4.1.2. International assessment and accountability of post-COVID-19 education

International organisations developed multiple surveys to assess the needs to address during the COVID-19 pandemic. These surveys also provided the first answers and evidence to use in response strategies to the crisis.

For instance, in March 2020, the OECD surveyed 98 countries about their general education. Questions about disadvantaged learners and schools were limited. Several recommendations referred to measures that could help a rapid response to the crisis in education:

- Create a committee representing multiple stakeholders, including teachers, families and learners
- Define the main principles that will guide time and funding resources
- Co-ordinate with health authorities
- Re-prioritise curriculum goals
- Identify adequate means of education delivery and of communication or collaboration among learners and teachers
- (Re)define teachers' roles and support to learners, in particular the most adapted support to vulnerable learners
- Create mechanisms to support teachers' development
- Define appropriate learner assessment
- Assist schools in their communication with learners and families
- Develop a communication plan about the new education strategies and their dissemination through various channels (Reimers and Schleicher, 2020).



The UN prepared multiple <u>policy briefs</u>. Education is one of the thematic areas, following the idea that 'Solid science, reliable data, and analysis are critical for policy- and decision-making, especially for the tough choices required during a pandemic' (United Nations, 2020a, p. 2).

Additionally, the UNESCO Institute for Statistics initiated partnerships with UN Member States to collect new data to measure COVID-19's impact in the short, medium and long term. This data can help governments minimise the impact of the crisis on national education systems (Committee for the Coordination of Statistical Activities, 2020).

Several online knowledge hubs assessed COVID-19's impact on education worldwide and provided useful resources for future action:

- UNESCO COVID-19 Education Response resource hub
- UNICEF COVID-19 resource hub
- Inter-agency Network for Education in Emergencies (INEE) COVID-19 resource hub
- ReliefWeb COVID-19 resource hub
- Global Education Monitoring (GEM) Report World Education blog
- Wikipedia page on the coronavirus impact on education
- World Bank COVID-19 Remote Learning resource hub.

4.1.3. Key points

- Evidence is available on the increased need to evaluate the impact of COVID-19 to inform policy decisions. Evaluation of COVID-19's impact is also needed to monitor the right to education for vulnerable learners.
- A systematic overview of country survey results is not available.
- Evidence on authorities' accountability measures for education in general and inclusive education in particular after COVID-19 is still limited.
- Surveys by civil society organisations make public authorities more accountable for their (lack of) action.
- International organisations offer rich resources, evidence for action and policy recommendations for European countries.

4.2. Financing and provision

4.2.1. Limited evidence on COVID-19's impact on inclusive education funding

The identified documents mainly cover non-European countries and education funding in general.

... government spending growth will also slow and, in some cases, turn negative. ... In ... high-income countries, education spending is forecast to decline in real terms along with overall government spending (Al-Samarrai, 2020, p. 4).



Forecasts predict that the pandemic will lead to slower growth in government spending in the coming year. If the share of government spending devoted to education were to remain unchanged, education spending would continue to grow but at significantly lower rates than before the pandemic (OECD, 2020a, p. 311).

A predicted cut in public expenditure in education will inevitably affect education, including inclusive education. In September 2020, UNESCO estimated that at least US\$ 210 billion would be cut from education budgets worldwide in 2021 (Right to Education Initiative, 2020).

The COVID-19 crisis has also increased the cost of achieving SDG 4 by 2030:

There is a US\$148 billion annual financing gap in low- and lower-middle-income countries to achieve SDG 4 from now until 2030. Additional costs due to COVID-19 related school closures risk increasing this financing gap by up to one-third, or US\$30 to US\$45 billion. But investing now in remedial and re-enrolment programmes could reduce this additional cost by as much as 75% (UNESCO, 2020b, p. 1).

The documents collected for this literature review scarcely mention COVID-19's impact on inclusive education funding. It is common knowledge that education budgets will 'grow but at lower rates'. The austerity policies and measures that followed the 2008 financial crash cannot be considered possible solutions for the future, as they led to significant health and social crises in Europe (WHO-UNICEF-Lancet Commissioners, 2020).

4.2.2. Emergency funding measures at European level

Some emergency funding measures were available in European countries early in the COVID-19 crisis. Countries faced difficult decisions on how to allocate education support, especially in Central and Eastern Europe (UNESCO, 2021a). The Slovenian government, for instance, decided to:

- support vulnerable tertiary students without pension and disability insurance;
- co-finance the publication of bilingual textbooks and e-materials in minority languages (Italian, Hungarian and Roma);
- ensure access to distance education for disadvantaged learners (European Commission, 2020a).

The Italian government increased the emergency education funds to include vulnerable groups in distance learning activities. Extended contracts with third sector associations ensured socio-educational services and home support for learners with disabilities (Croatian Presidency of the Council of the European Union, 2020).

4.2.3. Recommendations from civil society organisations

Major international non-governmental organisations (NGOs), such as Oxfam, Save the Children, Plan International and World Vision, expressed their fears of amplified education gaps due to lack of funding. Progress made towards achieving SDG 4 before the COVID-19 crisis was at risk of being undone (Right to Education Initiative, 2020).



A call to action from 190 civil society organisations made recommendations about the funding of general education. However, it also stressed the need to 'build back better' and to ensure that education budgets prioritise inclusion, equity and transformation:

Covid-19 could exacerbate these existing inequalities or could be used as a pivotal turning point where a gender and inclusion lens is applied to all aspects of education, including planning, budgeting and expenditure – to advance equality and inclusion in access, retention and achievement at all levels. A focus on equity and inclusion is central to effective education spending (ActionAid, 2020).

Other disadvantaged learner groups need funding support, including girls and learners from the most disadvantaged communities. In relation to this, Save the Children issued a recommendation to policy-makers:

Agree, implement and fund a global COVID-19 education action plan, ensuring that a coordinated global education response keeps learning alive, supports every child to return to school when it is safe to do so, and builds back better and more resilient education systems. The plan should be available on a public website, with progress tracked and reported on regularly.

... National governments should produce and implement fully funded, national COVID-19 education response and recovery plans, with targeted action to ensure that girls and the most marginalised children are able to keep learning through distance learning initiatives and return to school (Gordon and Burgess, 2020, pp. 9–10).

Funding is not enough; increased transparency and accountability are also key to more efficient spending for learners from disadvantaged communities:

Civil society actors can play a crucial role in tracking budgets, making sure that money allocated arrives and decisions are made transparently at the appropriate level. Strengthening civil society voices – and the political space for those voices to be heard – is essential (ActionAid, 2020, p. 3).

The COVID-19 crisis is seen as an opportunity for public-private partnerships and to use international resources and contributions to 'expand rather than displace sustainable domestic financing' (ibid.).

4.2.4. Future scenarios on funding for disadvantaged learners

The Global Education Monitoring (GEM) Report (UNESCO, 2020b) encourages governments to take early action and therefore follow a preventive approach. It suggests governments and education ministries could address four main dimensions in their funding decisions linked to schools' physical re-opening. This is especially the case in low-and lower-middle-income countries where the need to open schools is greater than in other countries worldwide. By adopting early funding and investment in education for the most disadvantaged learners and schools, worse outcomes can be avoided later.



The four dimensions are:

- Invest in **remediation**. Remediation can address learning loss, particularly in the case of disadvantaged learners. However, costs for remedial classes are:
 - ... assumed to be five times higher than the general unit cost of remediation. Remediation costs therefore depend on the length of school closures and the share of marginalized students involved (ibid., p. 4).
- Design and implement re-enrolment strategies for learners at risk of drop-out. This
 entails costs for 'national campaigns and incentives targeted at encouraging
 marginalized students to return to school', but also funding for teachers to
 'maintain the pupil/teacher ratio at target levels' (ibid.).
- Fund **second-chance education solutions**, in line with SDG 4 aims, particularly for out-of-school learners. However, encouraging efforts to re-enrol learners at risk of drop-out remains the priority.
- Invest in school infrastructure, in line with new public health recommendations.

Other external factors can affect possible funding, such as epidemiological developments that keep schools closed for longer periods and the economic impact of lockdowns and further recessions. However, ministries of education have less control over these things (UNESCO, 2020b).

4.2.5. Key points

- There is limited evidence on COVID-19's impact on the funding of inclusive education.
- Austerity policies are not recommended for inclusive education funding.
- Emergency funding measures for disadvantaged learners were available in a limited number of European countries.
- Civil society organisations formulated recommendations to ensure education budgets that prioritise inclusion, equity and transformation.
- Funding is not enough; increased transparency and accountability are key to more efficient spending for disadvantaged learners.
- The crisis could be an opportunity for new public-private partnerships that support more sustainable domestic financing.
- Early funding and investment in education can prevent further inequalities.
- Funding for vulnerable learners could be made available at four levels: remediation, re-enrolment, second-chance solutions and school infrastructure.



4.3. Helping schools to mitigate inequalities

4.3.1. Various guidelines

Most European countries took immediate measures to ensure continuity of teaching and learning after the physical school closures. Authorities issued guidelines and recommendations to schools, emphasising access to education content for all learners (UNESCO, 2021a). They also developed national websites for online and distance learning during school closures (School Education Gateway, 2020).

Ministries of education worldwide received some criticism at the beginning of the COVID-19 crisis, as the initial guidelines for education in general were not 'anchored in the realities of schools' (Reimers and Schleicher, 2020b, p. 11). Rather, successful actions were bottom-up initiatives, as shown in Hungary in March 2020:

Teaching has been shifted to digital. The government tries to give support for this to schools/teachers but most of the initiatives seem to be bottom up. One witnesses a remarkable dynamism and activity in many schools (ibid., p. 12).

Many recommendations to schools for education processes as a whole, without a specific focus on inclusive education, included:

- the need to develop a 'plan for continuity of operations', with authorities providing examples of other school plans;
- identifying alternative channels for food distribution, if available;
- developing 'alternative forms of provision', as in the case of mental health supports;
- developing daily communication with each learner, teacher and school staff member (ibid., pp. 5–6).

There is limited information on public guidelines, online resources or websites related to inclusive education.

In Portugal, the authorities reacted a few days after the school closures by adopting and implementing several measures: a website to support schools, guidance for distance learning, organisation of school support teams and free meal distribution. Initiatives also targeted at-risk learners, such as:

- an awareness and prevention campaign for Roma communities: 'For You, For All, Stay at Home!';
- general guidelines to protect children and young people in COVID-19 times;
- guidelines for good practice in online education in an emergency context for deaf learners;
- a strategy to monitor and support vulnerable learners, which was put in place after schools re-opened (OECD, 2020b).



4.3.2. School closure and re-opening strategies

From mid-March until May or June 2020, all European countries except Sweden suspended in-person contact in school at primary, secondary and tertiary levels (Dumbrava, 2020). Most countries worldwide had government directives forbidding learners to attend in-person classes.

School closure has a direct impact on teachers, learners and their families. However, it also has major economic and societal consequences that will become clear in time. In public discourse and debates, several issues were linked to school closure. These included student debt, digital learning, internet access, food and accommodation, childcare, healthcare, mental and emotional well-being, and services for learners with special educational needs and/or disabilities. Children and young people from disadvantaged backgrounds were affected to a higher degree (UNESCO, 2020a).

According to the analysed documents, equity and inclusion should be taken into account in re-opening strategies that include hybrid models and intermittent school closures. Placing vulnerable learners in a hybrid strategy (e.g. by prioritising them for in-person school attendance while other learners attend only for core subjects) is essential (OECD, 2020c).

Drop-out rates tend to rise after periods of school closure, according to lessons learned from previous health crises, such as Ebola. Refugee girls are at higher risk of drop-out, as they have to assume new roles as carers in their families (Chircop, 2020). Vulnerable learners may never return to the education system, which results in 'reduced overall human capital for the economies of the countries in which they work and live' (United Nations, 2020c, p. 15).

In Belgium (Flemish Community), various surveys indicated that the learners who were most difficult to reach during the school closures, despite efforts, included socio-economically disadvantaged and vulnerable learners and learners who speak another language at home. COVID-19 disruptions also affected the educational performance of learners with migrant backgrounds in Denmark (European Commission, 2020a). In Norway, schools stayed open to vulnerable learners during the first lockdown in March 2020 (OECD, 2020b).

Re-opening education settings can be an opportunity to increase collaboration between education ministries and other stakeholders. In Malta, the ministry and stakeholders discussed a framework for re-opening schools. The framework examines 'health and safety, information, development and training, the need for quality education, the need for education to reach every student, and the necessary financing' (European Commission, 2020a, p. 224).

UNESCO, UNICEF, the World Bank, World Food Programme and the UN High Commissioner for Refugees (UNHCR) published a *Framework for reopening schools* (2020a) and added a supplement in September 2020 (2020b). The *Framework* sets out six key dimensions to assess school readiness for re-opening and to inform further planning: policy, financing, safe operations, learning, reaching the most marginalised, and well-being or protection. Policy considerations and financial requirements are essential to support these dimensions (ibid., 2020a).



Lessons started to emerge on what worked and what did not in school re-opening worldwide. Some referred to learners' well-being and protection and measures for the most marginalised learners:

- Early and regular communication and support to teachers, parents, and students can help address concerns, surface innovations, and ensure a safe, widely accepted reopening
- (Re-)establishing regular and safe delivery of services such as vaccines and school feeding can help bring children back to school and limit secondary impacts of pandemic.
- Additional education funding can effectively be targeted to schools and communities hit hardest.
- Specific measures can help to support girls' and other vulnerable groups' return to school.
- Critical communications and outreach can be diversified by making them available in relevant languages and accessible formats.
- Specific provisions are needed for vulnerable staff and students (ibid., 2020b, pp. 4–5).

4.3.3. Support for teachers

The collected documents often mentioned support measures for schools, learners and their families. Teachers were expected to work to ensure education continuity (UNESCO, 2021a) and to adapt to the 'new normal'. However, there is limited evidence on COVID-19's impact on teachers and their well-being.

Teachers from all forms of education expressed concerns about their lack of knowledge and skills to implement online learning. They also had to deal with restricted access to technology and official guidelines with limited relevance to the new challenges (UNESCO, 2020c).

A study of about 1,000 primary school teachers in Poland found that 85% had no experience with distance learning before the pandemic outbreak, and 52% reported some difficulty using digital tools. In addition, 36% indicated that lack of equipment among students impeded distance education (UNESCO, 2021a, p. 146).

In particular cases at the beginning of the crisis, teachers were supported in their practices with:

- 'material banks and lists for teaching', e.g. in Finland (Reimers and Schleicher, 2020, p. 14);
- information or guidelines produced by universities on how to use specific digital tools, e.g. in France;



- direct advice on the websites of the ministries of education (ibid.);
- 'in-service training or online consultations on remote learning', e.g. in Lithuania (UNESCO, 2021a, p. 144).

Possible challenges identified in the collected material are that only teachers who are already motivated would use the resources and that teachers' increased workload could be a barrier to other activities. In particular cases, teachers provided informal training to each other by sharing their practices (UNESCO, 2021a). Evidence focused on inclusive education dimensions was not identified in the material collected for this review.

Selected recommendations from international organisations point to more specific measures for teachers' well-being (European Parliament, 2020). They also highlight the need for training on inclusive education and accessibility in teachers' education. For instance, Save the Children recommends providing:

... continuous professional development and financial support to teachers including support for wellbeing, mental health and psychosocial support so that they can adapt to new circumstances in schools as part of the frontline response and recovery, and to adapt to supporting distance learning when some schools remain closed or have to close again (Gordon and Burgess, 2020, p. 10).

The OECD urges support for and by teachers in education in general and in inclusive education in particular. Stronger support from education authorities is needed, along with more accessible resources (OECD, 2020c).

4.3.4. Support for vulnerable and disadvantaged schools

Prior to COVID-19, socio-economically disadvantaged schools received less guidance and could apply fewer practices linked to the use of digital devices in teaching and learning than more advantaged schools (Reimers and Schleicher, 2020).

Self-directedness is also important to cope with the crisis:

An important lesson from the COVID-19 pandemic is that schools and students who are more self-directed in their own teaching and learning often do better in both normal and crisis situations, than those who haven't learned to do so. Self-directedness strengthens student engagement, brings about authentic learning and helps schools respond better to emerging inequalities (Sahlberg, 2021, p. 7).

Several measures in European countries supported vulnerable and disadvantaged schools. For instance, Belgium (Flemish Community) provided additional funding (EUR 17 million) to enable socio-economically disadvantaged schools to hire more teachers for individualised support and remediation (European Commission, 2020a).



4.3.5. Key points

- Initial guidelines and resources were not anchored in school realities.
- Limited information is available on public guidelines, online resources or websites related to inclusive education.
- Children and young people from disadvantaged backgrounds were more affected by the school closures.
- There is a risk of increased drop-out rates due to the crisis and school closures.
- Measures, such as keeping schools open for vulnerable learners, might help reduce the number of learners at risk of drop-out.
- A European framework for re-opening schools could not be identified.
- Teachers received limited or no support for the new forms of remote education.
- Evidence of support for teachers to implement inclusive remote learning pedagogies or educational practices could not be identified.
- Some countries provided additional funds for schools in more disadvantaged areas.

4.4. Holistic approach to education and cross-sectoral collaboration

4.4.1. Cross-sectoral collaboration for more adapted responses to the COVID-19 crisis

Several European and international organisations expressed the need for a holistic approach to education in response to the COVID-19 crisis. A holistic approach addresses children and young people's learning, social and emotional needs in an interconnected, rather than a separate, way (Cerna, 2020; WHO-UNICEF-Lancet Commissioners, 2020; Council of Europe, 2020a). To succeed, a holistic approach requires co-ordinated action and collaboration from stakeholders representing various sectors working for children and young people's well-being:

Considerable joint efforts by school leaders, teachers, parents, students, educational and health care professionals, and communities will be needed to (re-)create schools as safe, supportive and inclusive places for all students (Cerna, 2020).

The OECD (2020c) identified policy areas for a holistic approach and examples of country initiatives during school closures:

- Equitable and inclusive access to good learning conditions
- Equitable and inclusive access to digital learning resources, e.g. educational television and radio programmes; free access to online learning platforms
- Information in different languages for learners and their families, e.g. letters to parents about school closures; communication with Roma families with the help of civil society organisations



- Meeting socio-emotional needs, e.g. online counselling services; virtual meetings between newcomers
- Equitable and inclusive access to extra services for vulnerable learners
- Support for and by teachers, e.g. guidelines and free online teaching resources
- Support for immigrant/refugee learners through WhatsApp groups.

Partnerships between stakeholders are essential to provide more holistic responses to the crisis. They rely on both horizontal and vertical collaborations for better governance:

- Collaborations between ministries of health and education
- Collaborations between ministries of education and teachers' unions or civil society organisations
- On-going communication between school staff, learners and their families (OECD, 2020c).

Evidence from several European countries indicates the existence of partnerships that support learners with special educational needs. These partnerships were mainly initiated and co-ordinated by education authorities:

- In Ireland, several categories of stakeholders were involved in decision-making after the emergence of COVID-19, particularly around the topic of secondary-level state examinations. The group also provided guidance on support for learner well-being and minimising the pandemic's impact on vulnerable people (European Commission, 2020a).
- At the beginning of the COVID-19 crisis in March 2020, Luxembourg worked to
 ensure support for children with special educational needs and their families.
 Competence centres provided support for distance learning, while local social
 services monitored vulnerable families (Croatian Presidency of the Council of the
 European Union, 2020).
- In Portugal, collaboration enabled quick contact and remote education for learners in March 2020. However, communication with socio-economically disadvantaged learners was a challenge. The education authorities created a network of partner institutions to meet this challenge (Reimers and Schleicher, 2020).
- In Slovenia, the education authorities provided guidelines and the National Institute for Education offered assistance to learners with special educational needs. The Slovenian Education Network provided 'additional guidance for teachers and professionals supporting individualized education plans' (UNESCO, 2021a, p. 143).
- In Slovakia, the National Institute for Education collaborated with social workers and community centres and 'provided online support to professional staff to help reach Roma students and their parents' (ibid., p. 144).



4.4.2. The COVID-19 crisis as a reason for new partnerships in education

Multi-level and cross-sectoral partnerships appeared or intensified during the COVID-19 crisis. Traditional partnerships in education in general and inclusive education in particular involved communication between schools and local or national education authorities, or communication between schools and health and care professionals.

With the increase in remote and digital learning needs, the COVID-19 crisis showed the importance of partnerships with the private sector for information and communication technology (ICT) infrastructure. These partnerships facilitated access to communications infrastructure and to internet or mobile data, but also to necessary materials for (disadvantaged) schools and learners (Joynes and Gibbs, 2020). Partnerships for pedagogical adaptation and use of new educational technologies still need to be developed, particularly for learners with special educational needs.

4.4.3. Further recommendations for cross-sectoral collaborations

Cross-sectoral collaboration in response to the COVID-19 crisis might represent a main ingredient for 'building back better', especially in education for vulnerable learners.

Several documents recommended collaborations with disabled people's organisations or civil society organisations for common responses to the challenges of the COVID-19 pandemic at a global level (European Disability Forum, 2020; Office of the United Nations High Commissioner for Human Rights, 2020; ActionAid, 2020). The possible support of disabled people's organisations and other civil society organisations could supplement limited public provision.

In one of its reports, for instance, Save the Children recommends paying particular attention to both education and safety aspects:

Implementing specific strategies to support marginalised children, especially children with disabilities to return to school while giving particular attention to safety and protection concerns. This should include:

- Continuous teacher skills development on inclusive education, special pedagogy and accessible learning materials.
- Work in partnership with representative organisations of persons with disabilities (OPDs) to better understand barriers to learning and the reasons for possible drop-out, identification of children with disabilities not being supported in learning from home and creation of parent support groups.
- Disaggregating disability data using UNICEF Child Functioning modules or other approved tools to understand the proportion of learners with disability in order to track dropout rates disaggregated by disability and take follow-up action, as children return to school (Gordon and Burgess, 2020, p. 11).



4.4.4. Key points

- Cross-sectoral intervention, particularly for vulnerable learners, is needed for more adapted responses to the crisis.
- The crisis determined new forms of partnerships in education for disadvantaged learners.
- International organisations recommend prioritising support strategies for marginalised learners.



5. TRANSFORMATION OF EDUCATION

This section seeks to highlight new opportunities and limitations for inclusive education. It does so by identifying specific transformations of education systems during the COVID-19 pandemic. Topics include learning loss and the redefinition of learners' vulnerabilities; transformed curricula, assessment and learner progress; the implications of remote and digital learning for disadvantaged learners; and adapted responses to reduce the new digital divide.

5.1. Acknowledgment of learning loss

5.1.1. Severe learning loss for vulnerable learners

Learning loss can be a consequence of less time spent in learning (Lavy, 2015). Prior to the COVID-19 crisis, studies on the effects of 'summer learning loss' (Cooper, Nye, Charlton, Lindsay and Greathouse, 1996) showed delays in learning compared to continuous study, as well as the loss of previously acquired knowledge and skills:

A review of research on summer learning loss in the United States, demonstrates that during the summer vacation students lose the equivalent of one month of academic year learning, the loss is greater in math than in reading, and the loss increases with grade. The loss is also greater for lower income students (Reimers and Schleicher, 2020, p. 4).

According to the World Bank, approximately 6.8 million children and young people in primary and secondary education are at risk of drop-out (Azevedo, Hasan, Goldemberg, Iqbal and Geven, 2020). UNESCO (2020b) estimated 11 million children worldwide may not return to school after the crisis.

The COVID-19 pandemic is considered to be the most severe disruption to the world's education in history, 'threatening a loss of learning for an entire generation of students, and erasing decades of progress' (European Parliament, 2020, p. 2). During the COVID-19 crisis, learning loss may be substantial for most learners (Müller and Goldenberg, 2020), but particularly severe for vulnerable learners who also lack quality home support or access to extra-curricular activities (Reimers and Schleicher, 2020).

Learners from disadvantaged or migrant backgrounds were hard to reach during the COVID-19 school closures, meaning they could not access available support and provision (European Commission, 2020a). Learning loss is increased for categories of learners already at risk, such as young women and girls living in poverty, those with disabilities and those in rural locations. They are more likely to leave school as they have to provide care and domestic work at home (UN Women, 2020). Consequences of learning loss or learning gaps can affect learners throughout their entire lives (European Commission, 2020c; Reimers and Schleicher, 2020).

At European level, few estimates of vulnerable learners' participation in remote education are available. Information from countries in Central and Eastern Europe shows the negative impact of the COVID-19 crisis on learners with disabilities and Roma learners. For



instance, an estimated 20% of learners with disabilities in Bulgaria 'did not receive additional support needed for education continuity' and about 30% of Roma learners in Croatia did not access remote learning (UNESCO, 2021a, p. 144).

5.1.2. Action to reduce learning gaps and learning loss

Several measures can address learning gaps. The aim is to ensure education continuity by developing alternative strategies and ways of learning and teaching (Reimers and Schleicher, 2020). The OECD suggests, for instance, action on:

- summer schools;
- accelerated education programmes;
- universal design curricula;
- after-school tutoring and study classes;
- informal learning (2020c).

Co-operation between school leaders also appears as a suitable strategy to address school issues during the pandemic. An accessible way to co-operate is by exchanging knowledge on methods that protect educational opportunities (Reimers and Schleicher, 2020).

5.1.3. Key points

- Learning loss risks are higher for already-vulnerable learners than for other categories of learners, as they do not have home support or are hard to reach.
- The consequences of COVID-19 could affect learners for their whole lives.
- Estimates show reduced participation in remote education for learners with disabilities, learners belonging to ethnic minorities and other vulnerable learners.
- Co-ordination, knowledge exchange, remediation measures and universal design curricula could help reduce learning gaps.

5.2. Impact on curricula, assessment and learner progress

5.2.1. The need to redefine curricula, assessment and learner progress

The COVID-19 crisis in education affected traditional pedagogy and assessment practices, as well as learners' progress in particular cases. It caused more infrequent lessons and incomplete curriculum coverage compared to the pre-COVID-19 period (UNESCO, 2020b).

The first worldwide government responses to the crisis focused on the creation or adaptation of online platforms and websites as main teaching and learning resources. Radio broadcasts, podcasts and television programmes allowed learners to follow instruction for specific subjects. However, governments and ministries of education did not establish priorities regarding which curriculum areas to teach first (Reimers and Schleicher, 2020).

Since the beginning of the COVID-19 crisis, alternative approaches and adapted methods of learner assessment have been considered the norm in education, particularly in Central



and Eastern European countries. The transformation of traditional and regular education imposed by the new pandemic conditions brought about more inclusive practices:

- Education authorities encouraged teachers and schools to assess learning achievements from before COVID-19. Formative assessment, however, was a more adapted solution for all learners in the current situation.
- Academic learning was still present, but the need to consider socio-emotional dimensions in learning increased (UNESCO, 2021a).

Evidence from several countries shows inclusive practices being adapted to all learners during the COVID-19 crisis:

The Czech Republic, Estonia and Latvia encouraged teachers to use formative assessment instead of numerical grading and concentrate on providing feedback and psychological support, even if that compromised academic knowledge and curriculum content coverage. End-of-year evaluations were to be based mostly on student performance before remote learning began (ibid., p. 147).

As a less expected outcome, evidence shows that learners who had difficulties in school environments seemed to adapt better to remote learning (Chircop, 2020). In Sweden, pupils with learning issues improved their achievements, possibly due to a more individualised approach and the absence of distractions (European Commission, 2020a).

Definitions of learner educational progress under COVID-19 conditions are dynamic. Available home resources, such as technology or parental support, played a role in learner outcomes in terms of emotional well-being, learning effort and learning achievements. However, in a study of approximately 8,000 learners in Austria, Germany and Switzerland, educational achievements appeared to be more directly related to learners' self-regulation skills and access to quality teaching (Huber and Helm, 2020).

5.2.2. Recommendation for educational practices during COVID-19

In its *Roadmap for Action* against COVID-19's impact on education, the Council of Europe recommends innovation, adaptation and support for policy-makers and schools facing the new realities. The adaptation of existing teaching and learning practices is therefore essential:

In the aftermath of the COVID-19 pandemic, the role of public authorities will comprise a responsibility for ensuring the right to education by inciting schools and institutions to adapt their teaching and learning to new circumstances, by supporting them in this endeavour and by developing the frameworks that will maintain education provision in areas that are particularly affected by the economic and societal consequences of the pandemic (Council of Europe, 2020a, p. 8).



Several international organisations recommend adapting provision and action to the needs of vulnerable learners, as distance learning is even more challenging for them:

- A more engaging curriculum for learners with attention-deficit hyperactivity disorder
- Subtitles and remote live captioning for learners with hearing impairments
- Digital textbooks or special software for learners with dyslexia
- Educational learning materials for the parents and communities of learners with autism
- Teaching assistants to support individual learners with their lessons (Cerna, 2020; Chircop, 2020).

5.2.3. Key points

- Governments worldwide focused on the physical and digital infrastructure of learning. Additional measures are needed to improve the learning process.
- Alternative approaches and methods in learner assessment allowed adaptation to the 'new normal', a term which the analysed literature uses often. The changed conditions brought the emergence of more inclusive practices.
- Adapted provision and action are recommended for vulnerable learners, as distance education is more challenging in their case.

5.3. Remote and distance learning

European countries showed commitment and resilience, while ensuring education continuity through remote learning modalities (UNESCO, 2021a). The COVID-19 crisis allowed previously underexplored resources, such as existing online learning platforms, to be re-used and developed (Bassot, 2020).

5.3.1. Online distance learning as a global solution

Distance learning programmes and open educational materials were among the first recommendations to compensate for the first school closure period in early 2020 (UNESCO, 2020d). New innovative solutions sometimes appeared, and some systems opted for hybrid forms of education – in-person and distance – when possible.

Despite discrepancies due to countries' income and adequate provision (World Bank, 2020), online distance learning has been implemented worldwide (UNESCO, 2020a). Digital learning practices started to be used on a large scale, including low- and high-tech practices.

There is not yet enough evidence and data available to judge the effectiveness of distance and online learning for quality education opportunities. Further research is needed to look 'beyond "access" to remote learning' into what is actually done in terms of teaching and learning practices (Joynes and Gibbs, 2020). Additionally, public consultations showed parents, learners and educators overwhelmed by digital change (European Commission, 2020d).



At a global level, initial results show that the first distance learning solutions failed to reach millions of learners:

... according to one estimate, based on a joint UNESCO-UNICEF-World Bank survey, at least 463 million students in low- and middle-income countries were not reached at all, representing 31% of total students in the countries surveyed. Where distance learning exists in these countries, they reach at most 62% of learners in the case of television and 24% in online delivery (UNESCO, 2020b, p. 3).

All EU Member States implemented remote learning measures to replace ordinary presence in schools. This was initially a temporary procedure (European Commission, 2020c).

Limited evidence is available on measures to allow vulnerable learners to access distance learning in some European countries. For instance, the Croatian Ministry of Education recommended that pedagogical staff in schools design online classes adapted to children with special needs and disabilities (World Bank, 2020). The COVID-19 pandemic revealed disparities in the digital readiness of Irish schools and society: 39% of primary school head teachers reported difficulties in providing distance learning, particularly to vulnerable learners (Burke and Dempsey, 2020). In Italy, most schools were able to implement distance learning in a very short time. However, inclusion and quality education for vulnerable learners remain challenging (European Commission, 2020a).

Initial distance learning solutions were considered 'an imperfect substitute to face-to-face interactions between teachers and learners' (UNESCO, 2020b, p. 3). Despite this, many stakeholders still consider distance and online learning as key factors for transforming post-COVID-19 education. They also call for stronger guidance at the European level (European Commission, 2020d).

5.3.2. International recommendations

Ministries in Council of Europe member states underline the need for inclusive, high-quality education associated with blended and online learning practices. To ensure this, they recommend further investment in adapted infrastructure and in developing learners' and teachers' information technology competences and skills (Greek Chairmanship of the Committee of Ministers, 2020).

Several international organisations recommended adapting digital learning to all learners' needs and working towards more inclusive online education systems (European Disability Forum, 2020; UNICEF, 2020a; 2020b). For example, Save the Children recommended the following:

Enable access to different learning resources, appropriate to the learners' needs and abilities, using different modalities whenever possible.

Provide effective, flexible and inclusive distance learning programmes, and including printed learning materials, with a focus on reaching the most marginalised children (Gordon and Burgess, 2020, p. 10).



The Education Development Trust formulated several recommendations, based on a summary of worldwide country-level responses to the management of school closures in 2020, with a focus on the needs of disadvantaged learners and the role of technology:

- In seeking to provide for disadvantaged students during school closures, governments should begin by systematically auditing the 'digital divide' and design a distance learning regime that is fit for context.
- During the design and implementation of the distance learning regime, governments should prioritise the needs of highly vulnerable, disadvantaged students, such as children with disabilities.
- There is a need to ensure that the distance learning of disadvantaged students is properly monitored and that quality assurance mechanisms give a 'voice' to disadvantaged students and their families.
- Governments should take action to close the household-level technology gap between disadvantaged and more privileged students, while recognising that remote learning requires skilful teaching as well as appropriate technology.
- For disadvantaged students, access to online learning can be enhanced through effective public-private partnerships with technology companies.
- There is a need to providing role clarity for teachers and other professionals so that they understand their responsibilities for ensuring the learning continuity of disadvantaged students.
- Governments should build coalitions with parents or caregivers and non-government organisations to support continuity of learning for disadvantaged students (McAleavy, Joynes, Gibbs and Sims, 2020, pp. 1–3).

UNESCO also made early recommendations for distance learning that focused on inclusion mechanisms. The recommendations were intended for general education systems. They mainly referred to:

- choosing the most adapted technological solutions in accordance with local access
 to internet and teachers' or learners' digital skills, accompanied by the use of
 digital learning platforms, video lessons, massive open online courses (MOOCs),
 radio and TV content;
- including learners with disabilities or from disadvantaged backgrounds in distance learning programmes, by re-distributing digital material that is already available in schools;
- protecting data privacy and data security;



- creating communities to minimise psycho-social challenges learners might face during isolation;
- planning related to learners' study schedules and specific needs;
- supporting teachers and parents to use digital tools;
- implementing blended learning, including both synchronous and asynchronous activities, aiming to avoid using too many applications and platforms;
- developing distance learning rules and monitoring learning processes (UNESCO, 2020e).

5.3.3. Key points

- Distance learning has been implemented worldwide and online learning was deployed on a large scale. Evidence of their effectiveness is not yet available.
- Some European countries supported accessible online learning for learners with special educational needs, but providing quality education for these learners remains challenging.
- Adapting digital learning to all learners' needs is a priority in many international recommendations.

5.4. Addressing the digital divide

5.4.1. Access to digital technologies and materials

Digital tools, online platforms and other educational media, such as television and radio, were the first solutions to ensure education continuity at the beginning of the COVID-19 crisis. However, these first responses did not mention inclusive education and previously disadvantaged learners (Reimers and Schleicher, 2020). Fragmented evidence started to be collected later at European level (UNESCO, 2021a).

Common issues for all learners during school closures were access to ICT, internet connection, proximity to extra services, and access to meals (OECD, 2020b). Vulnerable learners were more affected than others. In Bulgaria, authorities estimated that '11% of students could not engage in learning due to lack of internet access or ICT equipment, in particular students from disadvantaged backgrounds and Roma' (European Commission, 2020a, p. 37).

Learners with disabilities and disabled people's organisations reported difficulties related to online distance education. They mentioned:

... the accessibility of technology and digital educational material; availability of assistive technology; technical support provided to students with disabilities and the teacher competence on disability and accessibility matters (European Commission, 2020e, p. 6).

In specific cases, and pre-COVID-19, learners from disadvantaged backgrounds might have had greater access to computers in their schools. However, inequitable distribution arises



when disadvantaged schools are closed, as their school computers are less likely to be portable (Chircop, 2020).

Only some European countries, such as Malta, provided free internet and computers so that disadvantaged children could continue their studies at home during the COVID-19 lockdown (European Commission, 2020a). In addition – and given their member countries' limited measures – some international organisations, such as the OECD, suggested more policy options for equitable and inclusive access to digital learning resources (OECD, 2020c).

5.4.2. Inclusive digital practices and pedagogies

Policy measures limited to accessing online learning will not be enough to avoid the digital divide. New practices, guidelines and forms of assessment must also be developed:

... online education can support learning for many students but needs to be carefully designed and individualized to not deepen inequality and social divides. The forced move to online learning may have been the catalyst to create a new, more effective hybrid model of educating students in the future. Not one single model for online learning will provide equitable educational opportunities for all and virtual learning cannot be seen as a cheap fix for the ongoing financial crisis in funding education (Kaden, 2020, p. 1).

5.4.3. Key points

- The digital divide particularly affects learners who were already vulnerable before the COVID-19 pandemic. Additional measures are needed to improve accessibility, the availability of assistive technologies and teachers' competences.
- More inclusive practices must be implemented to reduce the digital divide in education.



6. LEARNERS' VOICES AND FAMILIES' EFFORTS IN EDUCATION

This section focuses on three dimensions that can affect learners' progress in education (Ramberg and Watkins, 2020) and therefore make education systems less equitable and inclusive. The first dimension considers parents' transformed role during the pandemic and the policy measures to support them. The second dimension focuses on all learners' rights to education and participation in shaping their future. The third dimension is a major topic that needs future intervention: well-being and mental and emotional health as support for education continuity.

6.1. Parents' transformed role in education

6.1.1. Assessing COVID-19's impact on families

Research by Save the Children in 46 countries included 31,683 parents and caregivers and 13,477 children aged between 11 and 17. It found that:

Two-thirds of parents and caregivers (67%) reported that their children had no contact at all with their teachers. A higher proportion of children with disabilities had contact with teachers than those without disabilities.

Nearly three-quarters of children (74%) were reported by parents and caregivers to have access to some form of learning materials, meaning just over one in four children (26%) did not have access to any learning material at all. Children with disabilities and children of parent/caregiver respondents with disabilities were less likely to have access to learning materials than children with no disabilities or children of parent/caregiver respondents where no disability was reported. ...

Parents and caregivers were more likely to report feeling unable to support the learning of children with disabilities (38%) compared to parents and caregivers of children without disabilities (28%) (Gordon and Burgess, 2020, pp. 7–8).

Most parents had to take a more active role in their children's education, especially in the case of very young learners (Chircop, 2020).

A qualitative study in Kosovo on the level of inclusion of children with disabilities in education before and during the COVID-19 period shows how children's isolation also has a large impact on their parents:

... during the pandemic period, home isolation, lack of socialization, changes in daily routines, and lack of services negatively affected the emotional states of children with disabilities, contributing to parental overload and stress (Hyseni Duraku and Nagavci, 2020, p. 1).



Families expressed concerns about their lack of experience in supporting their children with online distance learning, their access to technology and their economic limitations (UNESCO, 2020c). Families of children with disabilities became even more concerned due to limited – and sometimes completely absent – resources and support during the lockdown (European Disability Forum, 2020; UNESCO, 2021a).

In other cases, the language of instruction differed from the language spoken by families, which was a major barrier. For example, Russian-speaking parents in Estonia could not support their children who attended Estonian-speaking education settings (UNESCO, 2021a).

6.1.2. Recommendations on working with families

Prior to the COVID-19 crisis, collaboration between schools and families was recommended at several levels:

(1) supporting families in assisting their children, as well as helping schools understand family dynamics; (2) providing two-way communication through different channels; (3) encouraging families to volunteer in school activities to support their children; (4) learning at home; (5) involving parents in decision-making; and (6) cooperating with the community through coordinating resources and services among relevant parties (Hyseni Duraku and Nagavci, 2020, p. 34).

Frequent recommendations to support parents' involvement in their children's education were made after COVID-19 appeared. Parents' engagement in home-based education was considered essential, although it was recognised that 'many disadvantaged learners lack effective family support' (Joynes and Gibbs, 2020, p. 3).

In one of its recommendations for education in general, Save the Children stresses the role of interactions and play in education:

Support parents' role in home learning through social and behaviour change communications and mental health and psychosocial support, to enable caregivers to increase interaction and play with children (Gordon and Burgess, 2020, p. 10).

It was also recommended that schools support families and prevent risks related to the use of digital tools and technologies:

Schools should provide guidance to students and families about the safe use of screen time and online tools to preserve student well-being and mental health as well as provide protection from online threats to minors (Reimers and Schleicher, 2020, p. 6).

6.1.3. Evidence of measures encouraging communication between schools and families

The COVID-19 crisis exacerbated existing problems, but in some cases it forced leaders in vulnerable and disadvantaged schools to improve communication with parents. This was the case in the Czech Republic, for example (European Commission, 2020a).



Some other countries, including Ireland and Slovakia, developed guidelines for schools and parents on how to better support their children in education and how to maintain their well-being. The guidelines included developing new routines and ways to explain and discuss COVID-19 with children.

To ensure inclusion, school health messages in Ireland were translated into 17 different languages (Croatian Presidency of the Council of the European Union, 2020).

6.1.4. Key points

- Learners with disabilities and other vulnerable learners received limited support from schools, which increased the pressure on parents.
- Children's isolation contributed to families' vulnerability.
- Parents received limited support, which increased their concerns.
- Increased collaboration between schools and parents is needed to ensure better learning experiences for vulnerable learners.

6.2. Children's rights and disability rights to education

6.2.1. The right to education

The UN Committee on the Rights of the Child (2020) warned that COVID-19 had a great physical, emotional and psychological impact on children and called on countries to protect children's rights. The impact is even greater on already-vulnerable children. Several recommendations referred to:

- reflecting children's interests;
- children's rights to leisure and cultural activities;
- reducing the digital divide;
- access to nutritious food and healthcare;
- support for mental health;
- protecting children in vulnerable situations;
- children's contact with their families;
- children's access to accurate information about COVID-19;
- children's participation in decision-making processes about the pandemic (ibid.).

International organisations, civil society organisations and disabled people's organisations monitored whether the right to education of those with disabilities was being respected during the COVID-19 crisis:

To reduce the impact of disruption in education, some States are adopting remote learning practices. In these cases, however, students with disabilities are facing barriers on account of the absence of required equipment, access to internet, accessible materials and support necessary to permit them to follow online school programs. As a result, many students with disabilities are being



left behind, particularly students with intellectual disabilities (Office of the United Nations High Commissioner for Human Rights, 2020, p. 6).

... too many disabled students who cannot access standard online learning opportunities with minimum adjustments are being denied their basic right to mainstream education (Alliance for Inclusive Education, 2020, p. 7).

In specific cases, judicial action was taken against the lack of accessibility in remote and digital education. For instance, in Croatia, refugee learners and those belonging to national minorities faced language barriers which meant they did not receive appropriate education. National NGOs 'lodged a complaint for violations of the Convention on the Rights of the Child' (UNESCO, 2021a, p. 144). However, evidence from elsewhere in Central and Eastern Europe shows that online content was provided in minority languages, thereby respecting those learners' right to education (UNESCO, 2021a).

6.2.2. Participation in education responses

Children's voices are hardly heard in the educational responses to the crisis:

Amongst our children's participation and rights network colleagues who responded to the survey, 70% could not identify a single COVID-19 related children's participation initiative (local or national).

The other 30% of respondents indicated children's participation activities which were underway, but none had yet led to governmental action, beyond the provision of information to children. Examples gathered to date involve children's participation in:

- Receiving, designing and sharing information
- Contributing ideas to shape services, for example alternative care and inclusive education
- Developing national, organisational and family capacities by critiquing current practice (Centre for Children and Young People's Participation, 2020, pp. 2–3).

Recommendations to listen to all stakeholders' voices in education decisions intensified during the COVID-19 crisis. The Council of Europe recommended more active participation of learners and families, next to other categories of stakeholders, in developing education policy:

The role of stakeholders, including staff, students and parents as well as civil society in the development of education policy and practice must be strengthened and given greater recognition, nationally as well as in local communities (Greek Chairmanship of the Committee of Ministers, 2020, p. 5).



Families and learners could bring new perspectives and determine more adapted solutions to current realities. For instance, an OECD recommendation from spring 2020 underlines the need for a committee or task force to find responses to the new crisis:

To the extent possible ensure those in the task force represent different constituents in the education system or school network and bring important and diverse perspectives to inform their work, for example various departments curriculum [sic], teacher education, information technology, teacher representatives, parent representatives, students, representatives of industry when relevant (Reimers and Schleicher, 2020, p. 5).

International organisations recommend respecting children's rights by listening to their voices and encouraging their participation in public decisions:

Children should understand what is happening and feel that they are taking part in the decisions that are being made in response to the pandemic (Committee on the Rights of the Child, 2020, p. 3).

Country leaders should put child health and wellbeing at the centre of recovery plans, include experts in children's issues in the relevant task forces and legislative working groups, engage their ministries to work together for children, and ask children and adolescents what changes they would like to see (WHO-UNICEF-Lancet Commissioners, 2020).

6.2.3. Key points

- Vulnerable learners face more frequent barriers to their right to education.
- Some measures were taken to respect vulnerable learners' rights, but judicial action from civil society organisations was still needed in specific cases.
- Children's and young people's voices were almost absent when formulating public responses to the crisis.

6.3. Well-being – mental and emotional health

6.3.1. Increased vulnerability for at-risk children

As early as April 2020, international actors were concerned about children's well-being worldwide:

The Committee on the Rights of the Child expresses concern about the situation of children globally, particularly those in situations of vulnerability, due to the effects of the COVID-19 pandemic. Many children are gravely affected physically, emotionally and psychologically, especially in countries that have declared states of emergencies and mandatory lockdowns (Committee on the Rights of the Child, 2020, p. 1).

In a study from the United States in November 2020, 80% of the surveyed learners in mainstream secondary education had experienced some negative impact on their mental



health due to the COVID-19 crisis. Twenty percent said their mental health had significantly worsened (Active Minds, 2020).

COVID-19 increases risk factors for child abuse and neglect (UNICEF, 2020c). Social isolation, stress, family violence, poorer mental health, lack of contact and access to social care services affect the well-being of all children and young people. However, the risks are highest for certain groups of children, such as those who are poor or in out-of-home care, those with disabilities and those from migrant families or from disadvantaged and marginalised backgrounds (OECD, 2020d; Chircop, 2020; Giannini and Grant Lewis, 2020).

Separation from caregivers is a factor for increased vulnerability of children. In past health-related disasters, up to 30% of children separated from their caregivers developed post-traumatic disorders (WHO-UNICEF-Lancet Commissioners, 2020).

School closures also affect children's well-being, as the risk of poor nutrition among children increases (OECD, 2020d). The current crisis has wider social and mental health effects on learners, beyond the academic impact.

6.3.2. Making children's well-being a priority for policy action

The socio-emotional dimension is the most emphasised aspect of well-being in the available documents and recommendations for future action. Learners' socio-emotional well-being depends on measures to counteract both social isolation and one-way forms of knowledge transmission in education, which also suppress interaction between learners and teachers (UNESCO, 2021a).

Policy-makers in education can therefore act or have already acted on learners' well-being in several ways:

- providing individual psycho-social support: for example, in Bulgaria, families, learners and teachers could access a telephone hotline for psychological support (UNESCO, 2021a). In addition:
 - ... some countries put in place special support measures, for example Ireland, Croatia and Malta set up special forms of psychological support to pupils at risk of becoming disengaged (European Commission, 2020c, p. 11);
- adapting the curriculum to learners' needs: 'Belgium (French Community) decided not to provide new learning content to avoid inequalities' (ibid.);
- enabling teacher flexibility:

As children suffer from isolation, exemptions from the obligation to follow the core curriculum would give teachers an opportunity to be flexible and adapt to students' needs (UNESCO, 2021a, p. 148);

- supporting teachers to address both learners' academic needs and well-being: online training, mentoring or coaching, transformation of initial teacher education in the longer term;
- improving communication and co-operation among learners (UNESCO, 2021a).



The OECD recommends prioritising the well-being of vulnerable learners. This could be by providing psychological support, increased social and emotional learning or better measurements of their sense of belonging (OECD, 2020c).

In the areas of mental health and psychosocial support and education, Save the Children makes the following recommendations:

Work with schools to ensure that age-appropriate, gender-sensitive, inclusive, accessible messages on psychosocial wellbeing and stress prevention messages are conveyed to children and their caregivers in a way that reassures rather than distresses.

Support schools to identify and refer children in high distress and/or showing signs of a mental health condition (Gordon and Burgess, 2020, p. 11).

The issue of children's and learners' well-being, reinforced by the school closure, was prominent in national debates and consultations. In Ireland, for example, it 'is likely to provide a rationale for reform in upper secondary education' (European Commission, 2020a). The Irish ministry's National Educational Psychological Service psychologists developed resources to help young people manage their well-being during the school closure: Advice for young people while schools are closed; Plan for the Day; Relaxation Techniques (Croatian Presidency of the Council of the European Union, 2020).

Mental health remains an issue at a global level for both children and adults. Governments are advised to reinforce mental health services and to protect the human rights and dignity of those already affected by mental health conditions. The UN defined three areas for action in country responses to mental health issues:

- 1) consciously including this issue in response plans taking a whole-of-society approach;
- 2) ensuring widespread availability of emergency mental health and psychosocial support;
- 3) building mental health services for the future, overcoming the long-standing underinvestment in this area (United Nations, 2020a, p. 25).

Audrey Azoulay, Director-General of UNESCO, suggests another solution to ensure learners' well-being, which is the re-opening of schools:

Prolonged and repeated closures of education institutions are taking a rising psycho-social toll on students, increasing learning losses and the risk of dropping out, disproportionately impacting the most vulnerable. Full school closures must therefore be a last resort and reopening them safely a priority (UNESCO, 2021b).



6.3.3. Key points

- Through school closures, the COVID-19 crisis affected children's well-being worldwide and increased the risk of abuse and violence.
- Children's well-being and measures to preserve their mental and emotional health should be a priority for all European policy-makers.



7. A SYNTHESIS OF KEY MESSAGES AND RECOMMENDATIONS

The section provides a summary overview of key lessons learned during the crisis (with an **evidence** dimension) and key **recommendations** available and formulated in the analysed documents. It enables a better understanding of the links between the main findings of this literature review.

At the level of European co-ordination in education:

- Limited **evidence** is available on the initial measures and exchanges between countries. Initial guidance on inclusive education at EU level was also limited. Some sources mention aspects of inclusion, but no united perspective or platform for resources and information could be identified.
- Many recommendations are for urgent responses to vulnerable learners' needs, in line with the EU's focus on respecting all people's right to equitable and qualitative education. COVID-19 is considered both a challenge and an opportunity to review and implement more inclusive legislation, policies and actions in education across Europe. A 'build back better' message linked to education is predominant. Equality and inclusion are its core values.

At the level of country responses to the COVID-19 crisis in education:

- Available evidence shows the intervention of all categories of stakeholders for education continuity. There was a lack of co-ordination in initial actions, including authorities' inadequate responses to the realities of schools and disadvantaged learners. While multiple documents mention intermittent measures for vulnerable learners, there is limited evidence on funding or guidance for inclusion in education. School closures increase pre-existing inequalities, but are also an opportunity for new partnerships in (inclusive) education.
- National and international recommendations and interpretative analysis of the
 available documents indicate that all policy-makers could use previous expertise
 and key elements of inclusive education to address the needs of all learners. More
 structured assessment of COVID-19's impact on inclusive education across Europe
 is also recommended. New accountability measures are needed for the new policy
 measures. More prevention-based policies should replace the initial palliative
 response. The policies should be based on cross-sectoral collaborations.

Regarding the transformation of (inclusive) education systems:

There is a range of evidence of measures for remote or digital learning. As an
initial response to the COVID-19 crisis and school closures, countries developed or
adapted technological infrastructure. Initial measures to support vulnerable
learners focused on access to physical and digital tools, but no evidence on their
effectiveness could be identified. A limited number of cases focus attention on
more inclusive practices to address the 'new normal' in education.



 Available recommendations focus on missing aspects, such as more support for digital skills for learners with special educational needs. They also draw attention to learning loss and increased numbers of out-of-school learners, although no evidence of this was identified for this literature review. Limited evidence and recommendations were available on the transformation of inclusive pedagogies, curriculum, assessment and learners' progress.

Recommendations linked to the transformation of inclusive education during COVID-19 focus on the need for co-ordination and remediation measures to reduce learning loss that might have long-term consequences. More preventive measures are also recommended for accessibility of learning and quality in education.

Related to vulnerable learners' rights to education and participation:

 The review identified evidence that significant gaps exist in the effectiveness of the EU's programme for education as support for future democratic societies. There are multiple examples of limited or absent access to education for vulnerable learners. Sometimes judicial action is needed to re-establish children's right to education.

The crisis and subsequent school closures worldwide affected many aspects of children's well-being. Children's voices were seldom heard and they had limited participation in public debates. Listening to children's voices means accepting new perspectives that can inform the policy-making processes and education responses to the COVID-19 crisis.

Recommendations in the analysed literature highlight the need for more
collaboration between schools, parents and vulnerable learners. This can help to
prevent the increased risk of violence towards disadvantaged children.
 Policy-makers worldwide prioritised actions to preserve children's well-being.
 These measures are expected to be developed further.

Both evidence and recommendation documents show an on-going process of redefining traditional categories of vulnerability and well-being. New vulnerable categories tend to emerge in relation to transformed educational conditions. Mental and emotional health becomes a priority for all learners.



8. CONCLUDING COMMENTS

It is clear from the evidence that, during the COVID-19 crisis, the inclusive dimensions of education received less public attention than its more general or traditional dimensions. This is in line with pre-pandemic trends.

However, common themes appear across documents about all forms of education, inclusive or not: the need for European co-ordination, more holistic approaches, measures to alleviate or prevent learning loss, and learner well-being.

The literature review also highlighted that expertise and key elements associated with inclusive education could be a catalyst against COVID-19's negative impact on education systems.

The review's structure reflects the topics of interest identified in the collected documents. Not all topics received equal attention. Learning loss, cross-sectoral partnerships, remote digital education, and children's well-being can be considered priority topics in the collected literature.

The current review's findings could be a starting point for several areas of attention in future Agency work and discussions:

- Future work could further explore the priority topics already identified in the collected literature. Additional monitoring might also highlight new interests and topics related to inclusive education during the COVID-19 pandemic.
- The COVID-19 crisis has already started to change educational processes. Against this background, the transformation of the six principles of inclusive education systems, as identified in the Agency's position paper (European Agency, 2015a), would need to be further explored and promoted:
 - The raised achievement of learners under crisis conditions
 - Stakeholders' engagement for diversity
 - The availability of flexible continua of provision and resources that support the learning of all stakeholders at both individual and organisational levels
 - The use of more personalised approaches in education for all learners
 - The raised achievements, outcomes and outputs of inclusive education systems under crisis conditions
 - The development of inclusive systems and their stakeholders' participation under crisis conditions.
- Future work could focus on learners' voices, especially the voices of vulnerable learners and/or learners with special educational needs, in the debate on the most adapted responses to the challenges of the COVID-19 pandemic.
- The well-being of and support for teachers and other school staff do not emerge as priority areas in the collected literature. However, they are critical to enable teaching and learning to continue. Also, there is little evidence of teachers' voices



- being heard in relation to COVID-19's impact on education. Future work could explore these dimensions.
- The collected literature talks about 'learning', 'learners' and 'teachers'. However,
 the teaching dimension is under-represented. Future Agency work could explore
 and/or promote inclusive teaching and pedagogical approaches during and after
 the pandemic. This could also include the potential of the universal approach to
 teaching and learning, and the new role that specialist staff might play for systemic
 change.
- This review mainly focused on compulsory education. Future work could explore COVID-19's impact on early childhood education and care, post-school education, and the difficulties for young people in transition phases of their education (such as entering school, entering secondary school, entering higher education or vocational training).
- The review showed the key role of non-state actors in education, with many initiatives that might contribute to future inclusive education issues. In line with this finding, future Agency work could explore the dynamics and transformation of cross-sectoral collaboration during and after the pandemic.
- Systematic collection of evidence on inclusive education practices and measures during COVID-19 is also needed, alongside a unique platform with resources to inform future decisions at country and European levels.



9. REFERENCES

ActionAid, 2020. *Call to action on domestic financing of education*. actionaid.org/sites/default/files/publications/Call%20to%20action%20on%20domestic%2 Ofinancing%20of%20education%20WITH%20SIGN%20ONs.pdf (Last accessed April 2021)

Active Minds, 2020. COVID-19 Impacted College Students' Mental Health Hardest, According to Nationwide Survey of Students. www.activeminds.org/press-releases/active-minds-and-association-of-college-and-university-educators-release-guide-on-practical-approaches-for-supporting-student-wellbeing-and-mental-health-copy (Last accessed April 2021)

Ainscow, M., 2005. 'Developing inclusive education systems: What are the levers for change?' *Journal of Educational Change*, 6 (2), 109–124. doi.org/10.1007/s10833-005-1298-4 (Last accessed April 2021)

Ainscow, M. and Haile-Giorgis, M., 1999. 'Educational arrangements for children categorized as having special needs in Central and Eastern Europe' *European Journal of Special Needs Education*, 14 (2), 103–121. doi.org/10.1080/0885625990140201 (Last accessed April 2021)

Al-Samarrai, S., 2020. *The impact of the COVID-19 pandemic on education financing*. Washington, DC: World Bank Group.

<u>documents1.worldbank.org/curated/en/479041589318526060/pdf/The-Impact-of-the-COVID-19-Pandemic-on-Education-Financing.pdf</u> (Last accessed April 2021)

Alliance for Inclusive Education, 2020. *The impact of COVID-19 on education and children's services*. www.allfie.org.uk/news/briefings/submission-to-the-education-committee-impact-of-covid-19-on-education-and-childrens-services-inquiry (Last accessed April 2021)

Azevedo, J.P., Hasan, A., Goldemberg, D., Iqbal, S.A. and Geven, K., 2020. *Simulating the Potential Impacts of COVID-19 School Closures on Schooling and Learning Outcomes: A Set of Global Estimates*. Washington, DC: World Bank. doi:org/10.1596/1813-9450-9284 (Last accessed April 2021)

Bassot, É., 2020. *Ten opportunities for Europe post-coronavirus. Exploring potential for progress in EU policy-making*. Brussels: European Parliamentary Research Service. https://www.europarl.europa.eu/RegData/etudes/IDAN/2020/652035/EPRS IDA(2020)652035 EN.pdf (Last accessed April 2021)

Bundesministerium für Familie, Senioren, Frauen und Jugend, 2020. *Joint Declaration by the Ministers of the EPSCO Council: Overcoming poverty and social exclusion – mitigating the impact of COVID-19 on families – working together to develop prospects for strong children*. Berlin: Bundesministerium für Familie, Senioren, Frauen und Jugend Referat Öffentlichkeitsarbeit.

<u>www.bmfsfj.de/resource/blob/163116/92825af8e669b65f85de0521bbac9ddb/20201211-en-erklaerung-eu-mitgliedstaaten-poverty-armut-data.pdf</u> (Last accessed March 2021)



Burke, J. and Dempsey, M., 2020. *Covid-19 Practice in Primary Schools in Ireland Report*. Maynooth, Ireland: Maynooth University.

www.maynoothuniversity.ie/sites/default/files/assets/document/Covid-19 Practice in Primary Schools Report 0.pdf (Last accessed April 2021)

Centre for Children and Young People's Participation, 2020. Building on Rainbows.

Supporting Children's Participation in Shaping Responses to COVID-19.

clok.uclan.ac.uk/33087/1/28%20April%20Building%20on%20Rainbows-

%20Child%20Participation%20in%20COVID-19%20Respone%20%20-

%20Rapid%20Paper .pdf (Last accessed April 2021)

Cerna, L., 2020. *Coronavirus school closures: What do they mean for student equity and inclusion?* OECD Education and Skills Today.

<u>oecdedutoday.com/coronavirus-school-closures-student-equity-inclusion</u> (Last accessed April 2021)

Chircop, D., 2020. *Briefing. Education in isolation in the pandemic, following the path of Isaac Newton*. Brussels: European Parliamentary Research Service.

www.europarl.europa.eu/thinktank/en/document.html?reference=EPRS BRI(2020)65194 0 (Last accessed April 2021)

Committee for the Coordination of Statistical Activities, 2020. *How COVID-19 is changing the world: A statistical perspective*. Committee for the Coordination of Statistical Activities. openknowledge.worldbank.org/handle/10986/33773 (Last accessed April 2021)

Committee on the Rights of the Child, 2020. The Committee on the Rights of the Child warns of the grave physical, emotional and psychological effect of the COVID-19 pandemic on children and calls on States to protect the rights of children.

<u>tbinternet.ohchr.org/Treaties/CRC/Shared%20Documents/1 Global/INT CRC STA 9095</u> <u>E.docx</u> (Last accessed April 2021)

Cooper, H., Nye, B., Charlton, K., Lindsay, J. and Greathouse, S., 1996. 'The Effects of Summer Vacation on Achievement Test Scores: A Narrative and Meta-Analytic Review' *Review of Educational Research*, 66 (3), 227–268. doi.org/10.3102/00346543066003227 (Last accessed April 2021)

Council of Europe, 2012. *Recommendation CM/Rec(2012)13 of the Committee of Ministers to member States on ensuring quality education*.

<u>search.coe.int/cm/Pages/result_details.aspx?ObjectId=09000016805c94fb</u> (Last accessed April 2021)

Council of Europe, 2020a. Making the right to education a priority in times of COVID-19. A Roadmap for Action on the Council of Europe education response to COVID-19. Strasbourg: Council of Europe. rm.coe.int/making-the-right-to-education-a-reality-in-times-of-covid-19-a-roadmap/16809fee7b (Last accessed April 2021)

Council of Europe, 2020b. *Interview with Ligia Deca, Presidential Advisor to the President of Romania, Department of Education and Research*.

www.coe.int/en/web/education/interview-with-ligia-deca (Last accessed April 2021)



Council of the European Union, 2020. *Council conclusions on countering the COVID-19 crisis in education and training*. Brussels: Council of the European Union.

eur-lex.europa.eu/legal-

<u>content/EN/TXT/?uri=CELEX%3A52020XG0626%2801%29&qid=1620300743553</u> (Last accessed April 2021)

Croatian Presidency of the Council of the European Union, 2020. *Implications of the novel coronavirus (COVID-19) on education and training: State-of-play in Member States*. Version 9 as of 31/3/2020. Unpublished

Directorate-General for Internal Policies of the Union (European Parliament) and European Agency for Special Needs and Inclusive Education, 2017. *Inclusive education for learners with disabilities*. PE 596.807. Brussels: European Parliament.

www.europarl.europa.eu/thinktank/en/document.html?reference=IPOL_STU(2017)59680

Capture Common Commo

Dumbrava, C., 2020. *Briefing. Tracking key coronavirus restrictions on movement and social life*. Brussels: European Parliamentary Research Service.

www.europarl.europa.eu/thinktank/en/document.html?reference=EPRS_BRI(2020)65201

8 (Last accessed April 2021)

Education Endowment Foundation, 2020. Rapid evidence assessment on online distance learning for school aged pupils. Protocol for a rapid evidence assessment.

educationendowmentfoundation.org.uk/public/files/Campaigns/Distance Learning Rapid

Evidence Assessment Protocol.pdf (Last accessed April 2021)

European Agency for Development in Special Needs Education, 2009. *Development of a set of indicators – for inclusive education in Europe*. (M. Kyriazopoulou and H. Weber, eds.). Odense, Denmark.

<u>www.european-agency.org/resources/publications/development-set-indicators-inclusive-education-europe</u> (Last accessed April 2021)

European Agency for Development in Special Needs Education, 2011. *Participation in Inclusive Education – A Framework for Developing Indicators*. Odense, Denmark. www.european-agency.org/resources/publications/participation-inclusive-education-framework-developing-indicators (Last accessed April 2021)

European Agency for Development in Special Needs Education, 2013. *Organisation of Provision to Support Inclusive Education – Literature Review*. (S. D'Alessio and V. Donnelly, eds.). Odense, Denmark. www.europeanagency.org/publications/reviews/organisation-of-provision-to-support-inclusiveeducation-2013-literature-review (Last accessed April 2021)

European Agency for Special Needs and Inclusive Education, 2015a. *Agency Position on Inclusive Education Systems*. Odense, Denmark. www.european-agency.org/resources/publications/agency-position-inclusive-education-systems-flyer (Last accessed April 2021)

European Agency for Special Needs and Inclusive Education, 2015b. *Inclusive Pre-Primary Education Project: Conceptual Framework*. www.european-agency.org/sites/default/files/agency-projects/IECE/IPPE-Conceptual-Framework.pdf (Last accessed April 2021)



European Agency for Special Needs and Inclusive Education, 2017a. *Inclusive Early Childhood Education: New Insights and Tools – Contributions from a European Study*. (M. Kyriazopoulou, P. Bartolo, E. Björck-Åkesson, C. Giné and F. Bellour, eds.). Odense, Denmark. www.european-agency.org/resources/publications/inclusive-early-childhood-education-new-insights-and-tools-contributions (Last accessed April 2021)

European Agency for Special Needs and Inclusive Education, 2017b. *Raising the Achievement of All Learners: A Resource to Support Self-Review*. (V.J. Donnelly and A. Kefallinou, eds.). Odense, Denmark. www.european-agency.org/resources/publications/raising-achievement-all-learners-self-review (Last accessed April 2021)

European Agency for Special Needs and Inclusive Education, 2017c. Raising the Achievement of All Learners in Inclusive Education: Lessons from European Policy and Practice. (A. Kefallinou and V.J. Donnelly, eds.). Odense, Denmark. www.european-agency.org/resources/publications/raising-achievement-all-learners-project-overview (Last accessed April 2021)

European Agency for Special Needs and Inclusive Education, no date. 'Co-operation / cross-sectoral working' *Glossary*. www.european-agency.org/resources/glossary (Last accessed April 2021)

European Commission, 2018a. *Commission Staff working document accompanying the document Proposal for a Council Recommendation on High Quality Early Childhood Education and Care Systems*. {COM(2018) 271 final}. Brussels: European Commission. eur-lex.europa.eu/legal-content/EN/TXT/HTML/?uri=CELEX:52018SC0173&from=EN (Last accessed April 2021)

European Commission, 2018b. *National Policies Platform: YouthWiki*. national-policies.eacea.ec.europa.eu (Last accessed April 2021)

European Commission, 2020a. *Education and Training Monitor 2020 – Country Analysis*. Luxembourg: Publications Office of the European Union.

op.europa.eu/en/publication-detail/-/publication/cd97ef45-2497-11eb-9d7e-01aa75ed71a1/language-en/format-PDF/source-172905496 (Last accessed April 2021)

European Commission, 2020b. *Digital Education Action Plan 2021–2027: Resetting education and training for the digital age*.

<u>ec.europa.eu/education/sites/default/files/document-library-docs/deap-communication-sept2020_en.pdf</u> (Last accessed March 2021)

European Commission, 2020c. *Education and Training Monitor 2020. Teaching and learning in a digital age*. Luxembourg: Publications Office of the European Union. op.europa.eu/en/publication-detail/-/publication/92c621ce-2494-11eb-9d7e-01aa75ed71a1 (Last accessed April 2021)

European Commission, 2020d. Staff working document. Accompanying the document Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions – Digital Education action Plan 2021-2027: Resetting education and training for the digital age. eur-lex.europa.eu/legal-content/LT/TXT/?uri=CELEX:52020SC0209 (Last accessed April 2021)



European Commission, 2020e. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee of the regions – Digital Education Action Plan 2021-2027: Resetting education and training for the digital age {SWD(2020) 209 final}.

<u>eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52020DC0624</u> (Last accessed April 2021)

European Disability Forum, 2020. *Open letter to leaders at the EU and in EU countries: COVID-19 – Disability inclusive response*. <u>www.edf-feph.org/newsroom-news-open-letter-leaders-eu-and-eu-countries-covid-19-disability-inclusive-response</u> (Last accessed April 2021)

European Distance and E-Learning Network, 2020. 'Digital Education Action Plan 2021–2027 Setting education and training for the digital age' *EDEN webinar series: Education in time of a new normal*.

<u>www.eden-online.org/eden conference/no-6-digital-education-action-plan</u> (Last accessed April 2021)

European Parliament, 2020. The future of European education in the context of COVID-19 – European Parliament resolution of 22 October 2020 on the future of European education in the context of COVID-19 (2020/2760(RSP)).

<u>www.europarl.europa.eu/doceo/document/TA-9-2020-0282_EN.pdf</u> (Last accessed April 2021)

European Union Agency for Fundamental Rights, 2020. *Fundamental rights report – 2020*. Luxembourg: Publications Office of the European Union.

<u>fra.europa.eu/sites/default/files/fra_uploads/fra-2020-fundamental-rights-report-2020_en.pdf</u> (Last accessed April 2021)

Faulkner, S.L. and Trotter, S.P., 2017. 'Data Saturation' in J. Matthes, C.S. Davis and R.F. Potter (eds.), *The International Encyclopedia of Communication Research Methods*. doi.org/10.1002/9781118901731.iecrm0060 (Last accessed April 2021)

Florian, L., 2014. 'What counts as evidence of inclusive education?' *European Journal of Special Needs Education*, 29 (3), 286–294. doi.org/10.1080/08856257.2014.933551 (Last accessed April 2021)

Florian, L. and Black-Hawkins, K., 2011. 'Exploring inclusive pedagogy' *British Educational Research Journal*, 37, 813–828. doi.org/10.1080/01411926.2010.501096 (Last accessed April 2021)

Giannini, S. and Grant Lewis, S., 2020. 'Three ways to plan for equity during the coronavirus school closures' *World Education Blog*.

gemreportunesco.wordpress.com/2020/03/25/three-ways-to-plan-for-equity-during-the-coronavirus-school-closures (Last accessed April 2021)

Glaser, B.G. and Strauss, A.L., 1967. *The discovery of grounded theory: Strategies for qualitative research*. Piscataway, NJ: Aldine Publishing

GOETE, no date. *Disadvantaged Youth – GOETE Glossary*. <u>www.goete.eu/glossary33/index.php/Disadvantaged Youth</u> (Last accessed April 2021)



Gordon, M. and Burgess, M., 2020. *The Hidden Impact of COVID-19 on Children's Education and Learning*. London: Save the Children International. resourcecentre.savethechildren.net/node/18174/pdf/the hidden impact of covid-19 on child education.pdf (Last accessed April 2021)

Greek Chairmanship of the Committee of Ministers, 2020. *Informal Conference of Ministers of Education organised under the Greek Chairmanship of the Committee of Ministers. The education response to the COVID crisis Political declaration*. rm.coe.int/the-education-response-to-the-covid-crisis-political-declaration-for-t/16809fee7a (Last accessed April 2021)

Greenhalgh, T. and Peacock, R., 2005. 'Effectiveness and efficiency of search methods in systematic reviews of complex evidence: Audit of primary sources' *BMJ*, 331 (7524), 1064–1065. doi.org/10.1136/bmj.38636.593461.68 (Last accessed April 2021)

Hooge, E., Burns, T. and Wilkoszewski, H., 2012. 'Looking Beyond the Numbers: Stakeholders and Multiple School Accountability' *OECD Education Working Papers*, No. 85. Paris: OECD Publishing. doi.org/10.1787/5k91dl7ct6q6-en (Last accessed April 2021)

Huber, S.G. and Helm, C., 2020. 'COVID-19 and schooling: Evaluation, assessment and accountability in times of crises – reacting quickly to explore key issues for policy, practice and research with the school barometer' *Educational Assessment, Evaluation and Accountability*, 32 (2), 237–270. doi.org/10.1007/s11092-020-09322-y (Last accessed April 2021)

Hyseni Duraku, Z. and Nagavci, M., 2020. *The impact of the COVID-19 pandemic on the education of children with disabilities*. doi.org/10.13140/RG.2.2.17807.41125 (Last accessed April 2021)

Inclusion Europe, 2020. *Neglect and discrimination. Multiplied. How Covid-19 affected the rights of people with intellectual disabilities and their families.* Brussels: Inclusion Europe. www.inclusion-europe.eu/wp-content/uploads/2020/11/COVID-report-Final.pdf (Last accessed April 2021)

INEE, 2020a. An inclusive response to COVID-19: Ensuring education for all. inee.org/resources/inclusive-response-covid-19-ensuring-education-all (Last accessed April 2021)

INEE, 2020b. *Coronavirus (COVID-19) Resources*. <u>inee.org/covid-19/resources</u> (Last accessed April 2021)

Joynes, C. and Gibbs, E., 2020. 'Report for EdTechHub (ODI). An overview of emerging country-level responses to providing educational continuity under COVID-19: What's working? What isn't?' *Education Development Trust*, 30.

<u>edtechhub.org/wp-content/uploads/2020/05/whats-working-whats-not.pdf</u> (Last accessed April 2021)

Kaden, U., 2020. 'COVID-19 School Closure-Related Changes to the Professional Life of a K-12 Teacher' *Education Sciences*, 10 (6), 165. doi:org/10.3390/educsci10060165 (Last accessed April 2021)

Kittay, E.F., 2019. *Learning from my daughter: the value and care of disabled minds*. New York: Oxford University Press



Lancet Public Health, 2020. 'Will the COVID-19 pandemic threaten the SDGs?' *The Lancet Public Health*, 5 (9), e460. doi.org/10.1016/S2468-2667(20)30189-4 (Last accessed April 2021)

Lavy, V., 2015. 'Do Differences in Schools' Instruction Time Explain International Achievement Gaps? Evidence from Developed and Developing Countries' *The Economic Journal*, 125 (588), F397–F424. doi.org/10.1111/ecoj.12233 (Last accessed April 2021)

Loreman, T., Forlin, C. and Sharma, U., 2014. 'Measuring Indicators of Inclusive Education: A Systematic Review of the Literature', in C. Forlin and T. Loreman (eds.), *Measuring Inclusive Education (International Perspectives on Inclusive Education, Vol. 3)*. Bingley: Emerald Group Publishing Limited. doi:10.1108/S1479-363620140000003024 (Last accessed April 2021)

Mashford-Scott, A., Church, A. and Tayler, C., 2012. 'Seeking Children's Perspectives on their Wellbeing in Early Childhood Settings' *IJEC*, 44, 231–247. doi.org/10.1007/s13158-012-0069-7 (Last accessed April 2021)

McAleavy, T., Joynes, C., Gibbs, E. and Sims, K., 2020. What steps are being taken to reach the most disadvantaged students during the period of Covid-19 school closure? Reading: Education Development Trust.

<u>edtechhub.org/wp-content/uploads/2020/05/disadvantaged-students.pdf</u> (Last accessed April 2021)

Müller, L-M. and Goldenberg, G., 2020. *Education in times of crisis: The potential implications of school closures for teachers and students*. London: Chartered College of Teaching.

<u>my.chartered.college/wp-content/uploads/2020/05/CCTReport070520_FINAL.pdf</u> (Last accessed April 2021)

OECD, 2017. *How's Life? 2017: Measuring Well-being*. Paris: OECD Publishing. dx.doi.org/10.1787/how life-2017-en (Last accessed April 2021)

OECD, 2020a. *Education at a Glance 2020: OECD Indicators*. Paris: OECD Publishing. doi.org/10.1787/69096873-en (Last accessed April 2021)

OECD, 2020b. How can education systems and schools support vulnerable students during school closures and school re-openings? A short summary. Strength through Diversity webinar. Paris, 5 October 2020

OECD, 2020c. The impact of COVID-19 on student equity and inclusion: Supporting vulnerable students during school closures and school re-openings. Paris: OECD Publishing. www.oecd.org/education/strength-through-diversity/OECD COVID-19 Brief Vulnerable-Students.pdf (Last accessed April 2021)

OECD, 2020d. *OECD Policy Responses to Coronavirus (COVID-19): Combatting COVID-19's effect on children*. Paris: OECD Publishing. <u>doi.org/10.1787/2e1f3b2f-en</u> (Last accessed April 2021)

OECD, 2020e. *How's Life? 2020: Measuring Well-being*. Paris: OECD Publishing. doi.org/10.1787/9870c393-en (Last accessed April 2021)



Office of the United Nations High Commissioner for Human Rights, 2020. *COVID-19 and The Rights of Persons with Disabilities: Guidance*.

www.ohchr.org/Documents/Issues/Disability/COVID-

19 and The Rights of Persons with Disabilities.pdf (Last accessed April 2021)

Oxford Dictionaries, no date. *Accountability*. www.lexico.com/definition/accountability (Last accessed April 2021)

Ramberg, J. and Watkins, A., 2020. 'Exploring inclusive education across Europe: Some insights from the European Agency Statistics on Inclusive Education' *FIRE: Forum for International Research in Education*, 6 (1). doi:org/10.32865/fire202061172 (Last accessed April 2021)

Reimers, F.M. and Schleicher, A., 2020. *OECD Policy Responses to Coronavirus (COVID-19):* A framework to guide an education response to the COVID-19 Pandemic of 2020. Paris: OECD Publishing. doi:10.1787/6ae21003-en (Last accessed April 2021)

Right to Education Initiative, 2020. 190 organisations join call for increased funding for education post-Covid. www.right-to-education.org/news/190-organisations-join-call-increased-funding-education-post-covid (Last accessed April 2021)

Sahlberg, P., 2021. 'Does the pandemic help us make education more equitable?' Educational Research for Policy and Practice, 20, 11–18. doi.org/10.1007/s10671-020-09284-4 (Last accessed April 2021)

School Education Gateway, 2020. *European national websites for online and distance learning during school closures*.

<u>www.schooleducationgateway.eu/de/pub/latest/news/european-national-websites.htm</u> (Last accessed April 2021)

SDG-Education 2030 Steering Committees, no date. *Sustainable Development Goal 4* (SDG 4). The 2030 Agenda for Sustainable Development. sdg4education2030.org/the-goal (Last accessed April 2021)

UN General Assembly, 1989. *Convention on the Rights of the Child*. www.ohchr.org/EN/ProfessionalInterest/Pages/CRC.aspx (Last accessed April 2021)

UN Women, 2020. How COVID-19 impacts women and girls.

<u>interactive.unwomen.org/multimedia/explainer/covid19/en/index.html</u> (Last accessed April 2021)

UNESCO, 2017. *A guide for ensuring inclusion and equity in education*. Paris: UNESCO. <u>unesco.org/images/0024/002482/248254e.pdf</u> (Last accessed April 2021)

UNESCO, 2018. Global Education Monitoring Report 2019: Migration, Displacement and Education – Building Bridges, not Walls. Paris: UNESCO.

unesdoc.unesco.org/ark:/48223/pf0000265866 (Last accessed April 2021)

UNESCO, 2020a. *Education: From disruption to recovery*. en.unesco.org/covid19/educationresponse (Last accessed April 2021)

UNESCO, 2020b. *Policy Paper 42: Act now: Reduce the impact of COVID-19 on the cost of achieving SDG 4*. Paris: UNESCO. <u>en.unesco.org/gem-report/COVIDcostSDG4</u> (Last accessed April 2021)



UNESCO, 2020c. *Distance learning strategies in response to COVID-19 school closures*. Paris: UNESCO. unesdoc.unesco.org/ark:/48223/pf0000373305 (Last accessed April 2021)

UNESCO, 2020d. 290 million students out of school due to COVID-19: UNESCO releases first global numbers and mobilizes response. en.unesco.org/news/290-million-students-out-school-due-covid-19-unesco-releases-first-global-numbers-and-mobilizes (Last accessed April 2021)

UNESCO, 2020e. *COVID-19: 10 Recommendations to plan distance learning solutions*. <u>en.unesco.org/news/covid-19-10-recommendations-plan-distance-learning-solutions</u> (Last accessed April 2021)

UNESCO, 2020f. *Global Education Monitoring Report 2020: Inclusion and education: All means all*. Paris: UNESCO. <u>en.unesco.org/gem-report/report/2020/inclusion</u> (Last accessed April 2021)

UNESCO, 2021a. Global Education Monitoring Report 2021 – Central and Eastern Europe, the Caucasus and Central Asia – Inclusion and education: All means all. Paris: UNESCO. en.unesco.org/gem-report/Eurasia2021inclusion (Last accessed April 2021)

UNESCO, 2021b. UNESCO figures show two thirds of an academic year lost on average worldwide due to Covid-19 school closures. en.unesco.org/news/unesco-figures-show-two-thirds-academic-year-lost-average-worldwide-due-covid-19-school (Last accessed April 2021)

UNESCO IBE, 2008. *Inclusive Education: The Way of the Future. Conclusions and Recommendations of the 48th Session of the International Conference on Education (ICE).*https://www.ibe.unesco.org/fileadmin/user-upload/Policy-Dialogue/48th ICE/CONFINTED 48-5-Conclusions-english.pdf (Last accessed April 2021)

UNESCO IITE/European Agency, 2011. *ICTs in Education for People with Disabilities:* Review of innovative practice. Moscow: UNESCO Institute for Information Technologies in Education. www.european-agency.org/resources/publications/icts-education-people-disabilities-review-innovative-practice (Last accessed April 2021)

UNESCO Institute for Educational Planning, 2005. *Accountability in education*. Paris: UNESCO-IIEP. <u>unesdoc.unesco.org/ark:/48223/pf0000140986</u> (Last accessed April 2021)

UNESCO, UNICEF, the World Bank, World Food Programme and UNHCR, 2020a. *Framework for reopening schools*. www.unicef.org/sites/default/files/2020-06/Framework-for-reopening-schools-2020.pdf (Last accessed April 2021)

UNESCO, UNICEF, the World Bank, World Food Programme and UNHCR, 2020b. Supplement to Framework for reopening schools: Emerging lessons from country experiences in managing the process of reopening schools.

<u>reliefweb.int/sites/reliefweb.int/files/resources/5f6b3dec4.pdf</u> (Last accessed April 2021)

UNICEF, 2013. *Take us Seriously! Engaging Children with Disabilities in Decisions Affecting their Lives*. New York: UNICEF. <u>sites.unicef.org/disabilities/files/Take Us Seriously.pdf</u> (Last accessed April 2021)



UNICEF, 2020a. *COVID-19 response: Considerations for Children and Adults with Disabilities*. reliefweb.int/sites/reliefweb.int/files/resources/covid-19 response considerations for people with disabilities 190320.pdf (Last accessed April 2021)

UNICEF, 2020b. *Protecting children and adolescents with disabilities from the pandemic*. www.unicef.org/eca/protecting-children-and-adolescents-disabilities-pandemic (Last accessed April 2021)

UNICEF, 2020c. Safe to Learn during COVID-19: Recommendations to prevent and respond to violence against children in all learning environments.

<u>www.unicef.org/media/68781/file/STL-COVID-19-Response-Key-Messages.pdf</u> (Last accessed April 2021)

United Nations, 2020a. *United Nations Comprehensive Response to COVID-19: Saving Lives, Protecting Societies, Recovering Better.*

<u>unsdg.un.org/sites/default/files/2020-06/UN-Response-to-COVID-19.pdf</u> (Last accessed April 2021)

United Nations, 2020b. Opening remarks at virtual press encounter to launch the Report on the Socio-Economic Impacts of COVID-19.

www.un.org/sg/en/content/sg/speeches/2020-03-31/remarks-launch-of-report-the-socioeconomic-impacts-of-covid-19 (Last accessed April 2021)

United Nations, 2020c. *Global humanitarian response plan COVID-19*. Geneva: United Nations Office for the Coordination of Humanitarian Affairs.

www.unocha.org/sites/unocha/files/Global-Humanitarian-Response-Plan-COVID-19.pdf (Last accessed April 2021)

Venezky, R.L., 2000. 'The Digital Divide Within Formal School Education: Causes and Consequences' in OECD (ed.), *Learning to Bridge the Digital Divide*, 63–76. docplayer.net/22723450-Chapter-5-the-digital-divide-within-formal-school-education-causes-and-consequences.html (Last accessed April 2021)

Vuorikari, R., Velicu, A., Chaudron, S., Cachia, R. and Di Gioia, R., 2020. *How families handled emergency remote schooling during the Covid-19 lockdown in spring 2020*. EUR 30425 EN. Luxembourg: Publications Office of the European Union. ec.europa.eu/jrc/en/publication/eur-scientific-and-technical-research-reports/how-families-handled-emergency-remote-schooling-during-covid-19-lockdown-spring-2020 (Last accessed April 2021)

Watkins, A., Donnelly, V. and Meijer, C., forthcoming. 'Paying for inclusion and exclusion' *International Encyclopaedia of Education*

Watkins, A. and Ebersold, S., 2016. 'Efficiency, Effectiveness and Equity within Inclusive Education Systems' in A. Watkins and C. Meijer (eds.), *Implementing Inclusive Education: Issues in Bridging the Policy-Practice Gap (International Perspectives on Inclusive Education, Vol. 8)*. Bingley: Emerald Group Publishing Limited. doi.org/10.1108/S1479-363620160000008014 (Last accessed April 2021)



WHO, 2020a. Statement on the second meeting of the International Health Regulations (2005) – Emergency Committee regarding the outbreak of novel coronavirus (2019-nCoV). www.who.int/news/item/30-01-2020-statement-on-the-second-meeting-of-the-international-health-regulations-(2005)-emergency-committee-regarding-the-outbreak-of-novel-coronavirus-(2019-ncov) (Last accessed April 2021)

WHO, 2020b. WHO Director-General's opening remarks at the media briefing on COVID-19 – 11 March 2020. <a href="https://www.who.int/director-general/speeches/detail/who-director-general-speeches/de

WHO-UNICEF-Lancet Commissioners, 2020. 'After COVID-19, a future for the world's children?' *The Lancet*, 396 (10247), 298–300. doi.org/10.1016/S0140-6736(20)31481-1 (Last accessed April 2021)

World Bank, 2020. How countries are using edtech (including online learning, radio, television, texting) to support access to remote learning during the COVID-19 pandemic. www.worldbank.org/en/topic/edutech/brief/how-countries-are-using-edtech-to-support-remote-learning-during-the-covid-19-pandemic (Last accessed April 2021)



10. BIBLIOGRAPHY – COMPLETE LIST OF RESOURCES

This section includes the complete list of the collected documents about COVID-19's impact on (inclusive) education at the European level. The information covers the period March–December 2020.

Documents that are cited directly in sections 1 to 8 and/or in the <u>Annex</u> are also listed in the <u>References</u>.

Documents with information on surveys, which are extracted from this general bibliography, are also listed separately in section 10.2.

10.1 Bibliography

#CovidUnder19, 2020. Children's Rights during Coronavirus: Children's Views and Experiences. Belfast: Queen's University, Centre for Children's Rights.

childhub.org/sites/default/files/library/attachments/covidunder19 thematic summaries
english 20201209.pdf.pdf (Last accessed April 2021)

Accelerated Education Working Group (AEWG), 2020. *COVID-19: Pathways for the Return to Learning*. <u>inee.org/system/files/resources/AEWG COVID19 Pathways for the Return to Learning-screen_1.pdf</u> (Last accessed April 2021)

Accelerated Education Working Group (AEWG), 2020. *COVID-19: Pathways for the Return to Learning: Guidance on Condensing a Curriculum*.

<u>inee.org/system/files/resources/AEWG Guidance on Condensing a Curriculum-Considerations for Practitioners-screen 3.pdf</u> (Last accessed April 2021)

ActionAid, 2020. Call to action on domestic financing of education.

<u>actionaid.org/sites/default/files/publications/Call%20to%20action%20on%20domestic%2</u> <u>0financing%20of%20education%20WITH%20SIGN%20Ons.pdf</u> (Last accessed April 2021)

Active Minds, 2020. COVID-19 Impacted College Students' Mental Health Hardest, According to Nationwide Survey of Students. www.activeminds.org/press-releases/activeminds-and-association-of-college-and-university-educators-release-guide-on-practical-approaches-for-supporting-student-wellbeing-and-mental-health-copy (Last accessed April 2021)

Alasuutari, H., 2020. 'Tackling inequity in education during and after COVID-19' *World Bank Blogs*.

blogs.worldbank.org/education/tackling-inequity-education-during-and-after-covid-19 (Last accessed April 2021)

Al-Samarrai, S., 2020. *The impact of the COVID-19 pandemic on education financing*. Washington, DC: World Bank Group.

<u>documents1.worldbank.org/curated/en/479041589318526060/pdf/The-Impact-of-the-COVID-19-Pandemic-on-Education-Financing.pdf</u> (Last accessed April 2021)

Alliance for Inclusive Education, 2020. *The impact of COVID-19 on education and children's services*. www.allfie.org.uk/news/briefings/submission-to-the-education-committee-impact-of-covid-19-on-education-and-childrens-services-inquiry (Last accessed April 2021)



Assunção Flores, M. and Gago, M., 2020. 'Teacher education in times of COVID-19 pandemic in Portugal: National, institutional and pedagogical responses' *Journal of Education for Teaching*, 46 (4), 507–516. <u>doi.org/10.1080/02607476.2020.1799709</u> (Last accessed April 2021)

Ayllón, S., Barbovschi, M., Casamassima, G., Drossel, K., Eickelmann, B., Gheţău, C., Haragus, T.P., Holmarsdottir, H., Hyggen, C., Kapella, O., Karatzogianni, A., Lado, S., Levine, D.T., Lorenz, T., Mifsud, L., Parsanoglou, D., Port, S., Sisask, M., Symeonaki, M. and Teidla-Kunitsõn, G., 2020. *ICT usage across Europe: A literature review and an overview of existing data*. DigiGen working paper series 2.

www.digigen.eu/wp-content/uploads/2020/09/DigiGen ICT-usage-across-Europe aliterature-review-and-an-overview-of-existing-data.pdf (Last accessed April 2021)

Ayllón, S. and Lado, S., 2020. *Digitally deprived children in Europe*. DigiGen. www.digigen.eu/index.php/2020/12/10/digitally-deprived-children-in-europe (Last accessed April 2021)

Azevedo, J.P., Hasan, A., Geven, K., Goldemberg, D. and Iqbal, S.A., 2020. 'Learning losses due to COVID19 could add up to \$10 trillion' *World Bank Blogs*.

<u>blogs.worldbank.org/education/learning-losses-due-covid19-could-add-10-trillion</u> (Last accessed April 2021)

Azevedo, J.P., Hasan, A., Goldemberg, D., Iqbal, S.A. and Geven, K., 2020. *Simulating the Potential Impacts of COVID-19 School Closures on Schooling and Learning Outcomes: A Set of Global Estimates*. Washington, DC: World Bank. <u>doi.org/10.1596/1813-9450-9284</u> (Last accessed April 2021)

Banks, J., 2020. *Podcasts with Dr Joanne Banks: Inclusion Dialogue Podcast Launch*. Dublin: School of Education, Trinity College.

www.tcd.ie/Education/research/events/podcast-joanne-banks (Last accessed April 2021)

Barnes, R., Hall, R., Lowe, V., Pottinger, C. and Popham, A., 2020. 'Lessons from an online teacher preparation Program: Flexing work experience to meet student needs and regulators' requirements in the United States' *Journal of Education for Teaching*, 46 (4), 528–535. doi.org/10.1080/02607476.2020.1802203 (Last accessed April 2021)

Bassot, É., 2020. *Ten opportunities for Europe post-coronavirus. Exploring potential for progress in EU policy-making*. Brussels: European Parliamentary Research Service. https://www.europarl.europa.eu/RegData/etudes/IDAN/2020/652035/EPRS IDA(2020)652035 EN.pdf (Last accessed April 2021)

BERA, 2020. 'Covid-19, education and educational research' *British Educational Research Association*. www.bera.ac.uk/blog-series/covid-19-education-and-educational-research (Last accessed April 2021)

Bertling, J., Rojas, N., Alegre, J. and Faherty, K., 2020. 'A tool to capture learning experiences during Covid-19: The PISA Global Crises Questionnaire Module' *OECD Education Working Papers*, No. 232. Paris: OECD Publishing. doi:oorg/10.1787/9988df4e-en (Last accessed April 2021)

Bonal, X. and González, S., 2020. 'The impact of lockdown on the learning gap: Family and school divisions in times of crisis' *International Review of Education*. doi.org/10.1007/s11159-020-09860-z (Last accessed April 2021)



Brandenburg, J.E., Holman, L.K., Apkon, S.D., Houtrow, A.J., Rinaldi, R. and Sholas, M.G., 2020. 'School reopening during COVID-19 pandemic: Considering students with disabilities' *Journal of Pediatric Rehabilitation Medicine*, 13 (3), 425–431. doi.org/10.3233/PRM-200789 (Last accessed April 2021)

Bundesministerium für Familie, Senioren, Frauen und Jugend, 2020. *Joint Declaration by the Ministers of the EPSCO Council: Overcoming poverty and social exclusion – mitigating the impact of COVID-19 on families – working together to develop prospects for strong children*. Berlin: Bundesministerium für Familie, Senioren, Frauen und Jugend Referat Öffentlichkeitsarbeit.

www.bmfsfj.de/resource/blob/163116/92825af8e669b65f85de0521bbac9ddb/20201211-en-erklaerung-eu-mitgliedstaaten-poverty-armut-data.pdf (Last accessed March 2021)

Burke, J. and Dempsey, M., 2020. *Covid-19 Practice in Primary Schools in Ireland Report*. Maynooth, Ireland: Maynooth University.

www.maynoothuniversity.ie/sites/default/files/assets/document/Covid-19 Practice in Primary Schools Report 0.pdf (Last accessed April 2021)

Byrne, B. and O'Hagan, G., 2020. *Understanding Life in Lockdown for Autistic Young People*. Belfast: Queen's University Centre for Children's Rights. www.qub.ac.uk/research-centres/CentreforChildrensRights/CCRFilestore/Filetoupload,1000988,en.pdf (Last accessed April 2021)

Cairns, R., 2020. 'Exams tested by Covid-19: An opportunity to rethink standardized senior secondary examinations' *PROSPECTS*. doi.org/10.1007/s11125-020-09515-9 (Last accessed April 2021)

Campbell, V.A., Gilyard, J.A., Sinclair, L., Sternberg, T. and Kailes, J.I., 2009. 'Preparing for and Responding to Pandemic Influenza: Implications for People With Disabilities' *American Journal of Public Health*, 99 (S2), S294–S300. doi.org/10.2105/AJPH.2009.162677 (Last accessed April 2021)

Carrillo, C. and Flores, M.A., 2020. 'COVID-19 and teacher education: A literature review of online teaching and learning practices' *European Journal of Teacher Education*, 43 (4), 466–487. doi.org/10.1080/02619768.2020.1821184 (Last accessed April 2021)

Center för Skolutveckling, 2020. *Att skapa närvaro på distans*. Gothenburg: Center för Skolutveckling. <u>www.skolahemma.se/erfarenheter-fran-distansundervisning-samlas-infor-framtida-undervisning</u> (Last accessed April 2021)

Centre for Children and Young People's Participation, 2020. *Building on Rainbows. Supporting Children's Participation in Shaping Responses to COVID-19*. clok.uclan.ac.uk/33087/1/28%20April%20Building%20on%20Rainbows-%20Child%20Participation%20in%20COVID-19%20Respone%20%20-%20Rapid%20Paper .pdf (Last accessed April 2021)

Cerna, L., 2020. *Coronavirus school closures: What do they mean for student equity and inclusion?* OECD Education and Skills Today.

<u>oecdedutoday.com/coronavirus-school-closures-student-equity-inclusion</u> (Last accessed April 2021)



Chircop, D., 2020. *Briefing. Education in isolation in the pandemic, following the path of Isaac Newton*. Brussels: European Parliamentary Research Service.

Chronic Illness Inclusion Project, 2020. *How can we build on the COVID-19 'remote access revolution'?* Sheffield: Chronic Illness Inclusion Project.

inclusionproject.org.uk/blog/covid-remote-access-revolution (Last accessed April 2021)

COFACE Families Europe, 2020. *Back to school webinar – 14th September 2020*. <u>www.coface-eu.org/education/webinar-back-to-school-towards-inclusive-education-14th-september</u> (Last accessed April 2021)

COFACE Families Europe, 2020. Recommendations on inclusive education. Based on the COFACE Disability S.H.I.F.T. guide for a meaningful inclusion of persons with disabilities and their families. www.coface-eu.org/wp-content/uploads/2020/09/Back-to-School-recommendations FINAL.pdf (Last accessed April 2021)

Colao, A., Piscitelli, P., Pulimeno, M., Colazzo, S., Miani, A. and Giannini, S., 2020. 'Rethinking the role of the school after COVID-19' *The Lancet Public Health*, 5 (7), e370. doi.org/10.1016/S2468-2667(20)30124-9 (Last accessed April 2021)

Committee for the Coordination of Statistical Activities, 2020. *How COVID-19 is changing the world: A statistical perspective*. Committee for the Coordination of Statistical Activities. openknowledge.worldbank.org/handle/10986/33773 (Last accessed April 2021)

Committee on the Rights of the Child, 2020. The Committee on the Rights of the Child warns of the grave physical, emotional and psychological effect of the COVID-19 pandemic on children and calls on States to protect the rights of children.

tbinternet.ohchr.org/Treaties/CRC/Shared%20Documents/1 Global/INT CRC STA 9095 E.docx (Last accessed April 2021)

Coolican, M., Borras, J.C. and Strong, M., 2020. 'Argentina and the COVID-19: Lessons learned from education and technical colleges in Buenos Aires Province' *Journal of Education for Teaching*, 46 (4), 484–496. doi.org/10.1080/02607476.2020.1802204 (Last accessed April 2021)

Council of Europe, 2020. *ECRI webinar on Inclusive Education in Times of Covid-19*. Strasbourg: European Commission against Racism and Intolerance (ECRI). https://www.coe.int/en/web/european-commission-against-racism-and-intolerance/newsroom/-/asset_publisher/U2IwxeHB054o/content/ecri-webinar-on-inclusive-education-in-times-of-covid-19 (Last accessed April 2021)

Council of Europe, 2020. *Interview with Ligia Deca, Presidential Advisor to the President of Romania, Department of Education and Research*.

www.coe.int/en/web/education/interview-with-ligia-deca (Last accessed April 2021)

Council of Europe, 2020. Making the right to education a priority in times of COVID-19. A Roadmap for Action on the Council of Europe education response to COVID-19. Strasbourg: Council of Europe. rm.coe.int/making-the-right-to-education-a-reality-in-times-of-covid-19-a-roadmap/16809fee7b (Last accessed April 2021)



Council of Europe, no date. *Responses from our member States*. Education: COVID-19 Response. www.coe.int/en/web/education/responses-from-our-member-states (Last accessed April 2021)

Council of the European Union, 2020. *Council conclusions on countering the COVID-19 crisis in education and training*. Brussels: Council of the European Union. eur-lex.europa.eu/legal-

<u>content/EN/TXT/?uri=CELEX%3A52020XG0626%2801%29&qid=1620300743553</u> (Last accessed April 2021)

COVID-19 Disability Rights Monitor, 2020. *Disability rights during the pandemic*. enil.eu/wp-content/uploads/2020/10/Disability-Rights-During-the-Pandemic-report web-PDF.pdf (Last accessed April 2021)

Croatian Presidency of the Council of the European Union, 2020. *Implications of the novel coronavirus (COVID-19) on education and training: State-of-play in Member States*. Version 9 as of 31/3/2020. Unpublished

Daniels, H., Thompson, I., Porter, J., Tawell, A. and Emery, H., 2020. *School Exclusion Risks after COVID 19*. Oxford: Department of Education, University of Oxford. www.education.ox.ac.uk/wp-content/uploads/2019/11/Daniels-et-al.-2020 School-Exclusion-Risks-after-COVID-19.pdf (Last accessed April 2021)

Darling-Hammond, L., Schachner, A. and Edgerton, A.K., 2020. *Restarting and Reinventing School: Learning in the Time of COVID and Beyond*. Palo Alto, C.A.: Learning Policy Institute. learningpolicyinstitute.org/product/restarting-reinventing-school-covid (Last accessed April 2021)

Darmody, M., Smyth, E. and Russell, H., 2020. *The implications of the COVID-19 pandemic for policy in relation to children and young people: A research review*. Dublin: Economic and Social Research Institute. doi.org/10.26504/sustat94 (Last accessed April 2021)

Day, L., Percy-Smith, B., Rizzo, S., Erskine, C., Monchuk, L. and Shah, M., 2020. *To lockdown and back: Young people's lived experiences of the COVID-19 pandemic*. Ecorys and the University of Huddersfield.

<u>www.guc19.com/pdf/resource-bank/to-lockdown-and-back-research-report.pdf</u> (Last accessed April 2021)

Derbyshire, E., 2020. 'Virtual teaching for children with dyslexia: The realities and disturbing absence of research' *British Educational Research Association*. www.bera.ac.uk/blog/virtual-teaching-for-children-with-dyslexia-the-realities-and-disturbing-absence-of-research (Last accessed April 2021)

Di Pietro, G., Biagi, F., Costa, P., Karpinski, Z. and Mazza, J., 2020. *The likely impact of COVID-19 on education: Reflections based on the existing literature and recent international datasets*. Luxembourg: Publications Office of the European Union. ec.europa.eu/jrc/en/publication/likely-impact-covid-19-education-reflections-based-existing-literature-and-recent-international (Last accessed April 2021)

Disabled Children's Partnership, 2020. #LeftInLockdown – Parent carers' experiences of lockdown. London: Disabled Children's Partnership.

<u>disabledchildrenspartnership.org.uk/wp-content/uploads/2020/06/LeftInLockdown-Parent-carers'-experiences-of-lockdown-June-2020.pdf</u> (Last accessed April 2021)



Donitsa-Schmidt, S. and Ramot, R., 2020. 'Opportunities and challenges: Teacher education in Israel in the Covid-19 pandemic' *Journal of Education for Teaching*, 46 (4), 586–595. doi.org/10.1080/02607476.2020.1799708 (Last accessed April 2021)

Dorn, E., Hancock, B., Sarakatsannis, J. and Viruleg, E., 2020. *COVID-19 and learning loss – Disparities grow and students need help*. New York: McKinsey & Company. help (Last accessed April 2021)

Dumbrava, C., 2020. *Briefing. Tracking key coronavirus restrictions on movement and social life.* Brussels: European Parliamentary Research Service.

www.europarl.europa.eu/thinktank/en/document.html?reference=EPRS BRI(2020)65201 8 (Last accessed April 2021)

Edge Foundation, 2020. *The Impact of Covid-19 on Education: Evidence on the early impacts of lockdown*. London: Edge Foundation.

<u>www.edge.co.uk/sites/default/files/documents/covid-19 report final - web.pdf</u> (Last accessed February 2021)

Education Development Trust, 2020. *Emerging country-level responses to providing educational continuity under Covid-19: What's working?*

<u>www.educationdevelopmenttrust.com/our-research-and-insights/commentary/emerging-country-level-responses-to-providing-educ</u> (Last accessed April 2021)

Education Endowment Foundation, 2020. *Covid-19 Resources. Supporting schools and families through the pandemic.*

educationendowmentfoundation.org.uk/covid-19-resources (Last accessed April 2021)

Education Endowment Foundation, 2020. *Rapid evidence assessment. Distance learning*. educationendowmentfoundation.org.uk/public/files/REA -

<u>Impact of school closures on the attainment gap summary.pdf</u> (Last accessed April 2021)

Education Endowment Foundation, 2020. Rapid evidence assessment on online distance learning for school aged pupils. Protocol for a rapid evidence assessment.

<u>educationendowmentfoundation.org.uk/public/files/Campaigns/Distance_Learning_Rapid_Evidence_Assessment_Protocol.pdf</u> (Last accessed April 2021)

Education Scotland, 2021. Supporting remote learning: National Improvement Hub. education.gov.scot/improvement/supporting-remote-learning/supporting-remote-learning?dm i=LQE,76IXY,3LER3W,T3AXW,1 (Last accessed April 2021)

Escueta, M., Nickow, A.J., Oreopoulos, P. and Quan, V., 2020. 'Upgrading Education with Technology: Insights from Experimental Research' *Journal of Economic Literature*, 58 (4), 897–996. doi.org/10.1257/jel.20191507 (Last accessed April 2021)

ESHA, 2020. The impacts and challenges of COVID19 at the start of the new school year. Utrecht: European School Heads Association (ESHA). www.esha.org/news/the-impacts-and-challenges-of-covid19-at-the-start-of-the-new-school-year (Last accessed April 2021)

European Commission, 2020. *Blended learning – The new normal for teachers*. European Education Summit. www.beyond-events.eu/index.php?eventid=54&roomid=153 (Last accessed April 2021)



European Commission, 2020. Blended learning in school education – guidelines for the start of the academic year 2020/21.

<u>www.schooleducationgateway.eu/downloads/Blended learning in school</u> <u>education European Commission June 2020.pdf</u> (Last accessed April 2021)

European Commission, 2020. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee of the regions – Digital Education Action Plan 2021-2027 Resetting education and training for the digital age {SWD(2020) 209 final}.

<u>eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52020DC0624</u> (Last accessed April 2021)

European Commission, 2020. Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions on achieving the European Education Area by 2025 (SWD 2020, 212 final). eur-lex.europa.eu/legal-content/EN/TXT/?uri=CELEX%3A52020DC0625 (Last accessed April 2021)

European Commission, 2020. *Digital (dis)advantage: Creating an inclusive world for children and young people online*. Safer Internet Forum – Online Webinar. https://www.betterinternetforkids.eu/documents/167024/6210529/Safer+Internet+Forum+2020+-booklet+-+FINAL.pdf/29d9a998-1ea8-ca93-d2e6-ec42bc534981?t=1605892319665 (Last accessed April 2021)

European Commission, 2020. *Digital Education Action Plan 2021–2027: Resetting education and training for the digital age*.

<u>ec.europa.eu/education/sites/default/files/document-library-docs/deap-communication-sept2020 en.pdf</u> (Last accessed April 2021)

European Commission, 2020. Education and Training Monitor.

<u>ec.europa.eu/education/policy/strategic-framework/et-monitor en</u> (Last accessed April 2021)

European Commission, 2020. *Education and Training Monitor 2020 – Country Analysis*. Luxembourg: Publications Office of the European Union.

op.europa.eu/en/publication-detail/-/publication/cd97ef45-2497-11eb-9d7e-01aa75ed71a1/language-en/format-PDF/source-172905496 (Last accessed April 2021)

European Commission, 2020. Education and Training Monitor 2020. Teaching and learning in a digital age. Luxembourg: Publications Office of the European Union. op.europa.eu/en/publication-detail/-/publication/92c621ce-2494-11eb-9d7e-01aa75ed71a1 (Last accessed April 2021)

European Commission, 2020. Staff working document. Accompanying the document Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions – Digital Education action Plan 2021-2027 – Resetting education and training for the digital age. eur-lex.europa.eu/legal-content/LT/TXT/?uri=CELEX:52020SC0209 (Last accessed April 2021)



European Commission/EACEA/Eurydice, 2020. *Equity in school education in Europe: Structures, policies and student performance*. Luxembourg: Publications Office of the European Union.

<u>eacea.ec.europa.eu/national-policies/eurydice/sites/default/files/equity 2020 0.pdf</u> (Last accessed April 2021)

European Disability Forum, 2020. *Open letter to leaders at the EU and in EU countries: COVID-19 – Disability inclusive response*. <u>www.edf-feph.org/newsroom-news-open-letter-leaders-eu-and-eu-countries-covid-19-disability-inclusive-response</u> (Last accessed April 2021)

European Distance and E-Learning Network, 2020. 'Digital Education Action Plan 2021—2027 Setting education and training for the digital age' *EDEN webinar series: Education in time of a new normal*.

<u>www.eden-online.org/eden conference/no-6-digital-education-action-plan</u> (Last accessed April 2021)

European Parliament, 2020. The future of European education in the context of COVID-19 – European Parliament resolution of 22 October 2020 on the future of European education in the context of COVID-19 (2020/2760(RSP)).

<u>www.europarl.europa.eu/doceo/document/TA-9-2020-0282 EN.pdf</u> (Last accessed April 2021)

European Schoolnet, 2020. European Schoolnet and Covid-19: Supporting the School Community. www.eun.org/news/detail?articleld=4993184 (Last accessed April 2021)

European Schoolnet, 2020. Fostering inclusive education in times COVID-19. www.eun.org/news/detail?articleId=5419994 (Last accessed April 2021)

European Union Agency for Fundamental Rights, 2020. *Fundamental rights report – 2020*. Luxembourg: Publications Office of the European Union.

<u>fra.europa.eu/sites/default/files/fra_uploads/fra-2020-fundamental-rights-report-2020_en.pdf</u> (Last accessed April 2021)

Eurydice – European Commission, no date. Eurydice Portal.

<u>eacea.ec.europa.eu/national-policies/eurydice/search/node/pandemic_en</u> (Last accessed April 2021)

Friedrich-Ebert-Stiftung, 2020. Schule in Zeiten der Pandemie.

library.fes.de/pdf-files/studienfoerderung/16228.pdf (Last accessed April 2021)

Giannini, S. and Grant Lewis, S., 2020. 'Three ways to plan for equity during the coronavirus school closures' *World Education Blog*.

<u>gemreportunesco.wordpress.com/2020/03/25/three-ways-to-plan-for-equity-during-the-coronavirus-school-closures</u> (Last accessed April 2021)

Gladnetwork, 2020. COVID19 and Education for children with disabilities.

gladnetwork.net/covid19-and-education-children-disabilities (Last accessed April 2021)

Goodson, I.F. and Schostak, J.F., 2021. 'Curriculum and coronavirus: New approaches to curriculum in the age of uncertainty' *Prospects*, 17. doi.org/10.1007/s11125-020-09523-9 (Last accessed April 2021)



Gordon, M. and Burgess, M., 2020. *The Hidden Impact of COVID-19 on Children's Education and Learning*. London: Save the Children International. resourcecentre.savethechildren.net/node/18174/pdf/the hidden impact of covid-19 on child education.pdf (Last accessed April 2021)

Greek Chairmanship of the Committee of Ministers, 2020. *Informal Conference of Ministers of Education organised under the Greek Chairmanship of the Committee of Ministers. The education response to the COVID crisis Political declaration*. rm.coe.int/the-education-response-to-the-covid-crisis-political-declaration-for-t/16809fee7a (Last accessed April 2021)

Greenway, C.W. and Eaton-Thomas, K., 2020. 'Parent experiences of home-schooling children with special educational needs or disabilities during the coronavirus pandemic' *British Journal of Special Education*, 47 (4), 510–535. doi.org/10.1111/1467-8578.12341 (Last accessed April 2021)

Hallgarten, J., Gorgen, K. and Sims, K., 2020. Overview of emerging country-level response to providing educational continuity under COVID-19. What are the lessons learned from supporting education in conflicts and emergencies that could be relevant for EdTechrelated responses to COVID-19? Reading: Education Development Trust. edtechhub.org/wp-content/uploads/2020/05/supporting-education-conflict.pdf (Last accessed April 2021)

Handicap International, Humanity and Inclusion, 2020. *Report 2020 Executive Summary. Let's break silos now! Achieving disability-inclusive education in a post-COVID world.* Lyon: Handicap International.

<u>inee.org/system/files/resources/Report2020 Covid Summary EN final Web.pdf</u> (Last accessed April 2021)

Handicap International, Humanity and Inclusion, 2020. *Report November 2020. Let's break silos now! Achieving disability-inclusive education in a post-COVID world.* Lyon: Handicap International.

<u>reliefweb.int/sites/reliefweb.int/files/resources/Break Silos Now Inclusive Education in a post-Covid world.pdf</u> (Last accessed April 2021)

Hartshorne, R., Baumgartner, E., Kaplan-Rakowski, R., Mouza, C. and Ferdig, R., 2020. 'Special Issue Editorial: Preservice and Inservice Professional Development During the COVID-19 Pandemic' *Journal of Technology and Teacher Education*, 28 (2). oaks.kent.edu/covid19ksu/special-issue-editorial-preservice-and-inservice-professional-development-during-covid (Last accessed April 2021)

Henning Loeb, I. and Windsor, S., 2020. 'Online-and-alone (och ofta i sängen) – Elevers berättelser om gymnasietidens sista månader våren' *Paideia*, 20, 39–52. gubox.app.box.com/s/8hj5g6r3wwnlnlnaxrhj0o6vt3irjfvw (Last accessed April 2021)

Hill, C., Rosehart, P., St. Helene, J. and Sadhra, S., 2020. 'What kind of educator does the world need today? Reimagining teacher education in post-pandemic Canada' *Journal of Education for Teaching*, 46 (4), 565–575. doi.org/10.1080/02607476.2020.1797439 (Last accessed April 2021)



Hodges, C., Moore, S., Lockee, B., Trust, T. and Bond, A., 2020. 'The Difference Between Emergency Remote Teaching and Online Learning' *EDUCAUSE Review*. er.educause.edu/articles/2020/3/the-difference-between-emergency-remote-teaching-and-online-learning (Last accessed April 2021)

Huang, R., Liu, D., Tlili, A., Knyazeva, S., Chang, T.W., Zhang, X., Burgos, D., Jemni, M., Zhang, M., Zhuang, R. and Holotescu, C., 2020. *Guidance on Open Educational Practices during School Closures: Utilizing OER under COVID-19 Pandemic in line with UNESCO OER Recommendation*. Beijing: Smart Learning Institute of Beijing Normal University. Itte.unesco.org/wp-content/uploads/2020/05/Guidance-on-Open-Educational-Practices-during-School-Closures-English-Version-V1_0.pdf (Last accessed April 2021)

Huber, S.G. and Helm, C., 2020. 'COVID-19 and schooling: Evaluation, assessment and accountability in times of crises – reacting quickly to explore key issues for policy, practice and research with the school barometer' *Educational Assessment, Evaluation and Accountability*, 32 (2), 237–270. doi.org/10.1007/s11092-020-09322-y (Last accessed April 2021)

Humanity and Inclusion, 2020. *Inclusive Education sector guiding note on COVID-19*. <u>inee.org/system/files/resources/IE Sector COVID-19 6 pager last version_ENG.pdf</u> (Last accessed April 2021)

HundrED, 2020. *Quality Education for all during Covid-19 crisis*. hundred.org/en/collections/quality-education-for-all-during-covid-19 (Last accessed April 2021)

Hyseni Duraku, Z. and Nagavci, M., 2020. *The impact of the COVID-19 pandemic on the education of children with disabilities*. doi.org/10.13140/RG.2.2.17807.41125 (Last accessed April 2021)

INCLUDEnyc, no date. *IEPs and Program Adaptations Documents: What's the Difference?*New York: INCLUDEnyc. www.includenyc.org/index.php/resources/tip-sheet/ieps-remote-learning-plans-whats-the-difference (Last accessed April 2021)

INCLUDEnyc, no date. *Remote Individualized Education Program (IEP) Meetings*. New York: INCLUDEnyc. www.includenyc.org/index.php/resources/tip-sheet/remote-individualized-education-plan-iep-meetings (Last accessed April 2021)

Inclusion Europe, 2020. *Coronavirus (COVID-19) pandemic*. Brussels: Inclusion Europe. www.inclusion-europe.eu/coronavirus-pandemic (Last accessed April 2021)

Inclusion Europe, 2020. *Impact of Coronavirus emergency on people with intellectual disabilities, families in Italy*. Brussels: Inclusion Europe. www.inclusion-europe.eu/wp-content/uploads/2020/03/2020-04-09-Italy-impact-of-coronavirus.pdf (Last accessed April 2021)

Inclusion Europe, 2020. Lack of education for children with intellectual disabilities made worse in the Coronavirus emergency. Brussels: Inclusion Europe. www.inclusion-europe.eu/wp-content/uploads/2020/05/Lack-of-education-for-children-with-intellectual-disabilities-made-worse-in-the-Coronavirus-emergency.pdf (Last accessed April 2021)



Inclusion Europe, 2020. *Neglect and discrimination. Multiplied. How Covid-19 affected the rights of people with intellectual disabilities and their families.* Brussels: Inclusion Europe. www.inclusion-europe.eu/wp-content/uploads/2020/11/COVID-report-Final.pdf (Last accessed April 2021)

INEE, 2020. An inclusive response to COVID-19: Ensuring education for all. inee.org/resources/inclusive-response-covid-19-ensuring-education-all (Last accessed April 2021)

INEE, 2020. *Coronavirus (COVID-19) Resources*. <u>inee.org/covid-19/resources</u> (Last accessed April 2021)

International Disability and Development Consortium, 2020. *IDDC Inclusive Education Task Group response to COVID-19*. Brussels: IDDC.

<u>www.iddcconsortium.net/our-work/covid-19-and-disability-inclusion/inclusive-education</u> (Last accessed April 2021)

Joynes, C. and Gibbs, E., 2020. 'Report for EdTechHub (ODI). An overview of emerging country-level responses to providing educational continuity under COVID-19: What's working? What isn't?' *Education Development Trust*, 30.

<u>edtechhub.org/wp-content/uploads/2020/05/whats-working-whats-not.pdf</u> (Last accessed April 2021)

K-12 Dive, 2020. Rubric for Recovery: How schools are educating marginalized students during COVID-19. www.k12dive.com/news/rubric-for-recovery-how-schools-nationwide-are-educating-marginalized-stud/588250 (Last accessed April 2021)

Kaden, U., 2020. 'COVID-19 School Closure-Related Changes to the Professional Life of a K-12 Teacher' *Education Sciences*, 10 (6), 165. doi.org/10.3390/educsci10060165 (Last accessed April 2021)

Kalloo, R.C., Mitchell, B. and Kamalodeen, V.J., 2020. 'Responding to the COVID-19 pandemic in Trinidad and Tobago: Challenges and opportunities for teacher education' *Journal of Education for Teaching*, 46 (4), 452–462.

doi.org/10.1080/02607476.2020.1800407 (Last accessed April 2021)

Kamenetz, A., 2020. Survey Shows Big Remote Learning Gaps For Low-Income And Special Needs Children. San Francisco, C.A.: KQED.

www.kqed.org/mindshift/55976/survey-shows-big-remote-learning-gaps-for-low-incomeand-special-needs-children?fbclid=IwAR2DzR9FZLxk6zVJXRy65k (Last accessed April 2021)

Kukreja, P., 2020. 'The G20 in a Post-COVID-19 World: Bridging the Skills Gaps' *ORF Issue Brief*, 424.

<u>www.orfonline.org/wp-content/uploads/2020/11/ORF IssueBrief 424 G20-SkillsGap.pdf</u> (Last accessed April 2021)

Kurri, P. and Schmidt, P., 2020. Next Generation EU: Die Europa-Hauptstadt Brüssel für Jugendliche erlebbar machen. ÖGfE Policy Brief, 16'2020.

www.oegfe.at/wp-content/uploads/2020/10/PB-162020.pdf (Last accessed April 2021)



la Velle, L., Newman, S., Montgomery, C. and Hyatt, D., 2020. 'Initial teacher education in England and the Covid-19 pandemic: Challenges and opportunities' *Journal of Education for Teaching*, 46 (4), 596–608. doi.org/10.1080/02607476.2020.1803051 (Last accessed April 2021)

Lancet Public Health, 2020. 'Will the COVID-19 pandemic threaten the SDGs?' *The Lancet Public Health*, 5 (9), e460. doi.org/10.1016/S2468-2667(20)30189-4 (Last accessed April 2021)

Leacock, C.J. and Warrican, S.J., 2020. 'Helping teachers to respond to COVID-19 in the Eastern Caribbean: Issues of readiness, equity and care' *Journal of Education for Teaching*, 46 (4), 576–585. doi.org/10.1080/02607476.2020.1803733 (Last accessed April 2021)

Leonard Cheshire, 2020. Supporting children with disabilities at home during COVID-19. A resource pack for parents and caregivers. The Leonard Cheshire Working Group on Inclusive Education. www.leonardcheshire.org/sites/default/files/2020-05/supporting-children-disabilities-home-covid-19.pdf (Last accessed April 2021)

Levita, L., 2020. *Initial research findings on the impact of COVID-19 on the well-being of young people aged 13 to 24 in the UK*. Sheffield: COVID-19 Psychological Research Consortium (C19PRC). www.rcpch.ac.uk/sites/default/files/2020-08/Impact of COVID-19 on the well-being of young people aged 13 to 24 - University of Sheffield.pdf (Last accessed April 2021)

Lorente, L.M.L., Arrabal, A.A. and Pulido-Montes, C., 2020. 'The Right to Education and ICT during COVID-19: An International Perspective' *Sustainability*, 12 (21), 9091. doi.org/10.3390/su12219091 (Last accessed April 2021)

Lucas, M., Nelson, J. and Sims, D. 2020. 'Schools' responses to Covid-19: Pupil engagement in remote learning' *NFER*.

<u>www.nfer.ac.uk/schools-responses-to-covid-19-pupil-engagement-in-remote-learning</u> (Last accessed April 2021)

Marchese, J., 2020. *Elementary and Secondary School Children: Vulnerabilities of Online Learning*. New York: Utica College.

<u>search.proquest.com/openview/7f310b122f1c26f1cedc1a64f741a935/1?pq-origsite=gscholar&cbl=18750&diss=y</u> (Last accessed April 2021)

Marques de Miranda, D., da Silva Athanasio, B., Sena Oliveira, A.C. and Simoes-e-Silva, A.C., 2020. 'How is COVID-19 pandemic impacting mental health of children and adolescents?' *International Journal of Disaster Risk Reduction*, 51, 101845. doi.org/10.1016/j.ijdrr.2020.101845 (Last accessed April 2021)

Mavrou, K., no date. *COVID-19 and education: Use technology for a more inclusive approach*. Linz, Austria: Association for the Advancement of Assistive Technology in Europe.

<u>aaate.net/wp-content/uploads/sites/12/2020/04/inclusive education covid Mavrou.pdf</u> (Last accessed April 2021)



McAleavy, T., Joynes, C., Gibbs, E. and Sims, K., 2020. What steps are being taken to reach the most disadvantaged students during the period of Covid-19 school closure? Reading: Education Development Trust.

<u>edtechhub.org/wp-content/uploads/2020/05/disadvantaged-students.pdf</u> (Last accessed April 2021)

Mezzanotte, C., 2020. 'How coronavirus is affecting students with special education needs' *OECD Education and Skills Today*.

<u>oecdedutoday.com/coronavirus-students-special-education-needs</u> (Last accessed April 2021)

Mitchell, C. and Smith, A., 2020. 'The Lives of Girls and Young Women in the Time of COVID-19' *Girlhood Studies*, 13 (3), v–xi. doi.org/10.3167/ghs.2020.130301 (Last accessed April 2021)

Mohamad Nasri, N., Husnin, H., Mahmud, S.N.D. and Halim, L., 2020. 'Mitigating the COVID-19 pandemic: A snapshot from Malaysia into the coping strategies for pre-service teachers' education' *Journal of Education for Teaching*, 46 (4), 546–553. doi.org/10.1080/02607476.2020.1802582 (Last accessed April 2021)

Moore, G., 2020. 'School transition, socioeconomic inequalities in wellbeing and the Covid-19 pandemic' *British Educational Research Association*.

<u>www.bera.ac.uk/blog/school-transition-socioeconomic-inequalities-in-wellbeing-and-the-covid-19-pandemic</u> (Last accessed April 2021)

Moorhouse, B.L., 2020. 'Adaptations to a face-to-face initial teacher education course 'forced' online due to the COVID-19 pandemic' *Journal of Education for Teaching*, 46 (4), 609–611. doi.org/10.1080/02607476.2020.1755205 (Last accessed April 2021)

Moyo, N., 2020. 'Covid-19 and the future of practicum in teacher education in Zimbabwe: Rethinking the 'new normal' in quality assurance for teacher certification' *Journal of Education for Teaching*, 46 (4), 536–545. doi.org/10.1080/02607476.2020.1802702 (Last accessed April 2021)

Müller, L.-M. and Goldenberg, G., 2020. *Education in times of crisis: The potential implications of school closures for teachers and students*. London: Chartered College of Teaching. my.chartered.college/wp-

content/uploads/2020/05/CCTReport070520 FINAL.pdf (Last accessed April 2021)

Mutton, T., 2020. 'Teacher education and Covid-19: Responses and opportunities for new pedagogical initiatives' *Journal of Education for Teaching*, 46 (4), 439–441. doi.org/10.1080/02607476.2020.1805189 (Last accessed April 2021)

National Council on Disability, 2020. *NCD COVID-19 letter to HHS OCR*. ncd.gov/publications/2020/ncd-covid-19-letter-hhs-ocr (Last accessed April 2021)

NFER, no date. Schools' responses to Covid-19.

<u>www.nfer.ac.uk/news-events/nfer-spotlight/schools-responses-to-covid-19</u> (Last accessed April 2021)

NSPCC, no date. Supporting children with special educational needs and disabilities. www.nspcc.org.uk/keeping-children-safe/support-for-parents/coronavirus-supporting-children-special-educational-needs-disabilities (Last accessed April 2021)



Nugroho, D., Pasquini, C., Reuge, N., Amaro, D. and UNICEF Innocenti Research Centre, 2020. *COVID-19: How are Countries Preparing to Mitigate the Learning Loss as Schools Reopen? Trends and emerging good practices to support the most vulnerable children*. Innocenti Research Briefs. doi.org/10.18356/1d51512d-en (Last accessed April 2021)

OCAD University, 2020. *Defending the vulnerable during the COVID-19 pandemic*. www.ocadu.ca/news/defending-vulnerable-during-covid-19-pandemic (Last accessed April 2021)

OECD, 2020. 'Combatting COVID-19's effect on children' *OECD Policy Responses to Coronavirus (COVID-19)*. Paris: OECD Publishing. doi.org/10.1787/2e1f3b2f-en (Last accessed April 2021)

OECD, 2020. 'Learning remotely when schools close: How well are students and schools prepared? Insights from PISA' *OECD Policy Responses to Coronavirus (COVID-19)*. Paris: OECD Publishing. doi.org/10.1787/3bfda1f7-en (Last accessed April 2021)

OECD, 2020. 'Schooling disrupted, schooling rethought: How the Covid-19 pandemic is changing education' *OECD Policy Responses to Coronavirus (COVID-19)*. Paris: OECD Publishing. doi.org/10.1787/68b11faf-en (Last accessed April 2021)

OECD, 2020. *Education at a Glance 2020: OECD Indicators*. Paris: OECD Publishing. doi.org/10.1787/69096873-en (Last accessed April 2021)

OECD, 2020. Focus on social challenges.

www.oecd.org/coronavirus/en/themes/social-challenges (Last accessed April 2021)

OECD, 2020. How can education systems and schools support vulnerable students during school closures and school re-openings? A short summary. Strength through Diversity webinar, Paris, 5 October 2020. Unpublished

OECD, 2020. The impact of COVID-19 on student equity and inclusion: Supporting vulnerable students during school closures and school re-openings. Paris: OECD Publishing. www.oecd.org/education/strength-through-diversity/OECD COVID-19 Brief VulnerableStudents.pdf (Last accessed April 2021)

OECD, 2021. 'Overview and policy pointers for building a high-quality early childhood education and care workforce' *Building a High-Quality Early Childhood Education and Care Workforce: Further Results from the Starting Strong Survey 2018*. Paris: OECD Publishing. doi.org/10.1787/a9b1c97c-en (Last accessed April 2021)

OECD, no date. Education. www.oecd.org/education (Last accessed April 2021)

OECD, no date. *Lessons for education during the coronavirus crisis*. OECD Education and Skills Today. <u>oecdedutoday.com/coronavirus</u> (Last accessed April 2021)

OECD, no date. *OECD TopClass Podcast*. <u>soundcloud.com/oecdtopclasspodcast</u> (Last accessed April 2021)

Office of the United Nations High Commissioner for Human Rights, 2020. *COVID-19 and The Rights of Persons with Disabilities: Guidance*.

www.ohchr.org/Documents/Issues/Disability/COVID-

19 and The Rights of Persons with Disabilities.pdf (Last accessed April 2021)



Office of the United Nations High Commissioner for Human Rights, 2020. *COVID-19: Who is protecting the people with disabilities? – UN rights expert*.

<u>www.ohchr.org/EN/NewsEvents/Pages/DisplayNews.aspx?NewsID=25725</u> (Last accessed April 2021)

Open Education Policy Network, 2020. *Remote education in Poland during COVID-19 pandemic*. <u>oerpolicy.eu/remote-education-during-covid-19-pandemic</u> (Last accessed April 2021)

Osman, M.E., 2020. 'Global impact of COVID-19 on education systems: The emergency remote teaching at Sultan Qaboos University' *Journal of Education for Teaching*, 46 (4), 463–471. doi.org/10.1080/02607476.2020.1802583 (Last accessed April 2021)

Pålsson, S., 2020. 'Att skapa närvaro på distans' *Skola Hemma*. www.skolahemma.se/att-skapa-narvaro-pa-distans (Last accessed April 2021)

Petrie, C., Aladin, K., Ranjan, P., Javangwe, R., Gilliland, D., Tuominen, S. and Lasse, L., 2020. *Spotlight: Quality education for all during Covid-19 crisis*. HundrED and OECD. hundred-cdn.s3.amazonaws.com/uploads/report/file/15/hundred spotlight covid-19-digital.pdf (Last accessed April 2021)

Prata-Linhares, M.M., da Silva Gusmão Cardoso, T., Lopes-Jr, D.S. and Zukowsky-Tavares, C., 2020. 'Social distancing effects on the teaching systems and teacher education programmes in Brazil: Reinventing without distorting teaching' *Journal of Education for Teaching*, 46 (4), 554–564. doi.org/10.1080/02607476.2020.1800406 (Last accessed April 2021)

Primdahl, N.L., Borsch, A.S., Verelst, A., Jervelund, S.S., Derluyn, I. and Skovdal, M., 2020. "It's difficult to help when I am not sitting next to them": How COVID-19 school closures interrupted teachers' care for newly arrived migrant and refugee learners in Denmark' *Vulnerable Children and Youth Studies*, 1–11. doi.org/10.1080/17450128.2020.1829228 (Last accessed April 2021)

Psacharopoulos, G., Collis, V., Patrinos, H.A. and Vegas, E., 2020. 'Lost Wages: The COVID-19 Cost of School Closures' *SSRN Electronic Journal*. doi.org/10.2139/ssrn.3682160 (Last accessed April 2021)

Quezada, R.L., Talbot, C. and Quezada-Parker, K.B., 2020. 'From Bricks and Mortar to Remote Teaching: A Teacher Education Program's Response to COVID-19' *Journal of Education for Teaching*, 46 (4), 472–483. doi.org/10.1080/02607476.2020.1801330 (Last accessed April 2021)

Refle, J.-E., Voorpostel, M., Lebert, F., Kuhn, U., Klaas, H.S., Ryser, V.-A., Dasoki, N., Monsch, G.-A., Antal, E. and Tillmann, R., 2020. 'First results of the Swiss Household Panel – Covid-19 Study' *FORS Working Paper Series, paper 2020-1*. Lausanne: FORS. forscenter.ch/working-papers/first-results-of-the-swiss-household-panel-covid-19-study (Last accessed April 2021)

Regional Risk Communication and Community Engagement (RCCE), 2020. *COVID-19: How to include marginalized and vulnerable people in risk communication and community engagement*. interagencystandingcommittee.org/system/files/2020-03/COVID-19 - How to include marginalized and vulnerable people in risk communication and community engagement.pdf (Last accessed April 2021)



Reimers, F.M. and Schleicher, A., 2020. *OECD Policy Responses to Coronavirus (COVID-19):* A framework to guide an education response to the COVID-19 Pandemic of 2020. Paris: OECD Publishing. doi.org/10.1787/6ae21003-en (Last accessed April 2021)

Reimers, F., Schleicher, A., Saavedra, J. and Tuominen, S., 2020. Supporting the continuation of teaching and learning during the COVID-19 Pandemic: Annotated resources for online learning. Paris: OECD, Global Education Innovation Initiative, HundrED. globaled.gse.harvard.edu/files/geii/files/supporting-the-continuation-of-teaching-and-learning-during-the-covid-19-pandemic.pdf (Last accessed April 2021)

ReliefWeb, 2020. 'COVID-19 Global' *ReliefWeb – OCHA Services*. reliefweb.int/topics/covid-19-global (Last accessed April 2021)

Right to Education Initiative, 2020. 190 organisations join call for increased funding for education post-Covid. www.right-to-education.org/news/190-organisations-join-call-increased-funding-education-post-covid (Last accessed April 2021)

Rimfeld, K., Malanchini, M. and Plomin, R., 2020. 'Don't worry about cancelled exams – research shows we should switch to teacher assessment permanently' *The Conversation*. theconversation.com/dont-worry-about-cancelled-exams-research-shows-we-should-switch-to-teacher-assessment-permanently-134709 (Last accessed April 2021)

Robinson, M. and Rusznyak, L., 2020. 'Learning to teach without school-based experience: Conundrums and possibilities in a South African context' *Journal of Education for Teaching*, 46 (4), 517–527. doi.org/10.1080/02607476.2020.1800408 (Last accessed April 2021)

Roßbach, H.-G., Schmitt, M., Stanat, P., Gresch, C., Böhme, K. and Grosche, M., no date. INSIDE – Inclusive Education in Lower Secondary Schools in Germany. www.european-agency.org/sites/default/files/inside flyer en.pdf (Last accessed April 2021)

Sahlberg, P., 2021. 'Does the pandemic help us make education more equitable?' Educational Research for Policy and Practice, 20, 11–18. doi.org/10.1007/s10671-020-09284-4 (Last accessed April 2021)

Save Our Future, 2020. Averting an Education Catastrophe for the Worlds Children. Save Our Future. saveourfuture.world/wp-content/uploads/2020/10/White-Paper-FINAL.pdf (Last accessed April 2021)

Save the Children, 2020. *COVID-19: Operational Guidance for Migrant and Displaced Children*. Geneva: Save the Children. <u>bettercarenetwork.org/sites/default/files/2020-04/OperationalGuidanceCovid19andMigrationandDisplacement.pdf</u> (Last accessed April 2021)

Save the Children, 2020. The Impact of COVID-19 on children in Europe. A briefing by Save the Children. Brussels: Save the Children Europe.

<u>resourcecentre.savethechildren.net/node/17844/pdf/the impact of covid19 on childre</u> n in europe.pdf (Last accessed April 2021)



Schleicher, A., 2020. 'Education disrupted – education rebuilt: Some insights from PISA on the availability and use of digital tools for learning' *OECD Education and Skills Today*. oecdedutoday.com/coronavirus-education-digital-tools-for-learning (Last accessed April 2021)

Schleicher, A., 2020. 'How can teachers and school systems respond to the COVID-19 pandemic? Some lessons from TALIS' *OECD Directorate for Education and Skills*. oecdedutoday.com/how-teachers-school-systems-respond-coronavirus-talis (Last accessed April 2021)

School Education Gateway, 2020. *European national websites for online and distance learning during school closures*.

<u>www.schooleducationgateway.eu/de/pub/latest/news/european-national-websites.htm</u> (Last accessed April 2021)

School Education Gateway, 2020. *Inclusive education and tackling early school leaving*. www.schooleducationgateway.eu/en/pub/theme_pages/early_school_leaving.htm (Last accessed April 2021)

Scott, P., 2020. *The Impact of COVID-19 on Fair Access to Higher Education*. Edinburgh: The Scottish Government.

www.gov.scot/binaries/content/documents/govscot/publications/progress-report/2020/12/impact-covid-19-fair-access-higher-education2/documents/impact-covid-19-fair-access-higher-education/impact-covid-19-fair-access-higher-education.pdf (Last accessed April 2021)

Scull, J., Phillips, M., Sharma, U. and Garnier, K., 2020. 'Innovations in teacher education at the time of COVID19: An Australian perspective' *Journal of Education for Teaching*, 46 (4), 497–506. doi.org/10.1080/02607476.2020.1802701 (Last accessed April 2021)

Serban, A.M., Stefan, V., Potočnik, D. and Moxon, D., 2020. *Social Inclusion, Digitalisation and Young People*. Strasbourg: Council of Europe and European Commission. pip-eu.coe.int/documents/42128013/47261953/053120+Study+on+SID+Web.pdf/0057379c-2180-dd3e-7537-71c468f3cf9d (Last accessed April 2021)

Sharma, U. and May, F., 2020. 'Inclusive education during COVID-19: Lessons from teachers around the world' *Monash University Lens*. Melbourne: Monash University. lens.monash.edu/@education/2020/10/05/1381365?slug=inclusive-education-during-covid-19-lessons-from-teachers-around-the-world&fbclid=lwAR0u7gy5J3F2ZHasbgYxhg9 m-VHurj2jUtbAMUac6-HOneQnKbdj82LMnBM (Last accessed April 2021)

Sharp, C., Nelson, J., Lucas, M., Julius, J., McCrone, T. and Sims, D., 2020. 'Schools responses to Covid-19: The challenges facing schools and pupils in September 2020' *NFER*. www.nfer.ac.uk/schools-responses-to-covid-19-the-challenges-facing-schools-and-pupils-in-september-2020 (Last accessed April 2021)

Sonnemann, J., no date. 'Kids shouldn't have to repeat a year of school because of coronavirus. There are much better options' *The Conversation*.

theconversation.com/kids-shouldnt-have-to-repeat-a-year-of-school-because-of-coronavirus-there-are-much-better-options-134889 (Last accessed April 2021)



Special Needs Jungle, 2020. Coronavirus and SEND Education.

<u>www.specialneedsjungle.com/wp-content/uploads/2020/07/SNJ-Coronavirus-and-SEND-education_final.pdf</u> (Last accessed April 2021)

Special Needs Jungle, 2020. Evidence submitted by Special Needs Jungle to the 2020 House of Commons Education Committee Inquiry into the Impact of COVID-19 on Education and Children's Services. committees.parliament.uk/writtenevidence/9020/default (Last accessed April 2021)

Srivastava, P., Cardini, A., Matovich, I., Gagnon, A.A., Jenkins, R., Reuge, N. and Anderson, S., 2020. *Policy brief. COVID-19 and the global education emergency: planning systems for recovery and resilience*. inee.org/system/files/resources/T20 TF11 PB6.pdf (Last accessed April 2021)

Ștefan, V., Șerban, A. M., Mazeikis, I.Z. and Kovacs, V., no date. *Youth inclusion, digital solutions and the global pandemic*.

pip-eu.coe.int/documents/42128013/63918992/POY%2520EYE%2520Digital%2520platfor ms%2520covid%252026%2520May%25202020.pdf/ebef686d-c741-9e35-e2c1-96fb299eb759 (Last accessed April 2021)

Treviranus, J., 2020. *The Un-represented and Data to Assess Equity*. LinkedIn. www.linkedin.com/pulse/un-represented-data-assess-equity-jutta-treviranus (Last accessed April 2021)

UN Women, 2020. How COVID-19 impacts women and girls. interactive.unwomen.org/multimedia/explainer/covid19/en/index.html (Last accessed April 2021)

UNAPEI, 2020. *Ma rentrée scolaire – #jaipasecole*. Unapei – Marentrée. <u>marentree.org</u> (Last accessed April 2021)

UNESCO, 2020. 290 million students out of school due to COVID-19: UNESCO releases first global numbers and mobilizes response. en.unesco.org/news/290-million-students-out-school-due-covid-19-unesco-releases-first-global-numbers-and-mobilizes (Last accessed April 2021)

UNESCO, 2020. *COVID-19* education response: Preparing the reopening of schools: Resource paper. Paris: UNESCO. <u>unesdoc.unesco.org/ark:/48223/pf0000373401</u> (Last accessed April 2021)

UNESCO, 2020. COVID-19 response. en.unesco.org/covid19 (Last accessed April 2021)

UNESCO, 2020. *COVID-19: 10 Recommendations to plan distance learning solutions*. <u>en.unesco.org/news/covid-19-10-recommendations-plan-distance-learning-solutions</u> (Last accessed April 2021)

UNESCO, 2020. *Distance learning strategies in response to COVID-19 school closures*. Paris: UNESCO. <u>unesdoc.unesco.org/ark:/48223/pf0000373305</u> (Last accessed April 2021)

UNESCO, 2020. *Education: From disruption to recovery*. en.unesco.org/covid19/educationresponse (Last accessed April 2021)



UNESCO, 2020. Ensuring the right to quality inclusive education for persons with disabilities: From commitment to action. Online symposium, Paris, 25–27 November 2020. en.unesco.org/news/ensuring-right-quality-inclusive-education-persons-disabilities-commitment-action (Last accessed April 2021)

UNESCO, 2020. Global Education Monitoring Report 2020: Inclusion and education: All means all. Paris: UNESCO. en.unesco.org/gem-report/report/2020/inclusion (Last accessed April 2021)

UNESCO, 2020. *Inclusive School Reopening: Supporting the Most Marginalized Children to go to School*. en.unesco.org/sites/default/files/inclusive school reopening - supporting marginalised children during school re. .pdf (Last accessed April 2021)

UNESCO, 2020. International Day of Persons with Disabilities.

en.unesco.org/commemorations/personswithdisabilitiesday (Last accessed April 2021)

UNESCO, 2020. *Policy Paper 42: Act now: Reduce the impact of COVID-19 on the cost of achieving SDG 4*. Paris: UNESCO. <u>en.unesco.org/gem-report/COVIDcostSDG4</u> (Last accessed April 2021)

UNESCO, 2020. Supporting teachers in back-to-school efforts: Guidance for policy-makers. Paris: UNESCO. <u>unesdoc.unesco.org/ark:/48223/pf0000373479?posInSet=59&queryId=N-9c5b21c8-921e-4fdb-9bd6-79826ba42c55</u> (Last accessed April 2021)

UNESCO, 2020. Webinar 'Disability inclusive response for COVID-19 through Open and Distance Learning'. Online webinar, Paris, 3 December 2020. events.unesco.org/event?id=762896541&lang=1033 (Last accessed April 2021)

UNESCO, 2021. Global Education Monitoring Report 2021 – Central and Eastern Europe, the Caucasus and Central Asia – Inclusion and education: All means all. Paris: UNESCO. en.unesco.org/gem-report/Eurasia2021inclusion (Last accessed April 2021)

UNESCO, 2021. UNESCO figures show two thirds of an academic year lost on average worldwide due to Covid-19 school closures. en.unesco.org/news/unesco-figures-show-two-thirds-academic-year-lost-average-worldwide-due-covid-19-school (Last accessed April 2021)

UNESCO, no date. *World Education Blog*. <u>gemreportunesco.wordpress.com</u> (Last accessed April 2021)

UNESCO, UNICEF and the World Bank, 2020. *Reopening schools after COVID-19 closures:* webinars by UNESCO, UNICEF and the World Bank.

<u>en.unesco.org/news/joint-unesco-unicef-world-bank-webinar-series-reopening-schools</u> (Last accessed April 2021)

UNESCO, UNICEF and the World Bank, 2020. What Have We Learnt?: Overview of Findings from a Survey of Ministries of Education on National Responses to COVID-19. Paris, New York, Washington, DC: UNESCO, UNICEF, World Bank. License: CC BY-SA 3.0 IGO. openknowledge.worldbank.org/handle/10986/34700 (Last accessed April 2021)

UNESCO, UNICEF and the World Bank, 2021 (original work published 2020). *Survey on National Education Responses to COVID-19 School Closures*. github.com/worldbank/school-survey (Last accessed April 2021)



UNESCO, UNICEF, the World Bank, World Food Programme and UNHCR, 2020. *Framework for reopening schools*.

www.unicef.org/sites/default/files/2020-06/Framework-for-reopening-schools-2020.pdf (Last accessed April 2021)

UNESCO, UNICEF, the World Bank, World Food Programme and UNHCR, 2020. Supplement to Framework for reopening schools: Emerging lessons from country experiences in managing the process of reopening schools.

reliefweb.int/sites/reliefweb.int/files/resources/5f6b3dec4.pdf (Last accessed April 2021)

UNICEF, 2020. Averting a lost COVID generation: A six-point plan to respond, recover and reimagine a post-pandemic world for every child. New York: UNICEF.

<u>www.unicef.org/reports/averting-lost-generation-covid19-world-childrens-day-2020-brief</u> (Last accessed April 2021)

UNICEF, 2020. *Coronavirus disease (COVID-19) information centre*. www.unicef.org/coronavirus/covid-19 (Last accessed April 2021)

UNICEF, 2020. *COVID-19 and children. UNICEF data hub*. data.unicef.org/covid-19-and-children (Last accessed April 2021)

UNICEF, 2020. *COVID-19 response: Considerations for Children and Adults with Disabilities*. reliefweb.int/sites/reliefweb.int/files/resources/covid-

<u>19 response considerations for people with disabilities 190320.pdf</u> (Last accessed April 2021)

UNICEF, 2020. *Education: All Means All*. <u>sites.unicef.org/disabilities/index_65316.html</u> (Last accessed April 2021)

UNICEF, 2020. Protecting children and adolescents with disabilities from the pandemic. www.unicef.org/eca/protecting-children-and-adolescents-disabilities-pandemic (Last accessed April 2021)

UNICEF, 2020. Risk Communication and Community Engagement for COVID-19. Engaging with Children and Adults with Disabilities. resourcecentre.savethechildren.net/library/risk-communication-and-community-engagement-covid-19-engaging-children-and-adults (Last accessed April 2021)

UNICEF, 2020. Safe to Learn during COVID-19: Recommendations to prevent and respond to violence against children in all learning environments.

<u>www.unicef.org/media/68781/file/STL-COVID-19-Response-Key-Messages.pdf</u> (Last accessed April 2021)

UNICEF's Office of Research-Innocenti, no date. *About the COVID-19 and Children Research Library*. UNICEF: Office of Research-Innocenti.

www.unicef-irc.org/about-covid-children-library (Last accessed April 2021)

United Nations, 2020. *Global humanitarian response plan COVID-19*. Geneva: United Nations Office for the Coordination of Humanitarian Affairs.

<u>www.unocha.org/sites/unocha/files/Global-Humanitarian-Response-Plan-COVID-19.pdf</u> (Last accessed April 2021)



United Nations, 2020. Opening remarks at virtual press encounter to launch the Report on the Socio-Economic Impacts of COVID-19. www.un.org/sg/en/content/sg/speeches/2020-03-31/remarks-launch-of-report-the-socio-economic-impacts-of-covid-19 (Last accessed April 2021)

United Nations, 2020. *Policy brief: Education during COVID-19 and beyond*. New York: United Nations.

www.un.org/development/desa/dspd/wp-content/uploads/sites/22/2020/08/sg_policy_b rief_covid-19_and_education_august_2020.pdf (Last accessed April 2021)

United Nations, 2020. *United Nations Comprehensive Response to COVID-19: Saving Lives, Protecting Societies, Recovering Better*.

<u>unsdg.un.org/sites/default/files/2020-06/UN-Response-to-COVID-19.pdf</u> (Last accessed April 2021)

United Nations Children's Fund and International Telecommunication Union, 2020. How many children and young people have internet access at home. Estimating digital connectivity during the COVID-19 pandemic. New York: UNICEF.

data.unicef.org/wp-content/uploads/2020/11/How-many-children-and-young-people-have-internet-access-at-home-2020 v2.pdf (Last accessed April 2021)

Van Lancker, W. and Parolin, Z., 2020. 'COVID-19, school closures, and child poverty: A social crisis in the making' *The Lancet Public Health*, 5 (5), e243–e244. doi.org/10.1016/S2468-2667(20)30084-0 (Last accessed April 2021)

Van Nuland, S., Mandzuk, D., Tucker Petrick, K. and Cooper, T., 2020. 'COVID-19 and its effects on teacher education in Ontario: A complex adaptive systems perspective' *Journal of Education for Teaching*, 46 (4), 442–451. doi.org/10.1080/02607476.2020.1803050 (Last accessed April 2021)

Vuorikari, R., Velicu, A., Chaudron, S., Cachia, R. and Di Gioia, R., 2020. *How families handled emergency remote schooling during the Covid-19 lockdown in spring 2020*. EUR 30425 EN. Luxembourg: Publications Office of the European Union. ec.europa.eu/jrc/en/publication/eur-scientific-and-technical-research-reports/how-families-handled-emergency-remote-schooling-during-covid-19-lockdown-spring-2020 (Last accessed April 2021)

Wedell, K., 2020. 'Points from the SENCo-Forum: Lessons from the epidemic' *British Journal of Special Education*, 47 (3), 399–402. doi.org/10.1111/1467-8578.12331 (Last accessed April 2021)

WHO, 2020. Statement on the second meeting of the International Health Regulations (2005) – Emergency Committee regarding the outbreak of novel coronavirus (2019-nCoV). www.who.int/news/item/30-01-2020-statement-on-the-second-meeting-of-the-international-health-regulations-(2005)-emergency-committee-regarding-the-outbreak-of-novel-coronavirus-(2019-ncov) (Last accessed April 2021)

WHO, 2020. WHO Director-General's opening remarks at the media briefing on COVID-19 – 11 March 2020. <a href="www.who.int/director-general/speeches/detail/who-director-general-speeches/detail/who-



WHO-UNICEF-Lancet Commissioners, 2020. 'After COVID-19, a future for the world's children?' *The Lancet*, 396 (10247), 298–300. doi.org/10.1016/S0140-6736(20)31481-1 (Last accessed April 2021)

Wikipedia, 2021. Impact of the COVID-19 pandemic on education.

en.wikipedia.org/w/index.php?title=Impact_of_the_COVID
19 pandemic on education&oldid=1002718264 (Last accessed February 2021)

Wikipedia, 2021. *Impact of the COVID-19 pandemic on people with disabilities*. en.wikipedia.org/w/index.php?title=Impact of the COVID
19 pandemic on people with disabilities&oldid=1000272811 (Last accessed February 2021)

World Bank, 2020. How countries are using edtech (including online learning, radio, television, texting) to support access to remote learning during the COVID-19 pandemic. www.worldbank.org/en/topic/edutech/brief/how-countries-are-using-edtech-to-support-remote-learning-during-the-covid-19-pandemic (Last accessed April 2021)

World Bank, 2020. *Inclusive Education Initiative: Opportunities to strengthen disability-inclusive education*. <u>www.inclusive-education-initiative.org/events/opportunities-strengthen-disability-inclusive-education-promoting-inclusive-environments</u> (Last accessed April 2021)

World Bank, 2020. *International Day of Persons with Disabilities*. live.worldbank.org/international-day-persons-with-disabilities (Last accessed April 2021)

World Bank, 2020. *Pivoting to Inclusion: Leveraging Lessons from the COVID-19 Crisis for Learners with Disabilities*. www.worldbank.org/en/topic/disability/publication/pivoting-to-inclusion-leveraging-lessons-from-the-c-ovid-19-crisis-for-learners-with-disabilities (Last accessed April 2021)

World Bank, 2020. *Pivoting to Inclusion: Leveraging Lessons from the COVID-19 Crisis for Learners with Disabilities*. Washington, DC: International Bank for Reconstruction and Development / The World Bank. pubdocs.worldbank.org/en/147471595907235497/IEI-lssues-Paper-Disability-Inclusive-Education-FINAL-ACCESSIBLE.pdf (Last accessed April 2021)

World Bank, 2020. *Remote Learning, EdTech and COVID-19*. www.worldbank.org/en/topic/edutech/brief/edtech-covid-19 (Last accessed April 2021)

World Bank, 2020. *The COVID-19 Pandemic: Shocks to Education and Policy Responses*. Washington, DC: World Bank. License: CC BY 3.0 IGO.

openknowledge.worldbank.org/handle/10986/33696 (Last accessed April 2021)

World Bank, no date. *World Bank Education and COVID-19*. www.worldbank.org/en/topic/education/coronavirus (Last accessed April 2021)

Ydo, Y., 2020. 'Inclusive education: Global priority, collective responsibility' *Prospects*, 49 (3), 97–101. doi.org/10.1007/s11125-020-09520-y (Last accessed April 2021)

Yildirim, H. and Mountford-Zimdars, A., 2020. 'Lessons for adapting home learning from parents with children with special educational needs' *British Educational Research Association*. www.bera.ac.uk/blog/lessons-for-adapting-home-learning-from-parents-with-children-with-special-educational-needs (Last accessed April 2021)



10.2 List of surveys on COVID-19's impact on education

#CovidUnder19, 2020. Children's Rights during Coronavirus: Children's Views and Experiences. Belfast: Queen's University, Centre for Children's Rights.

childhub.org/sites/default/files/library/attachments/covidunder19 thematic summaries

english-20201209.pdf.pdf (Last accessed April 2021)

Active Minds, 2020. COVID-19 Impacted College Students' Mental Health Hardest, According to Nationwide Survey of Students. www.activeminds.org/press-releases/activeminds-and-association-of-college-and-university-educators-release-guide-on-practical-approaches-for-supporting-student-wellbeing-and-mental-health-copy (Last accessed April 2021)

Alliance for Inclusive Education, 2020. *The impact of COVID-19 on education and children's services*. www.allfie.org.uk/news/briefings/submission-to-the-education-committee-impact-of-covid-19-on-education-and-childrens-services-inquiry (Last accessed April 2021)

Burke, J. and Dempsey, M., 2020. *Covid-19 Practice in Primary Schools in Ireland Report*. Maynooth, Ireland: Maynooth University.

<u>www.maynoothuniversity.ie/sites/default/files/assets/document/Covid-19 Practice in Primary Schools Report 0.pdf</u> (Last accessed April 2021)

Centre for Children and Young People's Participation, 2020. *Building on Rainbows. Supporting Children's Participation in Shaping Responses to COVID-19*.

clok.uclan.ac.uk/33087/1/28%20April%20Building%20on%20Rainbows-

%20Child%20Participation%20in%20COVID-19%20Respone%20%20-

%20Rapid%20Paper .pdf (Last accessed April 2021)

Croatian Presidency of the Council of the European Union, 2020. *Implications of the novel coronavirus (COVID-19) on education and training: State-of-play in Member States*. Version 9 as of 31/3/2020. Unpublished

Daniels, H., Thompson, I., Porter, J., Tawell, A. and Emery, H., 2020. *School Exclusion Risks after COVID 19*. Oxford: Department of Education, University of Oxford. www.education.ox.ac.uk/wp-content/uploads/2019/11/Daniels-et-al.-2020_School-Exclusion-Risks-after-COVID-19.pdf (Last accessed April 2021)

Disabled Children's Partnership, 2020. #LeftInLockdown – Parent carers' experiences of lockdown. London: Disabled Children's Partnership.

<u>disabledchildrenspartnership.org.uk/wp-content/uploads/2020/06/LeftInLockdown-Parent-carers'-experiences-of-lockdown-June-2020.pdf</u> (Last accessed April 2021)

Edge Foundation, 2020. *The Impact of Covid-19 on Education: Evidence on the early impacts of lockdown*. London: Edge Foundation.

<u>www.edge.co.uk/sites/default/files/documents/covid-19 report final - web.pdf</u> (Last accessed February 2021)

Education Endowment Foundation, 2020. *Rapid evidence assessment. Distance learning* educationendowmentfoundation.org.uk/public/files/REA -

<u>Impact of school closures on the attainment gap summary.pdf</u> (Last accessed Avril 2021)



Education Endowment Foundation, 2020. Rapid evidence assessment on online distance learning for school aged pupils. Protocol for a rapid evidence assessment.

educationendowmentfoundation.org.uk/public/files/Campaigns/Distance_Learning_Rapid_Evidence_Assessment_Protocol.pdf (Last accessed April 2021)

European Commission, 2020. *Education and Training Monitor 2020 – Country Analysis*. Luxembourg: Publications Office of the European Union. op.europa.eu/en/publication-detail/-/publication/cd97ef45-2497-11eb-9d7e-01aa75ed71a1/language-en/format-PDF/source-172905496 (Last accessed April 2021)

European Commission, 2020. Staff working document. Accompanying the document Communication from the Commission to the European Parliament, the Council, the European Economic and Social Committee and the Committee of the Regions – Digital Education action Plan 2021-2027 – Resetting education and training for the digital age. eur-lex.europa.eu/legal-content/LT/TXT/?uri=CELEX:52020SC0209 (Last accessed April 2021)

Gordon, M. and Burgess, M., 2020. *The Hidden Impact of COVID-19 on Children's Education and Learning*. London: Save the Children International. resourcecentre.savethechildren.net/node/18174/pdf/the_hidden_impact_of_covid-19_on_child_education.pdf (Last accessed April 2021)

Huber, S.G. and Helm, C., 2020. 'COVID-19 and schooling: Evaluation, assessment and accountability in times of crises – reacting quickly to explore key issues for policy, practice and research with the school barometer' *Educational Assessment, Evaluation and Accountability*, 32 (2), 237–270. doi.org/10.1007/s11092-020-09322-y (Last accessed April 2021)

Joynes, C. and Gibbs, E., 2020. 'Report for EdTechHub (ODI). An overview of emerging country-level responses to providing educational continuity under COVID-19: What's working? What isn't?' *Education Development Trust*, 30. edtechhub.org/wp-content/uploads/2020/05/whats-working-whats-not.pdf (Last accessed April 2021)

Kaden, U., 2020. 'COVID-19 School Closure-Related Changes to the Professional Life of a K-12 Teacher' *Education Sciences*, 10 (6), 165. doi:org/10.3390/educsci10060165 (Last accessed April 2021)

Levita, L., 2020. *Initial research findings on the impact of COVID-19 on the well-being of young people aged 13 to 24 in the UK*. Sheffield: COVID-19 Psychological Research Consortium (C19PRC). www.rcpch.ac.uk/sites/default/files/2020-08/Impact of COVID-19 on the well-being of young people aged 13 to 24 - University of Sheffield.pdf (Last accessed April 2021)

Open Education Policy Network, 2020. *Remote education in Poland during COVID-19 pandemic*. <u>oerpolicy.eu/remote-education-during-covid-19-pandemic</u> (Last accessed April 2021)



Refle, J.-E., Voorpostel, M., Lebert, F., Kuhn, U., Klaas, H.S., Ryser, V.-A., Dasoki, N., Monsch, G.-A., Antal, E. and Tillmann, R., 2020. 'First results of the Swiss Household Panel – Covid-19 Study' *FORS Working Paper Series*, *paper 2020-1*. Lausanne: FORS. forscenter.ch/working-papers/first-results-of-the-swiss-household-panel-covid-19-study (Last accessed April 2021)

Reimers, F.M. and Schleicher, A., 2020. *OECD Policy Responses to Coronavirus (COVID-19):* A framework to guide an education response to the COVID-19 Pandemic of 2020. Paris: OECD Publishing. doi.org/10.1787/6ae21003-en (Last accessed April 2021)

Sharp, C., Nelson, J., Lucas, M., Julius, J., McCrone, T. and Sims, D., 2020. 'Schools responses to Covid-19: The challenges facing schools and pupils in September 2020' *NFER*. www.nfer.ac.uk/schools-responses-to-covid-19-the-challenges-facing-schools-and-pupils-in-september-2020 (Last accessed April 2021)

Special Needs Jungle, 2020. *Coronavirus and SEND Education*. <u>www.specialneedsjungle.com/wp-content/uploads/2020/07/SNJ-Coronavirus-and-SEND-education_final.pdf</u> (Last accessed April 2021)

Special Needs Jungle, 2020. Evidence submitted by Special Needs Jungle to the 2020 House of Commons Education Committee Inquiry into the Impact of COVID-19 on Education and Children's Services. committees.parliament.uk/writtenevidence/9020/default (Last accessed April 2021)

UNESCO, UNICEF and the World Bank, 2020. What Have We Learnt?: Overview of Findings from a Survey of Ministries of Education on National Responses to COVID-19. Paris, New York, Washington, D.C.: UNESCO, UNICEF, World Bank. License: CC BY-SA 3.0 IGO. openknowledge.worldbank.org/handle/10986/34700 (Last accessed April 2021)



11. ANNEX: OPERATIONAL DEFINITIONS

Table 2. Operational definitions used in the literature review

Term	Operational definition				
Accountability	The fact or condition of being accountable; responsibility (Oxford Dictionaries, no date). In the field of education there are three main types of accountability system: (a) compliance with regulations, (b) adherence to professional norms, and (c) results driven. School accountability systems operate according to a set of principles and use a variety of implementation strategies (UNESCO Institute for Educational Planning, 2005, p. 1). Accountability can be defined as a process by which actors provide reasons for their actions against the backdrop of possible negative (or positive) consequences (Hooge, Burns and Wilkoszewski, 2012). The concept of accountability is particularly important in the context of decentralised education systems that encourage school autonomy, including decisions about the curriculum.				
At-risk children	Children can be at risk of disadvantage because of their individual circumstances or because they, or their families belong to a group which is disadvantaged in society. These children may include those with disabilities, with mental health problems, in alternative care, at risk of neglect/abuse, undocumented child migrants/asylum seekers, those whose families live in poverty or are socially disadvantaged, those whose families have a migrant and/or second language background, those whose families have limited access to services, Roma and traveller children (European Commission, 2018a).				
Blended learning	Blended learning is a pedagogical approach mixing face-to-face and online learning, with some element of learner control over time, place, path, and pace. An example of blended learning is the flipped classroom model, in which students view lecture material prior to class, then spend class time engaging in exercises under the supervision of the teacher (European Commission, 2020d, p. 94).				
Children's rights to education	Article 28 of the UN Convention on the Rights of the Child mentions: State Parties recognise the right of children to education with a view to achieving this right progressively and on the basis of equal opportunity (UN General Assembly, 1989).				
Conceptual framework	A framework of theories, assumptions, principles and rules. It underpins a project's work and provides a 'shared vision' to guide project thinking, ensuring coherence and consistency.				



Term	Operational definition				
Cross-sectoral working	Cross-sectoral policies (e.g. for youth) should be: 'formulated and implemented with the participation of the authorities in charge of all important domains for the life of young people' (European Commission, 2018b, cited in European Agency, no date).				
Digital divide in education	Digital divide refers to 'the gap between those who can benefit from digital technology and those who cannot' (Digital Divide Institute, 2015, cited in UNESCO IITE/European Agency, 2011, p. 101).				
	The digital divide in formal schooling is not simply an equipment differential that can be overcome with further selective investments in hardware, software, and networking. Instead it derives from both within school and within home differences that extend to learning standards as well as support. Student self-learning ability, and in particular, student ability for independent learning, is an additional factor. National policies that attempt to close the digital divide for schooling must attend to all of these contributing factors to be successful (Venezky, 2000, p. 76).				
Digital education	Digital education comprises two different but complementary perspectives: the pedagogical use of digital technologies to support and enhance teaching, learning and assessment and the development of digital competences by learners and education and training staff (European Commission, 2020d, p. 95).				
Disability rights to education	The 2006 UN Convention on the Rights of Persons with Disabilities (CRPD) guaranteed the right to inclusive education but stopped short of precisely defining inclusion in education. The struggle of people with disabilities has shaped perspectives on inclusion in education. In 2016, General Comment No. 4 to CRPD Article 24 described inclusive education as involving 'a process to provide all				
	students with an equitable and participatory learning experience and environment that best corresponds to their requirements and preferences' (UNESCO, 2020f, p. 4).				



Term	Operational definition				
Disadvantage / disadvantaged	Refers to those who are more likely to experience low levels of well-being (OECD, 2017).				
	'Disadvantaged youth' in many countries is used as an umbrella category which embraces all young people with fewer opportunities than their peers and in some countries other terms like youth-at-risk, vulnerable youth, disconnected youth or social excluded youth are preferred to describe social inequality among young people Empirically, access to education and transitions in the life course are structured by categories of social inequality such as socio-economic status and class, gender and 'ethnicity'. These categories impact the educational and career options available for a young person in an inter-sectional way and lead to different forms of social inclusion and exclusion (GOETE, no date).				
Disadvantaged schools	A socio-economically disadvantaged (advantaged) school is a school whose socio-economic profile (i.e. the average socio-economic status of the students in the school) is in the bottom (top) quarter of the PISA index of economic, social and cultural status amongst all schools in the relevant country/economy (Reimers and Schleicher, 2020, p. 20).				
Distance learning	Distance learning can be understood as referring to 'methods of teaching that take place entirely outside of the classroom environment' (Education Endowment Foundation, 2020, p. 2).				
	In a broad sense, distance learning is a term often used synonymously with online learning, e-learning, distance education, correspondence education, external studies, flexible learning, and massive open online courses (MOOCs). Common features of any form of distance learning are: the teacher-learner separation by space or time, or both, and the use of media and technology to enable communication and exchange during the learning process despite this separation. This may be achieved through print-based learning materials, or one-way massive broadcasting (TV and radio programmes), or through web-based exchange using social media channels or learning platforms. Distance learning tends to require a high level of self-directed learning on the part of the learner, and study skills, which must be supported through new teaching, learning and guidance strategies (UNESCO, 2020b).				
Formative assessment	Formative assessment puts the learner at the centre of the assessment process. It provides the basis for personalisation according to the learner's interests and aptitudes.				
	Unlike summative assessment ('assessment of learning'), which has been traditionally linked to standardised, high-stakes tests and accountability, formative assessment can involve learners, enabling them to take a more active part in their learning. It is usually carried out in collaboration with others and can have substantial positive impacts on learner achievement.				



Term	Operational definition				
General education	In this review, 'general education' refers to all forms of education, which might be based on inclusive principles or not. General education might encompass mainstream education or inclusive education, but it has a broader meaning.				
Inclusive education	Various definitions of 'inclusive education' are available. They depend most often on the context where they appeared (e.g. Florian 2014, Loreman et al. 2014). According to UNESCO (2020d), inclusive education has three main justifications: educational, social and economic.				
	UNESCO IBE (2008) states that inclusive education is:				
	an ongoing process aimed at offering quality education for all while respecting diversity and the different needs and abilities, characteristics and learning expectations of the students and communities, eliminating all forms of discrimination (p. 18).				
	The Agency's position is that inclusive education is a human-rights issue, and also a part of a wider societal process working towards more inclusive and equitable societies. International organisations (UNESCO, 2017, 2020d) and European Union organisations (Council of Europe, 2020b; European Union Agency for Fundamental Rights, 2020) also argue this.				
	The Agency describes inclusive education as 'a systemic approach to providing high quality education in mainstream schools that effectively meets the academic and social learning needs of all the learners from the school's local community' (European Agency, 2015b, p. 2).				
	Inclusive education supposes a real change at both policy and practice levels regarding education. Learners are placed at the centre of a system that needs to be able to recognise, accept and respond to learner diversity. Inclusive education aims to respond to the principles of efficiency, equality and equity, where diversity is perceived as an asset. Learners also need to be prepared to engage in society, to access meaningful citizenship and to acknowledge the values of human rights, freedom, tolerance and non-discrimination (Directorate-General for Internal Policies of the Union (European Parliament) and European Agency, 2017, p. 6).				



Term	Operational definition				
Inclusive pedagogy	Inclusive pedagogy refers to 'an approach to teaching and learning that supports teachers to respond to individual differences between learners, but avoids the marginalisation that can occur when some students are treated differently' (Florian, 2014, p. 289).				
	Inclusive pedagogy requires: 1. A shift in focus from one that is concerned with only those individuals who have been identified as having 'additional needs', to learning for all – the idea of everybody (not most and some);				
	 Rejection of deterministic beliefs about ability (and the associated idea that the presence of some will hold back the progress of others); and Ways of working with and through other adults that respect the dignity of learners as full members of the community of the classroom (Florian and Black-Hawkins, 2011, p. 818). 				
Out-of-school learners	The literature on national-level work shows that different terms are used across and within countries to refer to out-of-school: not enrolled, dropout, early school leavers and not in education, employment or training (NEET) are some, but not all in evidence. In addition, there is ambiguity around the notion of 'absenteeism'. A learner's transition from being 'absent from school' to formally 'dropping-out' is rarely clearly defined in research or data collection.				
	Within work linked to Sustainable Development Goal 4, UNESCO has developed the following working definitions for: Out-of-school children: 'Children in the official primary school age				
	 range who are not enrolled in either primary or secondary school'; Out-of-school adolescents and youth: 'Those of lower or upper secondary school age who are not enrolled in primary, secondary, post-secondary non-tertiary or tertiary education' (UNESCO, 2018, p. 356). 				
	Being considered out-of-school is most often linked to a learner's age in relation to compulsory education, and their access to, enrolment in and participation in some form of educational provision.				
	The COVID-19 crisis in education will probably reshape the notions of 'out-of-school' and 'drop-out'. School closures, moves towards distance and blended teaching and learning methods and the rise in home schooling have implications for access to education and access to an inclusive education for a growing number of learners.				



Term	Operational definition				
Participation	JNICEF describes participation as:				
	an ongoing process of children's expression and active involvement in decision-making at different levels in matters that concern them, requiring information-sharing and dialogue between children/adolescents and adults based on mutual respect, and [requiring] that full consideration of their views be given, taking into account the child's age and maturity (2013, p. 7).				
	Like educational inclusion, participation is often seen as a human rights issue and an essential component of social justice (European Agency, 2011).				
	In inclusive education, 'participation' includes – but is different from – other notions such as presence, placement and progress.				
	[Presence] is about learners' access to and attendance within the education system				
	[Placement] refers to where learners are placed within the education system, meaning to what extent they are enrolled and educated in inclusive or segregated settings				
	[Participation] is about the quality of the learning experience from a learner perspective, and therefore it must incorporate the views of the learners themselves It relates to school-level process factors which facilitate or hinder a sense of belonging and a sense of autonomy to the learner, as well as a sense of a meaningful participation with peers of the same age. As such, participation mainly refers to processes at the meso (school or classroom) and micro (individual learner) levels.				
	[Progress] refers to the results of participating in educational activities, the learning processes and outcomes across the curriculum and other outcomes such as school well-being, and what future opportunities this provides for the individual learner (Ramberg and Watkins, 2020, pp. 89–90).				
Remote education	Method of delivery, which involves teaching and learning activities where educators and learners are not physically present in one location at the same time. In this case, learning happens away from the physical site of an educational provider with educators and learners using different means to connect and engage with a programme, course or educational activity remote education is used as a broad term which compromises, among others, the possibility to organise and deliver teaching and learning activities at distance (e.g. by using radio, TV or electronic resources) or online (e.g. requiring learners to use a connected device) (European Commission, 2020d, pp. 97–98).				



Term	Operational definition			
Vulnerability and vulnerable learners	According to SDG 4, target 4.5 on inclusion and equity: All people, irrespective of sex, age, race, colour, ethnicity, language, religion, political or other opinion, national or social origin, property or birth, as well as persons with disabilities, migrants, indigenous peoples, and children and youth, especially those in vulnerable situations or other status, should have access to inclusive, equitable quality education and lifelong learning opportunities. Vulnerable groups that require particular attention and targeted strategies include persons with disabilities, indigenous peoples, ethnic minorities and the poor (SDG-Education 2030 Steering Committees, no date).			
	Vulnerabilities have a dynamic dimension and can vary by place (Ainscow, 2005). Vulnerabilities can refer to poverty, ethnicity, disability and remoteness. many types of vulnerability are not outwardly apparent making it impossible to distinguish neatly between students with and without disabilities or special needs (UNESCO, 2020f, p. 66).			
	Many countries identify specific groups as vulnerable in constitutions, social inclusion legislation, education legislation or documents directly related to inclusive education. The group most identified is people with disabilities, but women and girls, rural or remote populations and the poor are also commonly recognized. Few countries link recognition of specific groups with a mandate to collect data on their inclusion in education, however (UNESCO, 2020f, p. 67).			
	Characteristics that expose individuals to risk do not affect everybody the same way. For instance, life at the intersections of disability with race, class, gender, sexual orientation and gender identity expression is more than the sum of each vulnerability (UNESCO, 2020f, p. 73).			



mental and emotional health same time, a multidimensional concept (OECD, 2020e, p. 20) with a social and economic dimension, a psychological and mental health dimension, a philosophical dimension and an educational dimension (Mashford-Scott, Church and Taylor, 2012). It includes a sociological focus on external living conditions, a psychological and public health focus on person-related indicators, such as self-efficacy or self-esteem, and a focus on subjective experiences and subjective feelings of happiness or satisfaction. Well-being is close to the notion of 'good life', which refers to a person's quality of life, and not the quantity of skills they have acquired. It is about	Term	Operational definition
Well-being depends on culture, historical circumstances, socio-economic class and other variables, which illustrate the differences in how well-	mental and	Church and Taylor, 2012). It includes a sociological focus on external living conditions, a psychological and public health focus on person-related indicators, such as self-efficacy or self-esteem, and a focus on subjective experiences and subjective feelings of happiness or satisfaction. Well-being is close to the notion of 'good life', which refers to a person's quality of life, and not the quantity of skills they have acquired. It is about enjoying 'being', judged by what it brings in the life of others, and the capacity to derive joy from life, which everyone can achieve (Kittay, 2019). Well-being depends on culture, historical circumstances, socio-economic class and other variables, which illustrate the differences in how well-being is conceived. The notion of well-being has been reshaped during the COVID-19 pandemic in relation to mental and emotional health, among other things. In a rights-based approach, children can be active participants in defining

Secretariat:

Østre Stationsvej 33 DK-5000 Odense C Denmark Tel: +45 64 41 00 20 secretariat@european-agency.org

Brussels Office:

Rue Montoyer 21 BE-1000 Brussels Belgium Tel: +32 2 213 62 80

brussels.office@european-agency.org

www.european-agency.org